# CH \$1365.00 138298

# TRADEMARK ASSIGNMENT COVER SHEET

Electronic Version v1.1 Stylesheet Version v1.2 ETAS ID: TM335832

SUBMISSION TYPE:	NEW ASSIGNMENT
NATURE OF CONVEYANCE:	RELEASE OF SECURITY INTEREST

## **CONVEYING PARTY DATA**

Name	Formerly	Execution Date	Entity Type
ISI BRANDS INC.		12/23/2003	CORPORATION: MICHIGAN

### **RECEIVING PARTY DATA**

Name:	CHEMICAL BANK
Street Address:	270 PARK AVENUE
City:	NEW YORK
State/Country:	NEW YORK
Postal Code:	10017
Entity Type:	CORPORATION: NEW YORK
Name:	ROSS BLECHMAN
Street Address:	150 MOTOR PARKWAY
City:	HAUPPAUGE
State/Country:	NEW YORK
Postal Code:	11788
Entity Type:	INDIVIDUAL: UNITED STATES
Name:	LINDA BLECHMAN
Street Address:	150 MOTOR PARKWAY
City:	HAUPPAUGE
State/Country:	NEW YORK
Postal Code:	11788
Entity Type:	INDIVIDUAL: UNITED STATES
Name:	DEAN BLECHMAN
Street Address:	150 MOTOR PARKWAY
City:	HAUPPAUGE
State/Country:	NEW YORK
Postal Code:	11788
Entity Type:	INDIVIDUAL: UNITED STATES
Name:	SHARON BLECHMAN
Street Address:	150 MOTOR PARKWAY
City:	HAUPPAUGE
State/Country:	NEW YORK
	TRANFMARK

TRADEMARK

900319328 REEL: 005482 FRAME: 0845

Entity Type: INDIVIDUAL: UNITED STATES  Name: BRIAN BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: ROBIN BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: STEVE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: STEVE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: ELYSE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES	Postal Code:	11788
Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: ROBIN BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: STEVE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: STEVE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: ELYSE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788	Entity Type:	INDIVIDUAL: UNITED STATES
City: HAUPPAUGE State/Country: NEW YORK Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES Name: ROBIN BLECHMAN Street Address: 150 MOTOR PARKWAY City: HAUPPAUGE State/Country: NEW YORK Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES Name: STEVE BLECHMAN Street Address: 150 MOTOR PARKWAY City: HAUPPAUGE State/Country: NEW YORK Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES Name: STEVE BLECHMAN Street Address: 150 MOTOR PARKWAY City: HAUPPAUGE State/Country: NEW YORK Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES Name: ELYSE BLECHMAN Street Address: 150 MOTOR PARKWAY City: HAUPPAUGE State/Country: NEW YORK Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES Name: NEIL BLECHMAN Street Address: 150 MOTOR PARKWAY City: HAUPPAUGE State/Country: NEW YORK Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES Name: NEIL BLECHMAN Street Address: 150 MOTOR PARKWAY City: HAUPPAUGE State/Country: NEW YORK Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES Name: NEIL BLECHMAN Street Address: 150 MOTOR PARKWAY City: HAUPPAUGE State/Country: NEW YORK Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES Name: HELENA BLECHMAN Street Address: 150 MOTOR PARKWAY City: HAUPPAUGE State/Country: NEW YORK Postal Code: 11788	Name:	BRIAN BLECHMAN
State/Country: NEW YORK  Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: ROBIN BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE State/Country: NEW YORK  Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: STEVE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE State/Country: NEW YORK  Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: STEVE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE State/Country: NEW YORK  Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: ELYSE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE State/Country: NEW YORK  Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE State/Country: NEW YORK  Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788	Street Address:	150 MOTOR PARKWAY
Postal Code:	City:	HAUPPAUGE
Entity Type: INDIVIDUAL: UNITED STATES  Name: ROBIN BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: STEVE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: ELYSE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: ELYSE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788	State/Country:	NEW YORK
Name: ROBIN BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: STEVE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: ELYSE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: ELYSE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788	Postal Code:	11788
Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: STEVE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: ELYSE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: BLYSE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788	Entity Type:	INDIVIDUAL: UNITED STATES
City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: STEVE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: ELYSE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: ELYSE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788	Name:	ROBIN BLECHMAN
State/Country: NEW YORK  Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: STEVE BLECHMAN Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE State/Country: NEW YORK  Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: ELYSE BLECHMAN Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE State/Country: NEW YORK  Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: BLYSE BLECHMAN Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE State/Country: NEW YORK  Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE State/Country: NEW YORK  Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: NEW YORK  Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE State/Country: NEW YORK  Postal Code: 11788	Street Address:	150 MOTOR PARKWAY
Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES Name: STEVE BLECHMAN Street Address: 150 MOTOR PARKWAY City: HAUPPAUGE State/Country: NEW YORK Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES Name: ELYSE BLECHMAN Street Address: 150 MOTOR PARKWAY City: HAUPPAUGE State/Country: NEW YORK Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES Name: NEW YORK Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES Name: NEIL BLECHMAN Street Address: 150 MOTOR PARKWAY City: HAUPPAUGE State/Country: NEW YORK Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES Name: NEIL BLECHMAN Street Address: 150 MOTOR PARKWAY City: HAUPPAUGE State/Country: NEW YORK Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES Name: HELENA BLECHMAN Street Address: 150 MOTOR PARKWAY City: HAUPPAUGE State/Country: NEW YORK Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES Name: HELENA BLECHMAN Street Address: 150 MOTOR PARKWAY City: HAUPPAUGE State/Country: NEW YORK Postal Code: 11788	City:	HAUPPAUGE
Entity Type: INDIVIDUAL: UNITED STATES  Name: STEVE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: ELYSE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788	State/Country:	NEW YORK
Name: STEVE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: ELYSE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788	Postal Code:	11788
Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: ELYSE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Postal Code: 11788	Entity Type:	INDIVIDUAL: UNITED STATES
City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: ELYSE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788	Name:	STEVE BLECHMAN
State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: ELYSE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788	Street Address:	150 MOTOR PARKWAY
Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: ELYSE BLECHMAN Street Address: 150 MOTOR PARKWAY City: HAUPPAUGE State/Country: NEW YORK Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN Street Address: 150 MOTOR PARKWAY City: HAUPPAUGE State/Country: NEW YORK Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES Name: HELPAUGE State/Country: NEW YORK Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN Street Address: 150 MOTOR PARKWAY City: HAUPPAUGE State/Country: NEW YORK Postal Code: 150 MOTOR PARKWAY Street Address: 150 MOTOR PARKWAY City: HAUPPAUGE State/Country: NEW YORK Postal Code: 11788	City:	HAUPPAUGE
Entity Type: INDIVIDUAL: UNITED STATES  Name: ELYSE BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788	State/Country:	NEW YORK
Name: ELYSE BLECHMAN Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE State/Country: NEW YORK  Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE State/Country: NEW YORK  Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN Street Address: 150 MOTOR PARKWAY  City: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE State/Country: NEW YORK  Postal Code: 11788	Postal Code:	11788
Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788	Entity Type:	INDIVIDUAL: UNITED STATES
City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788	Name:	ELYSE BLECHMAN
State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788	Street Address:	150 MOTOR PARKWAY
Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788	City:	HAUPPAUGE
Entity Type: INDIVIDUAL: UNITED STATES  Name: NEIL BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788	State/Country:	NEW YORK
Name: NEIL BLECHMAN Street Address: 150 MOTOR PARKWAY City: HAUPPAUGE State/Country: NEW YORK Postal Code: 11788 Entity Type: INDIVIDUAL: UNITED STATES Name: HELENA BLECHMAN Street Address: 150 MOTOR PARKWAY City: HAUPPAUGE State/Country: NEW YORK Postal Code: 11788	Postal Code:	11788
Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788	Entity Type:	INDIVIDUAL: UNITED STATES
City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788	Name:	NEIL BLECHMAN
State/Country: NEW YORK  Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788	Street Address:	150 MOTOR PARKWAY
Postal Code: 11788  Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788	City:	HAUPPAUGE
Entity Type: INDIVIDUAL: UNITED STATES  Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788	State/Country:	NEW YORK
Name: HELENA BLECHMAN  Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788	Postal Code:	11788
Street Address: 150 MOTOR PARKWAY  City: HAUPPAUGE  State/Country: NEW YORK  Postal Code: 11788	Entity Type:	INDIVIDUAL: UNITED STATES
City: HAUPPAUGE State/Country: NEW YORK Postal Code: 11788	Name:	HELENA BLECHMAN
State/Country: NEW YORK Postal Code: 11788	Street Address:	150 MOTOR PARKWAY
Postal Code: 11788	City:	HAUPPAUGE
	State/Country:	NEW YORK
Entity Type: INDIVIDUAL: UNITED STATES	Postal Code:	11788
	Entity Type:	INDIVIDUAL: UNITED STATES

Property Type	Number	Word Mark
Registration Number:	1382984	ALLERDOPHILUS
Registration Number:	1428089	ALLERGY FIGHTERS
Registration Number:	1380032	ALLERGY-C
Registration Number:	1759398	ALVITA
Registration Number:	1437049	AMINO FUEL
Registration Number:	1509858	ANABOLIC FUEL
Registration Number:	1502771	ANIMAL FRIENDS
Registration Number:	1167990	BIO-FORMED
Registration Number:	1333948	CAL-QUICK
Registration Number:	1432677	CARDIO-CARE
Registration Number:	1119092	CHER-AMINO
Registration Number:	1912216	CHOLINE COCKTAIL
Registration Number:	1583283	CHROMIC FUEL
Registration Number:	1632890	CITRUS-POWER
Registration Number:	1864175	CREATINE FUEL
Registration Number:	1148634	ENDURANCE
Registration Number:	1374544	EVER-SLENDER
Registration Number:	1281261	FIBERSOL
Registration Number:	1508880	GAINERS FUEL
Registration Number:	1777621	GENTLE C-250
Registration Number:	1954899	GLUTAMINE FUEL
Registration Number:	2139522	HERBAL REMETEAS
Registration Number:	1597121	HUMAN MACHINE OIL
Registration Number:	1655100	HYDRA FUEL
Registration Number:	1428090	IMMUNE PROTECTORS
Registration Number:	1417161	LIQUI-E
Registration Number:	1295575	LIQUI-K
Registration Number:	1121723	LPP
Registration Number:	0915606	LPP
Registration Number:	1692371	MASS FUEL
Registration Number:	1321725	MAXILIFE
Registration Number:	1597123	MCT FUEL
Registration Number:	1782018	METABOLIFT
Registration Number:	1636638	MORE OF NATURE'S GOODNESS
Registration Number:	1784724	OCUGUARD
Registration Number:	1763989	OPTI-CAL
Registration Number:	1584280	OPTI FUEL

Property Type	Number	Word Mark
Registration Number:	1698746	PHOS FUEL
Registration Number:	1400866	POWER
Registration Number:	1709127	POWER FUEL
Registration Number:	1773392	RIPPED FUEL
Registration Number:	1635933	SKINNY FORMULA
Registration Number:	1471933	STRESSMATES
Registration Number:	1364129	STRESS-RELEASE
Registration Number:	1954900	SUPER VANADYL FUEL
Registration Number:	1692361	THE LEADER IN SPORTS NUTRITION
Registration Number:	1500151	TRI-BORON
Registration Number:	1603361	TWINFAST
Registration Number:	1298349	TWINLAB
Registration Number:	1577237	TWINLAB
Registration Number:	1540057	ULTRA FUEL
Registration Number:	1781039	VEGE FUEL
Registration Number:	1192419	VITA QUICK
Registration Number:	1369779	YEAST FIGHTERS

### **CORRESPONDENCE DATA**

**Fax Number:** 2485677423

Correspondence will be sent to the e-mail address first; if that is unsuccessful, it will be sent using a fax number, if provided; if that is unsuccessful, it will be sent via US Mail.

**Phone:** 2485677400

**Email:** tkkroninger@varnumlaw.com

Correspondent Name: Timothy K. Kroninger

Address Line 1: 39500 High Pointe Boulevard

Address Line 2: Suite 350

Address Line 4: Novi, MICHIGAN 48375-5516

ATTORNEY DOCKET NUMBER:	ISI RELEASES BILL #110548
NAME OF SUBMITTER:	Timothy K. Kroninger
SIGNATURE:	/Timothy K. Kroninger/
DATE SIGNED:	03/20/2015

### **Total Attachments: 147**

source=ISI Brands-Insert for Trademark Release-Twinlab re Chemical Bank &

Blechman 3-11-2015ms 3-21-15tk#page1.tif

source=ISI Brands-Twin Lab-Trademark Assignment Agreement with Schedule A\_12-23-2003 Acquisition Closing#page1.tif

source=ISI Brands-Twin Lab-Trademark Assignment Agreement with Schedule A\_12-23-2003 Acquisition Closing#page2.tif

source=ISI Brands-Twin Lab-Trademark Assignment Agreement with Schedule A\_12-23-2003 Acquisition Closing#page3.tif

source=ISI Brands-Twin Lab-Trademark Assignment Agreement with Schedule A\_12-23-2003 Acquisition Closing#page4.tif

source=ISI Brands-Twin Lab-Trademark Assignment Agreement with Schedule A\_12-23-2003 Acquisition Closing#page5.tif

source=ISI Brands-Twin Lab-Trademark Assignment Agreement with Schedule A\_12-23-2003 Acquisition Closing#page6.tif

source=ISI Brands-Twin Lab-Trademark Assignment Agreement with Schedule A\_12-23-2003 Acquisition Closing#page7.tif

source=ISI Brands-Twin Lab-Trademark Assignment Agreement with Schedule A\_12-23-2003 Acquisition Closing#page8.tif

source=ISI Brands-Twin Lab-Trademark Assignment Agreement with Schedule A\_12-23-2003 Acquisition Closing#page9.tif

source=ISI Brands-Twin Lab-Trademark Assignment Agreement with Schedule A\_12-23-2003 Acquisition Closing#page10.tif

source=ISI Brands-Twin Lab-Trademark Assignment Agreement with Schedule A\_12-23-2003 Acquisition Closing#page11.tif

source=ISI Brands-Twin Lab-Trademark Assignment Agreement with Schedule A\_12-23-2003 Acquisition Closing#page12.tif

source=ISI Brands-Twin Lab-Trademark Assignment Agreement with Schedule A\_12-23-2003 Acquisition Closing#page13.tif

source=ISI Brands-Twin Lab-Trademark Assignment Agreement with Schedule A\_12-23-2003 Acquisition Closing#page14.tif

source=ISI Brands-Twin Lab-Trademark Assignment Agreement with Schedule A\_12-23-2003 Acquisition Closing#page15.tif

source=ISI Brands-Twin Lab-Trademark Assignment Agreement with Schedule A\_12-23-2003 Acquisition Closing#page16.tif

source=ISI Brands-Twin Lab-Trademark Assignment Agreement with Schedule A\_12-23-2003 Acquisition Closing#page17.tif

source=ISI Brands-Twin Lab-Trademark Assignment Agreement with Schedule A\_12-23-2003 Acquisition Closing#page18.tif

source=ISI Brands-Twin Lab-Trademark Assignment Agreement with Schedule A\_12-23-2003 Acquisition Closing#page19.tif

source=ISI Brands-Twin Lab-Trademark Assignment Agreement with Schedule A\_12-23-2003 Acquisition Closing#page20.tif

source=ISI Brands-Twin Lab-Trademark Assignment Agreement with Schedule A\_12-23-2003 Acquisition Closing#page21.tif

source=ISI Brands-Twin Lab-Trademark Assignment Agreement with Schedule A\_12-23-2003 Acquisition Closing#page22.tif

source=ISI Brands-Twin Lab-Trademark Assignment Agreement with Schedule A\_12-23-2003 Acquisition Closing#page23.tif

source=ISI Brands-Twin Lab-Trademark Assignment Agreement with Schedule A\_12-23-2003 Acquisition Closing#page24.tif

source=ISI Brands-Twin Lab-Trademark Assignment Agreement with Schedule A\_12-23-2003 Acquisition Closing#page25.tif

 $source = ISI\ Brands - Idea Sphere - Twin Lab - Bankrupt cy\ Order\ Authorizing\ Sale\ \&\ Transfer\_10 - 30 - 2003 \# page 1. tifned the properties of the p$ 

source=ISI Brands-IdeaSphere-TwinLab-Bankruptcy Order Authorizing Sale & Transfer\_10-30-2003#page2.tif

source=ISI Brands-IdeaSphere-TwinLab-Bankruptcy Order Authorizing Sale & Transfer\_10-30-2003#page3.tif

source=ISI Brands-IdeaSphere-TwinLab-Bankruptcy Order Authorizing Sale & Transfer\_10-30-2003#page4.tif source=ISI Brands-IdeaSphere-TwinLab-Bankruptcy Order Authorizing Sale & Transfer 10-30-2003#page5.tif

source=ISI Brands-IdeaSphere-TwinLab-Bankruptcy Order Authorizing Sale & Transfer\_10-30-2003#page6.tif

source=ISI Brands-IdeaSphere-TwinLab-Bankruptcy Order Authorizing Sale & Transfer\_10-30-2003#page7.tif

source=ISI Brands-IdeaSphere-TwinLab-Bankruptcy Order Authorizing Sale & Transfer\_10-30-2003#page8.tif

source=ISI Brands-IdeaSphere-TwinLab-Bankruptcy Order Authorizing Sale & Transfer 10-30-2003#page9.tif source=ISI Brands-IdeaSphere-TwinLab-Bankruptcy Order Authorizing Sale & Transfer\_10-30-2003#page10.tif source=ISI Brands-IdeaSphere-TwinLab-Bankruptcy Order Authorizing Sale & Transfer 10-30-2003#page11.tif source=ISI Brands-IdeaSphere-TwinLab-Bankruptcy Order Authorizing Sale & Transfer 10-30-2003#page12.tif source=ISI Brands-IdeaSphere-TwinLab-Bankruptcy Order Authorizing Sale & Transfer\_10-30-2003#page13.tif source=ISI Brands-IdeaSphere-TwinLab-Bankruptcy Order Authorizing Sale & Transfer 10-30-2003#page14.tif source=ISI Brands-IdeaSphere-TwinLab-Bankruptcy Order Authorizing Sale & Transfer 10-30-2003#page15.tif source=ISI Brands-IdeaSphere-TwinLab-Bankruptcy Order Authorizing Sale & Transfer\_10-30-2003#page16.tif source=ISI Brands-IdeaSphere-TwinLab-Bankruptcy Order Authorizing Sale & Transfer\_10-30-2003#page17.tif source=ISI Brands-IdeaSphere-TwinLab-Bankruptcy Order Authorizing Sale & Transfer 10-30-2003#page18.tif source=ISI Brands-IdeaSphere-TwinLab-Bankruptcy Order Authorizing Sale & Transfer\_10-30-2003#page19.tif source=ISI Brands-IdeaSphere-TwinLab-Bankruptcy Order Authorizing Sale & Transfer\_10-30-2003#page20.tif source=ISI Brands-IdeaSphere-TwinLab-Bankruptcy Order Authorizing Sale & Transfer 10-30-2003#page21.tif source=ISI Brands-IdeaSphere-TwinLab-Bankruptcy Order Authorizing Sale & Transfer 10-30-2003#page22.tif source=ISI Brands-IdeaSphere-TwinLab-Bankruptcy Order Authorizing Sale & Transfer 10-30-2003#page23.tif source=ISI Brands-IdeaSphere-TwinLab-Bankruptcy Order Authorizing Sale & Transfer\_10-30-2003#page24.tif source=ISI Brands-IdeaSphere-TwinLab-Bankruptcy Order Authorizing Sale & Transfer\_10-30-2003#page25.tif source=ISI Brands-IdeaSphere-TwinLab-Bankruptcy Order Authorizing Sale & Transfer 10-30-2003#page26.tif source=ISI Brands-Blechman Release re Trademarks Asset Purchase Agreement 12-18-2003#page1.tif source=ISI Brands-Blechman Release re Trademarks Asset Purchase Agreement 12-18-2003#page2.tif source=ISI Brands-Blechman Release re Trademarks Asset Purchase Agreement 12-18-2003#page3.tif source=ISI Brands-Blechman Release re Trademarks Asset Purchase Agreement 12-18-2003#page4.tif source=ISI Brands-Blechman Release re Trademarks Asset Purchase Agreement 12-18-2003#page5.tif source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page1.tif source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page2.tif source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page3.tif source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page4.tif source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page5.tif source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page6.tif source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement\_9-4-2003#page7.tif source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page8.tif source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page9.tif source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement\_9-4-2003#page10.tif source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement\_9-4-2003#page11.tif source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page12.tif source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page13.tif source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement\_9-4-2003#page14.tif source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement\_9-4-2003#page15.tif source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page16.tif source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page17.tif source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement\_9-4-2003#page18.tif source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page19.tif source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page20.tif source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page21.tif source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page22.tif source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page23.tif source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page24.tif source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page25.tif

```
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page26.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page27.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page28.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page29.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page30.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page31.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page32.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page33.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page34.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page35.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement_9-4-2003#page36.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement_9-4-2003#page37.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page38.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page39.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page40.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement_9-4-2003#page41.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page42.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page43.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page44.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page45.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page46.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page47.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page48.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page49.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page50.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page51.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page52.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page53.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page54.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement_9-4-2003#page55.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page56.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page57.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement_9-4-2003#page58.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement_9-4-2003#page59.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page60.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page61.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement_9-4-2003#page62.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement_9-4-2003#page63.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page64.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page65.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement_9-4-2003#page66.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page67.tif
source=ISI Brands-IdeaSphere-Twin Lab-Asset Purchase Agreement 9-4-2003#page68.tif
source=ISI Brands-IdeaSphere-Twin Lab-Amendments Nos 1 2 3 to Asset Purchase Agreement 12-5-2003 &
12-19-2003#page1.tif
source=ISI Brands-IdeaSphere-Twin Lab-Amendments Nos 1 2 3 to Asset Purchase Agreement 12-5-2003 &
12-19-2003#page2.tif
source=ISI Brands-IdeaSphere-Twin Lab-Amendments Nos 1 2 3 to Asset Purchase Agreement 12-5-2003 &
```

12-19-2003#page3.tif

source=ISI Brands-IdeaSphere-Twin Lab-Amendments Nos 1\_2\_3 to Asset Purchase Agreement\_12-5-2003 & 12-19-2003#page4.tif

source=ISI Brands-IdeaSphere-Twin Lab-Amendments Nos 1\_2\_3 to Asset Purchase Agreement\_12-5-2003 & 12-19-2003#page5.tif

source=ISI Brands-IdeaSphere-Twin Lab-Amendments Nos 1\_2\_3 to Asset Purchase Agreement\_12-5-2003 & 12-19-2003#page6.tif

source=ISI Brands-IdeaSphere-Twin Lab-Amendments Nos 1\_2\_3 to Asset Purchase Agreement\_12-5-2003 & 12-19-2003#page7.tif

source=ISI Brands-IdeaSphere-Twin Lab-Amendments Nos 1\_2\_3 to Asset Purchase Agreement\_12-5-2003 & 12-19-2003#page8.tif

source=ISI Brands-IdeaSphere-Twin Lab-Amendments Nos 1\_2\_3 to Asset Purchase Agreement\_12-5-2003 & 12-19-2003#page9.tif

source=ISI Brands-IdeaSphere-Twin Lab-Amendments Nos 1\_2\_3 to Asset Purchase Agreement\_12-5-2003 & 12-19-2003#page10.tif

source=ISI Brands-IdeaSphere-Twin Lab-Amendments Nos 1\_2\_3 to Asset Purchase Agreement\_12-5-2003 & 12-19-2003#page11.tif

source=ISI Brands-IdeaSphere-Twin Lab-Amendments Nos 1\_2\_3 to Asset Purchase Agreement\_12-5-2003 & 12-19-2003#page12.tif

source=ISI Brands-IdeaSphere-Twin Lab-Amendments Nos 1\_2\_3 to Asset Purchase Agreement\_12-5-2003 & 12-19-2003#page13.tif

source=ISI Brands-IdeaSphere-Twin Lab-Amendments Nos 1\_2\_3 to Asset Purchase Agreement\_12-5-2003 & 12-19-2003#page14.tif

source=ISI Brands-IdeaSphere-Twin Lab-Amendments Nos 1\_2\_3 to Asset Purchase Agreement\_12-5-2003 & 12-19-2003#page15.tif

source=ISI Brands-IdeaSphere-Twin Lab-Amendments Nos 1\_2\_3 to Asset Purchase Agreement\_12-5-2003 & 12-19-2003#page16.tif

source=ISI Brands-IdeaSphere-Twin Lab-Amendments Nos 1\_2\_3 to Asset Purchase Agreement\_12-5-2003 & 12-19-2003#page17.tif

source=ISI Brands-IdeaSphere-Twin Lab-Amendments Nos 1\_2\_3 to Asset Purchase Agreement\_12-5-2003 & 12-19-2003#page18.tif

source=ISI Brands-IdeaSphere-Twin Lab-Amendments Nos 1\_2\_3 to Asset Purchase Agreement\_12-5-2003 & 12-19-2003#page19.tif

source=ISI Brands-IdeaSphere-Twin Lab-Amendments Nos 1\_2\_3 to Asset Purchase Agreement\_12-5-2003 & 12-19-2003#page20.tif

source=ISI Brands-IdeaSphere-Twin Lab-Amendments Nos 1\_2\_3 to Asset Purchase Agreement\_12-5-2003 & 12-19-2003#page21.tif

source=ISI Brands-IdeaSphere-Twin Lab-Amendments Nos 1\_2\_3 to Asset Purchase Agreement\_12-5-2003 & 12-19-2003#page22.tif

ISI Brands Inc. files this release based on the following facts:

On August 7, 1996, Chemical Bank recorded a security interest against approximately 250 marks owned by Twin Laboratories Incorporated. On May 25, 2001 several individuals (the "Blechmans") recorded a security interest against approximately 370 marks owned by Twin Laboratories Inc.

On September 4, 2003, Twin Laboratories Inc. filed bankruptcy in the Southern District of New York, Case No 03-11564 (a jointly administered case with other related entities). For purposes of this narrative, Twin Laboratories Inc. and the other related entities are referred to as the "Debtors."

On that same day, the Debtors entered into an Asset Purchase Agreement (the "APA") with IdeaSphere, Inc. and TL Acquisition Corp (along with ISI Brands Inc., "Twinlab").

On October 30, 2003, the Bankruptcy Court approved the sale free and clear of all liens (including those of Chemical Bank and the Blechmans) under Section 363 of the Bankruptcy Code.

In December 2003, the Debtors entered into three Amendments to the APA with Twinlab: Amendment No. 1 dated December 5, 2003; Amendment No. 2 dated December 19, 2003; and Amendment No. 3 dated December 19, 2003. On December 18, 2003, the Blechmans issued a release of security interest with respect to the Purchased Assets under the APA including the subject trademarks.

The subject acquisition/sale closed on December 23, 2003 with a Trademark Assignment delivered at closing from the Debtors to ISI Brands Inc.

Based on the sale under Section 363 of the Bankruptcy Code, the liens (security interests) of Chemical Bank and the Blechmans (as well as any other entities and individuals) were terminated as of December 23, 2003.

Requests for additional information or materials may be made directly to:

Mary Kay Shaver Varnum LLP 333 Bridge Street NW Grand Rapids, Michigan 49504 Main: (616) 336-6000

Fax: (616) 336-7000 Ref Doc No. 8999253

### TRADEMARK ASSIGNMENT

WHEREAS, Twinlab Corporation, a Delaware corporation and/or Twin Laboratories Inc., a Utah corporation, and/or Twin Laboratories (UK) Ltd., a company organized under the laws of the United Kingdom, (together, "ASSIGNORS") own the entire right, title and interest in and to trademark registrations identified on Schedule A attached hereto, and the underlying marks including the goodwill associated therewith (the "Trademarks");

WHEREAS, ISI Brands, Inc., a Michigan corporation having its principal place of business located at 3133 Orchard Vista Drive, SE, Grand Rapids, MI 49546 ("ASSIGNEE") is desirous of acquiring all of ASSIGNOR'S right, title and interest in and to the Trademarks, including, without limitation, all registrations therefor, in the United States of America, its territories and possessions and worldwide; and

WHEREAS, ASSIGNORS and ASSIGNEE are parties to an Asset Purchase Agreement, dated as of September 4, 2003 (the "Asset Purchase Agreement"), by and among ASSIGNORS and ASSIGNEE, pursuant to which ASSIGNORS have agreed to sell and ASSIGNEE has agreed to purchase the entire right, title and interest to the Trademarks.

NOW THEREFORE, for good and valuable consideration, the receipt and legal sufficiency of which is hereby acknowledged, ASSIGNOR hereby sells, assigns, transfers and conveys to the ASSIGNEE, its successors and assigns, all of ASSIGNOR'S right, title and interest in and to the Trademarks, including, without limitation, all registrations therefor, together with the goodwill of the business symbolized by the Trademarks being assigned, in the United States of America, its territories and possessions, and worldwide, all causes of action for any and all past infringements of the rights being assigned and the right to collect and retain any proceeds therefrom.

ASSIGNORS HEREBY authorize and request the Commissioner of Patents and Trademarks of the United States, and any Official of any country or countries foreign to the United States whose duty it is to register trademarks on applications as aforesaid, to record this Assignment, in accordance with the terms of this instrument.

IN WITNESS WHEREOF, ASSIGNOR has caused this Assignment to be executed as of the date set forth below.

	I WINLAB CORPORATION
	Stuber )
Dated:	By: Jos Menman
	Nam <b>k</b> :
	Title:
STATE OF New york)	
STATE OF New york ) ss:	
On this 18th day of Decamber 200	63 before me personally appeared Ross Blechmen to me
"nersonally known, who, being duly sworn, did so	ay that he is the of [ASSIGNOR] and that he duly
executed the foregoing instrument for and on be	chalf of [ASSIGNOR], being duly authorized to do so and that said
individual acknowledged said instrument to be th	ne free act and deed of said corporation.
1 1 1 Monation	
Plates Deliner	
Notary Public; State of New York	
No. 01TO5053749	TWIN LABORATORIES INC.
Qualified in New York County Commission Expires December 26, 2005	
Dated:	By Jon Blukmin
	Name:
	Title:
COUNTY OF New York ). ss	
COUNTRY OF Me. CAROCK	S:
COUNTY OF JOSEPH )	
oft.	20 11 1
On this 1 day of Vecamber, 20	before me personally appeared Ross Blachman, to me
personally known, who, being duly sworn, did s	say that he is the of [ASSIGNOR] and that he duly
executed the foregoing instrument for and on be	ehalf of [ASSIGNOR], being duly authorized to do so and that said
individual acknowledged said instrument to be the	he tree act and deed of said corporation.
Judith Compairs	
Notary Public	
JUDITH TOMPKINS	
Notary Public, State of New York	
No. 01TO5053749  Qualified in New York County	
Commission Expires December 26, 2005	

KU3 2308551

	TWIN LA	3ORATORI	ES (UK) LTD	).	
Dated:	Ву: 10	no Bless	from	·	
	Nam <b>d</b> :		•		•
	Title:				
STATE OF New York ).				•	
	ss: · ·				. •
COUNTY OF New York					
	•	•		<i>.</i> ·	
ich o	××3		0.55	Bl. I.	•.
On this 18th day of December 2	before me	personally app	cared Nouve	Jech Mas	) to me
personally known, who, being duly sworn, did	say that he is the		of [ASSIGN	OR] and that	he duly
executed the foregoing instrument for and on t	pehalf of [ASSIG	NOR], being d	iuly authorized t	o do so and t	hat said
-individual acknowledged said instrument to be	the free act and d	eed of said cor	poration,		
a lett Santin				•	
Gudeth comprises	,			- · ·	
Notary Public		•	5.		. • .
JUDITH TOMPKINS	,				
Notary Public, State of New York					
No. 01TO50537491	•	•	• :		
Qualified in New York County	02				
Commission Expires December 26, 20	OD .				

### SCHEDULE <u>A</u>

COUNTRY	MARK	REGISTRATION DATE	REGISTRATION NUMBER
Argentina	Alvita	02/23/00	1776083
Argentina	Alvita	07/07/99 .	1743574
Argentina	Amino Fuel	12/30/94	1546528
Argentina ·	Gainers Fuel	01/31/94	1499959
Argentina	Nature's Herbs	07/13/95	1569085
Argentina	Ripped Fuel	01/31/94	1499960
Argentina	Twinlab	07/25/95	1569976
Argentina	Twinlab	01/31/94	1499961
Argentina	Ultra Fuel	08/10/95	1571871
Australia	Alvita	11/23/98	713372
Australia	Amino Fuel	04/18/98	554141
Australia	Chromic Fuel	04/18/98	.554142
Australia	Fuel	04/18/98	554143
Australia	GH Fuel	04/18/98	554140
Australia	Nature's Herbs/A Twinlab Division	11/17/00	782464
Australia	Twin-Labs/ Twin Labs/Twinlabs	09/29/02	712941
Australia	Twinlab	03/15/01	712942
Austria	Twinlab	02/7/01	134597
Barbados	Twinlab	05/08/98	81/6667
Benelux	Amino Fuel	10/12/99	468439
Benelux	Amino Fuel 2000	07/27/00	483456
Benelux	Anabolic Fuel	10/12/99	468440
Benelux	Diet Fuel	10/12/99	468087
Benelux	Energy Fuel	05/02/01 .	674907
Benelux	Gainers Fuel	10/12/99	468285
Benelux	Hydra Fuel	05/01/92	503159

KI.3:2308550.2

COUNTRY	MARK	REGISTRATION DATE	REGISTRATION NUMBER
Benelux	Nature's Herbs	08/07/00	485622
Benelux ·	Phos Fuel	07/27/00	482613
Benelux	Power Fuel	07/27/00	482612
Benelux	Twinlab	05/10/98	445671
Benelux	Ultra Fuel	10/12/99	468438
Benelux	Vita Fuel	10/12/99	468540
Bermuda	Twinlab	05/28/99	21742
Bolivia	Alvita	01/22/99	70174
Bolivia	Alvita	01/22/99	70173
Bolivia	Nature's Herbs	03/10/00	77471
Bolivia	Twinlab	03/30/93	C/54568
Brazil	Alvita	01/11/94	817617191
Brazil	Alvita	06/25/96	817178347
Brazil	BCAA Fuel 3000	01/27/00	821933892
Brazil	Carbo Fuel	01/23/92	816563632
Brazil	Creatine Fuel	06/27/01	823378098
Brazil	Diet Fuel	01/23/92	816563624
Brazil	Energy Fuel .	12/03/99	828139772
Brazil	Energy Fuel	12/03/99	821839780
Brazil	Gainers Fuel	01/23/92	816563659
Brazil	Hydra Fuel	03/12/96	817678530
Brazil	Nature's Herbs A Twinlab Company	01/28/98	820474738
Brazil	Opti Fuel	01/23/92	816563640
Brazil	Ripped Fuel	06/27/01	823378101
Brazil	Soy Sensations	06/27/01	823378055
Brazil	Truherbs (Special Form)	12/28/99	821914367
Brazil	Twin Laboratories Inc.	06/27/01	823378080
Brazil	Twinfast	09/2:1/93	816507287

COUNTRY	MARK	REGISTRATION DATE	REGISTRATION NUMBER
Brazil	Twinlab :	12/04/90	851903324
Brazil ·	Twinlab Energy Fizz	06/27/01	823378063
Brazil	Twinlab Truherbs (Special Form)	12/28/99	.821914367
Brazil	Twinlab.com.br	Domain name owned b Arnaud & Scatamburlo	
Brazil	Ultra Fuel	12/20/93	817608680
Brazil	Whey fuel	06/27/01	823378071
Canada	Alvita	06/18/98	TMA496423
Canada	B-12 Dots	10/23/92	TMA403831
Canada	Calmag + D	10/09/92	TMA403473
Canada	Caro Caps	10/09/92	TMA403471
Canada	Dale Alexander	10/26/90	TMA374836
Canada	Effervescent C with Minerals	06/10/94	TMA428439
Canada	Endurance	10/09/92	TMA403474
Canada	Fibersol	10/18/91	TMA389399
Canada	Mega Lecithin (E) Plus	10/09/92	TMA403475
Canada	Mini B	10/09/92	TMA403472
Canada	Nature's Herbs	09/16/02	TMA567458
Canada	Nature's Herbs	. 03/20/87	TMA324886
Canada	Power Herb	03/27/02	TMA325148
Canada	Ripped	01/27/03	TMA574409
Canada ·	Ripped Fuel	10/21/02	TMA569203
Canada	Stressmates	10/25/91	TMA389646
Canada	Twinlab	04/26/91	TMA383406
Canada	Twinlab Buffered Vitamin C plus Magnesium	02/05/97	TMA470567
Canada	Ultra Fuel	11/17/95	TMA450069
Chile	Alvita	-06/14/99	542532

COUNTRY	MARK	REGISTRATION DATE	REGISTRATION NUMBER
Chile	Nature's Herbs A Twinlab Company	03/18/98	507563
Chile	Ripped Fuel	11/22/01	609534
Chile	Twinlab	06/19/01	597872
Chile	Twinlab.cl	04/03/02	None
China	Alvita	10/14/98	1214277
China	Alvita	11/14/98	1223364
China	Alvita in Chinese Characters	08/28/99	1307781
China	Alvita in Chinese Character	09/28/99	1319694
China	Fuel	08/28/99	1307783
China	Power-Herbs in Chinese Character	08/28/99	1307782
China	Remeteas	08/07/99	1300284
China	Remeteas	10/07/99	1321462
China	Remeteas in Chinese characters	08/28/99	1307784
China	Remeteas in Chinese characters	09/28/99	1319693
China	Twinlab	05/28/97	1014920
China	Twinlab (Tian Lai)	08/28/99	1307776
China	Twinlab (Tian Lai)	10/07/99	1321461
China	Twinlab in Chinese characters I	08/28/99	1307779
China	Twinlab in Chinese characters III	10/07/99	1321463
China <sub>,</sub>	Twinlab in Chinese characters III	08/28/99	1307780
China (Taiwan)	Alvita	05/16/97	762238
China (Taiwan)	Nature's Herbs	05/16/00	891760
China (Taiwan)	Nature's Herbs/ A Twinlab Division	08/01/01	952525

COUNTRY	MARK	REGISTRATION DATE	REGISTRATION NUMBER
China (Taiwan)	Twin Labs	02/16/97	748551
China (Taiwan)	Twinlab	08/16/96	724412
Colombia	Nature's Herbs	03/26/98	209964
Colombia	Twinlab	11/19/99	223106
Costa Rica	Alvita	03/18/97	100423
Costa Rica	Twinlab	01/31/96	94474
Cyprus	Twinlab	04/29/99	36456
Czech Republic	Alvita	05/27/98	209474
Czech Republic	Nature's Herbs/ A Twinlab Company	03/23/00	223338
Czech Republic	Twinlab	05/27/98	209475
Denmark	Energy Fuel	06/28/00	2000 2994 VR
Denmark	Twinlab	10/18/91	1991 6951 VR
Ecuador	Alvita	02/09/00	301/00
Ecuador	Alvita	01/22/98	378/98
Ecuador	Nature's Herbs	01/22/98	377/98
Ecuador	Twinlab	01/26/98	435/98
Ecuador	Twinlab	01/26/98	434/98
. Egypt	Alvita	07/17/00	.102445
Egypt	Alvita	02/17/02	102477
Egypt	Twinlab	11/25/01	81499
El Salvador	Alvita	01/09/98	239BOOK65
El Salvador	Alvita	06/29/00	135BOOK107
El Salvador	Twinlab	04/15/96	173
European Community	Soy Sensations	04/03/02	1751254
European Community	Twinlab Energy Fizz	08/20/01	1751296
Finland .	Energy Fuel	09/29/00	218781
Finland	Nature's Herbs A Twinlab Company	06/15/99	214338
Finland	Twinlab	01/20/93	124463

COUNTRY	MARK	REGISTRATION DATE	REGISTRATION NUMBER
France	Diet Fuel	07/31/00	1607032
France	Energy Fuel	11/16/99	99823482
France	Fuel	02/20/00	1738959
France	Muscular Development Fitness Health Logo	07/24/96	96/635797
France	Twinlab	06/02/98	1651607
Germany	Diet Fuel	04/11/94	1190091
Germany	Energy Fuel	02/22/01	39969745
Germany	Power Fuel	04/11/94	1190088
Germany	Twinlab	06/7/93	1187882
Germany	.Ultra Fuel	04/11/94	1190089
Greece :	Twinlab	06/27/00	99596
Guatemala	Alvita	02/02/03	122006
Guatemala	Alvita	11/05/00	107177
Hong Kong	Alvita in Chinese characters	01/12/00	687/2000
Hong Kong	Fuel	10/10/00	B13444/2000
Hong Kong	Fuel in Chinese characters	07/13/99	8747/1999
Hong Kong	Nature's Herbs in Chinese characters	01/25/00	B1586/2000
Hong Kong	Power-Herbs	10/27/99	B13072/1999
Hong Kong	Power-Herbs in Chinese characters	08/24/99	10372/1999
Hong Kong	Remeteas	03/10/99	2874/1999
Hong Kong	Remeteas	03/10/99	2875/1999
Hong Kong	Remeteas in Chinese characters	05/09/00	B06588/2000
Hong Kong	Twiniab	08/11/97	11383/96
Hong Kong	Twinlab (Tian Lai)	01/12/00	688/2000
Hong Kong	Twinlab (Tian Lai)	01/12/00	689/2000

COUNTRY	MARK	REGISTRATION DATE	REGISTRATION NUMBER
Hong Kong	Twinlab in Chinese characters III	07/15/99	9031/1999
Hong Kong	Twinlab in Chinese characters III	07/15/99	9032/1999
Hungary	Alvita	06/26/98	151617
Hungary	Nature's Herbs	09/11/97	146385
Hungary	Twinlab	04/02/98	150516
Iceland	Twinlab .	09/25/02	1001/1992
Indonesia	Alvita	09/30/97	394435
Indonesia	Alvita	09/30/97	394619
Indonesia	Nature's Herbs/ A Twinlab company	01/07/99	421668
Indonesia	Twin Labs	09/01/95	345350
Indonesia	Twinlab .	09/24/96	368397
Ireland	Carbo Fuel	12/11/98	150567
Ireland	MCT Fuel .	12/11/98	150566
Ireland	Twinlab	10/09/94	125290
Israel	Alvita	04/05/98	106543
Israel	Nature's Herbs	07/06/98	106550
Israel	Twinlab	12/11/95	71083
Italy	Anabolic Fuel	10/18/93	605303
Italy	Amino Fuel	10/18/93	605301
Italy	Carbo Fuel	10/18/93	305304
Italy	Chromic Fuel	10/18/93	605305
Italy	Diet Fuel	10/18/93	605306
Italy	Diet Fuel	06/17/94	625428
Italy	Gainers Fuel	10/18/93	605307
Italy	GH Fuel	10/18/93	605308
Italy	Herba Fuel	10/18/93	605309
Italy	Hydra Fuel	01/26/96	666975
Italy	Mass Fuel	01/26/96	666974

COUNTRY	MARK	REGISTRATION DATE	REGISTRATION NUMBER
Italy	MCT Fuel	10/18/93	605310
Italy	Opti Fuel	10/18/93	605311
Italy	Phos Fuel	10/18/93	605312
Italy	Power Fuel	10/18/93	605313
Italy	Pro Fuel	08/13/00	605314
Italy	Twinlab	11/23/91	.554518
Italy	Twinlab	04/30/02	864977
Italy	Ultra Fuel	10/18/93	605315
Italy	Vita Fuel	10/18/93	605316
Japan	Amino Fuel	08/31/92	2445872
Japan	Fuel	08/18/00	4410813
Japan	Fuel	03/01/02	4548170
Japan	Nature's Herbs A Twinlab Division	06/01/01	4479420
Japan	Power Herbs	06/15/01	4482903
Japan	Power Herbs	03/01/02	4548169
Japan	Soy Sensations	06/01/01	4479525
Japan	Twinlab	05/29/92	2412821
Japan	Twinlab	07/27/01	4495012
Japan	Twinlab	06/15/01	4482904
Japan	Twinlab Energy Fizz	06/01/01	4479524
Japan	Twinlab Truherbs (Special Form)	03/03/00	4365402
Japan	Truherbs (Special Form)	03/01/02	4547713
Korea (South)	Alvita	08/07/98	414844
Korea (South)	Nature's Herbs A Twinlab Company	07/29/98	413315
Korea (South)	Nature's Herbs/ A Twinlab Company	12/02/98	431877
Korea (South)	Twinlab	07/25/00	197157 .

COUNTRY	MARK	REGISTRATION DATE	REGISTRATION NUMBER
Malaysia	Amino Fuel	07/16/98	91003624
Malaysia	Chromic Fuel	07/16/98	91003625
Malaysia	Fuel	07/16/98	91003627
Malaysia	GH Fuel	07/16/98	91003626
Malaysia	Migracin	08/27/97	90/05670
Malaysia	Willoprin	08/27/97	90/05669
Mexico	Alvita	01/27/98	568167
Mexico	Amino Fuel	11/30/94	382542
Mexico	Amino Fuel 2000	08/06/95	388908
Mexico	Anabolic Fuel	11/30/94	382541
Mexico	Carbo Fuel	11/29/94	384806
Mexico	Chromic Fuel	11/29/94	391329
Mexico.	Coles Free	09/21/99	623703
Mexico	Diet Fuel	11/30/94	382539
Mexico	Energy Fuel	01/25/00	638358
Mexico	Gainers Fuel	11/29/94	384807
Mexico	GH Fuel	11/30/94	382540
Mexico	Infant Care	11/29/94	397273
Mexico	MCT Fuel	08/06/95	388907
Mexico	Nature's Herbs	11/08/00	443021
Mexico	Opti Fuel	11/30/94	384469
Mexico	Phos Fuel	08/06/95	388906
Mexico	Power Fuel	08/06/95	388905
Mexico	Pro Fuel	08/06/95	388903
Mexico	Soy Sensations	01/30/01	685258
Mexico	Thermolift	12/20/00	453758
Mexico	Twinfast	11/29/94	384804
Mexico	Twinlab	06/04/00	399474
Mexico	Twinlab	07/04/99	445164
Mexico	Ultra Fuel	08/06/95	388904

COUNTRY	MARK	REGISTRATION DATE	REGISTRATION NUMBER
Mexico	Ultra Fuel	02/26/96	517897
Mexico	Vita Fuel	11/29/94	384805
New Zealand	Amino Fuel	04/18/98	B209589
New Zealand	Chromic Fuel	04/18/98	B209591
New Zealand	GH Fuel	04/18/98	B209590
New Zealand	Nature's Herbs Power-Herbs (Stylized)	07/01/01	B238497
New Zealand	Nature's Herbs/ A Twinlab Company	12/15/98	280316
New Zealand	Twinlab	06/16/96	193987
Norway	Energy Fuel	11/16/00	205598
Norway	Twinlab	10/22/02	152852
Oman .	Twinlab	07/29/01	9626
Panama	Alvita	07/09/97	88616
Panama	Nature's Herbs	05/25/99	88795
Panama	Twinlab	03/17/94	63392
Paraguay	Alvita	08/19/97	. 195914
Paraguay	Nature's Herbs	10/26/98	208060
Paraguay	Twinlab	06/26/99	216699
Peru	Alvita	02/20/97	33652
Peru	Alvita	12/16/96	31804
Peru	Nature's Herbs	11/25/96	31258
Peru	Twinlab	06/24/93	549
Philippines	Twinlab	02/19/91	50011
Portugal	Alvita	03/05/97	317847
Portugal	Power-Herbs Nature's Herbs	02/12/93	271308
Portugal	Twinlab	11/20/92	266975
Puerto Rico	Nature's Herbs	01/30/98	39935
Puerto Rico	Twinlab	03/04/93	32128

- COUNTRY	MARK	REGISTRATION DATE	REGISTRATION NUMBER
Qatar	Twinlab	08/30/00	8300
Qatar	Twinlab & Arabic	08/30/00	8299
Russia	Muscular Development Logo	05/15/98	164411
Saudi Arabia	Alvita	08/19/97	409/36
Saudi Arabia	Alvita	09/01/97	410/62
Saudi Arabia	Twinlab	11/05/94	324/40
Singapore	Alvita	07/25/96	T96/07759D
Singapore	Amino Fuel	01/05/94	85/94
Singapore	Fuel	04/20/01	T91/0404031
Singapore	Nature's Herbs	07/25/96	B96/07757H
Singapore	Twinlab	10/02/94	T87/04819F
Singapore	Twinlab GH Fuel	01/05/94	86/94
Singapore	Willowprin	08/21/97	5829/90
South Africa	Alvita	03/29/00	96/10012
South Africa.	Alvita	08/05/99	96/10013
South Africa	Nature's Herbs	01/10/00	96/10014
South Africa	Twinlab	06/15/99	89/5302
Spain .	Energy Fuel	06/05/00	2279154MX
Spain	Twinlab	04/05/91	1291087M2
Spain	Twinlab Herba Fuel	09/05/94	1783263M2
Spain	Twinlab Amino Fuel	04/05/94	1782421M4
Spain	Twinlab Diet Fuel	05/20/94	1790762M4
Spain	Twinlab Twin Laboratories Inc. New York	04/05/91	1291087M2
Sweden	Energy Fuel	10/19/01	349368
Sweden	Twinlab	03/31/94	256913
Switzerland	Twinlab	11/23/89	373155
Thailand	Alvita	08/24/98	KOR76754
Thailand	Alvita	01/07/99	TM84715

COUNTRY	MARK	REGISTRATION DATE	REGISTRATION NUMBER
Trinidad & Tobago	Alvita	03/30/99	26013
Trinidad & Tobago	Alvita	06/28/02	26012
Trinidad & Tobago	Nature's Herbs A Twinlab Company	12/27/00	27514
Trinidad & Tobago	Twinlab	10/28/01	17146
Turkey	Twinlab	09/20/99	116080
United Arab Emirates	Nature's Herbs/ A Twinlab Division	04/04/02	31237
United Arab Emirates	Twinlab	06/24/98	16241
United Kingdom	Chromic Fuel	04/19/98	1461937
United Kingdom	Energy Fuel	02/16/01	2213595
United Kingdom	Hydra Fuel	03/15/96	2016661
United Kingdom	Twinlab	12/07/95	1366182
United Kingdom	Twinlab Ultra Fuel	10/28/00	1551867
Uruguay	Nature's Herbs	05/04/01	296844
Venezuela	Twinlab	08/19/94	F-164245
Vietnam	Nature's Herbs	02/03/99	29535

COUNTRY	MARK	APPLICATION DATE	APPLICATION NUMBER
Canada	Chromic Fuel	11/06/01	112168100
Canada	Creatine Fuel	11/06/01	112168300
Canada -	Energy Fuel	11/08/99	103537200
Canada	Ever-Slender	11/6/01	112168000
Canada	Joint Rescue	11/06/01	112168400
Canada	Ocuguard	11/06/01	112167900
Canada	Metabolift	11/06/01	112168200
Canada	Soy Sensations	06/28/00	106549700
Canada	Today's Woman	06/28/00	106549500
Canada	Twinlab Energy Fizz	06/28/00	106549600
Canada	Twinlab Fuel	05/23/00	106020400

KL3:2308550.2

COUNTRY	MARK	APPLICATION DATE	APPLICATION NUMBER
Chile	Diet Fuel	01/25/00	474010
Chile	Fuel	05/25/00	487640
Colombia	Alvita	09/24/96	96/040821
Colombia	Alvita	08/01/96	96/040823
European Community	Twinlab Aquanutri	02/21/03	3063682
Greece	Energy Fuel	11/12/99	142301
Guatemala	Nature's Herbs	08/19/96	6369
Guatemala	Twinlab	08/16/96	6336
Guatemala	Twinlab	08/16/96	6337/96
Italy	Diet Fuel	01/07/02	66 2002 MI
Italy	Diet Fuel	08/07/00	9239 2000 MI
Italy	Energy Fuel	01/05/00	82 2000 MI
Italy	Gainers Fuel	08/08/00	9238 2000 MI
Italy	Hydra Fuel	01/23/03	588 2003 MI
Italy	Mass Fuel	01/31/03	979 2003 MI
Italy	Power Fuel	08/07/00	9236 2000 MI
Italy	Ultra Fuel	08/07/00	9237 2000 MI
Japan	Alvita	9/18/97	H09-158868
Malaysia	Twinlab	01/04/89	89/00019
Mexico	Alvita	8/1/96	269886
Portugal	Energy Fuel	11/18/99	341288
Portugal	Nature's Herbs	6/21/96	317846.
Qatar	Alvita & Arabic	7/29/96	15546
Qatar	Alvita & Arabic	7/29/96	15547
Qatar	Nature's Herbs and Arabic	6/14/97	17021
Saudi Arabia	Nature's Herbs	11/30/96	36246
Uruguay	Alvita	7/16/97	296843
Venezuela	Alvita	10/18/96	17318/96
Venezuela	Alvita	10/16/96	17319/96

COUNTRY	MARK .	APPLICATION DATE	APPLICATION NUMBER
Venezuela	Nature's Herbs	10/16/96	17320/96

# UNITED STATES

TRADEMARK	PEGNORUS	REG DAIL
Canadian Natur-Tussin	2,143,977	3/17/98
CARIBBEAN BREATHABILITEA	2,150,373	4/14/98
CHINESE MELLOW MAGIC	2,111,834	11/11/97
DHEA FUEL	2,110,713	11/4/97
FEMALE FUEL	2,109,045	10/28/97
MALE FUEL	2,057,046	4/29/97
RIPPED FUEL	1,773,392	5/25/93
MCT FUEL	1,597,123	5/22/90
GULF STREAM PROSTI-BREW	2,103,521	10/7/97
HERBAL REMETEAS	2,139,522	2/24/98
REMETEAS	2,090,920	8/26/97
LANCASTER COUNTY MIGRA-WONDER	2,055,356	4/22/97
PACIFIC COAST SLIMMER	2,055,355	4/22/97
HIGHLAND LULLABY	2,055,354	4/22/97
HIMALAYAN SENNALAX	2,252,270	6/15/99
MANCHURIAN BRAIN BLEND	2,131,456	1/20/98
IVORY COAST SIP-N-C+	2,181,810	8/18/98
Jamaican Digesti Brew	2,110,495	11/4/97
JOINT FUEL	2,066,656	6/3/97
YOHIMBE FUEL	2,064,764	5/27/97
MAXILIFE BRAIN PROTECTOR	2,109,039	10/28/97
Maxilife Cardio Protector	2,107,224	10/21/97
MAXILIFE COLON PROTECTOR	2,107,226	10/21/97
MAXILIFE JOINT PROTECTOR	2,109,038	10/28/97
MAXILIFE PHYTONUTRIENT PROTECTOR	2,170,812	7/7/98
MAXILIFE PROSTATE PROTECTOR	2,112,214	11/11/97
MEDITERRANEAN DETOXITEA	2,151,930	4/21/98
MUSCULAR DEVELOPMENT	2,176,192	7/28/98
Nature's Herbs	1,257,337	11/15/83
SCANDINAVIAN WINTER COMFORT	2,174,350	7/21/98
SCOTTISH THROAT MYST	2,176,126	7/28/98
TIBETAN FOREST MASCULINITEA	2,197,017	10/20/98
THE WORD FOR GOOD HEALTH	1,892,911	5/9/95

KL3:2308550.2

TRADEMARK		A REGEDARE
KIC FUEL	1,848,966	8/9/94
TWINLAB.COM	2,267,492	8/3/99
WHEY FUEL	2,144,516	3/17/98
ALLERDOPHILUS	1,382,984	2/18/86
ALLERGY-C	1,380,032	1/28/86
ALLERGY A CAPS	1,748,091	1/26/93
ALLERGY D CAPS	1,746,752	1/19/93
ALLERGY FIGHTERS	1,428,089	2/10/87
ALVITA	1,759,398	3/23/93
AMINO FUEL	1,437,049	4/21/87
AMINO FUEL STACK	2,209,776	12/8/98
ANABOLIC FUEL	1,509,858	10/25/88
Animal Friends	1,502,771	9/6/88
ANTI-CATABOLIC FUEL	1,920,669	9/19/95
ANTIOXIDANT FUEL	1,885,533	3/21/95
T-AMINO ·	1,123,517	8/7/79
B-12 Dots	1,811,803	12/21/93
B-BITES	1,451,744	8/11/87
BIO-FORMED	1,167,990	9/8/81
B-SLIM	1,428,986	2/17/87
CAL-QUICK	1,333,948	5/17/85
CARNI FUEL	1,847,227	7/26/94
CARBO FUEL	1,388,002	4/1/86
POLYNESIAN PMS RESCUE	2,227,172	3/2/99
ALPINE-PEAK VISIBILITEA	2,168,616	6/23/98
BRIGHTON PICK-TEA-UP	2,168,617	6/23/98
ENDURANCE QUICK FIX	1,541,164	5/30/89
EyL	1,496,507	7/19/88
FIBERSOL	1,281,261	6/12/84
GAINERS FUEL	1,508,880	10/18/88
GLUTAMINE FUEL	1,954,899	2/6/96
Guaraid	1,461,650	10/20/87
HAIR FACTORS	1,306,747	11/27/84
Herba Fuel	1,487,374	5/10/88
HUMAN MACHINE OIL	1,597,121	5/22/90
HYDRA FUEL	1,655,100	8/27/91
HYPO E-CAPS	1,327,943	4/2/85
IMMUNE PROTECTORS	1,428,090	2/10/87
LIGHTS OUT	1,347,148	7/9/85
LIQUI-E	1,417,161	11/18/86
Liqui-K	1,295,575	9/18/84

TURADEMARK	je lejertenoeti	
LPP	1,121,723	7/10/79
LPP(STYLIZED)	915,606	6/29/71
MASS FUEL	1,692,371	6/9/92
MAXILIFE (STYLIZED)	1,321,725	2/26/85
MAXILIFE	1,301,149	10/23/84
CHER-AMINO	1,119,092	5/29/79
CHOLINE COCKTAIL	1,912,216	8/15/95
CHROMIC FUEL	1,583,283	2/20/90
CREATINE FUEL	1,864,175	11/22/94
CREATINE FUEL COCKTAIL	2,145,871	3/24/98
DMAE-H3	1,279,590	5/29/84
ENDURANCE	1,148,634	3/24/81
CELLMINS	1,773,390	5/25/93
DIET FUEL	1,544,159	6/20/89
DIET FUEL	1,782,373	9/20/93
SPORT FUEL	1,851,871	8/30/94
STRESSMATES	1,471,933	1/12/88
SUPERTWIN	1,480,417	3/15/88
SUPER VANADYL FUEL	1,954,900	2/6/96
THE LEADER IN SPORTS NUTRITION	1,692,361	6/9/92
TWINFAST	1,603,361	6/26/90
TWINLAB	1,298,349	10/2/84
TWINLAB	1,577,237	1/16/90
ULTRA FUEL	1,540,057	5/23/89
VEGE FUEL	1,781,039	7/13/93
VITA FUEL	1,501,862	8/30/88
VITA QUICK & DESIGN	1,192,419	3/23/82
YEAST FIGHTERS	1,369,779	11/12/85
MEGA HMB FUEL	2,188,532	9/8/98
HMB FUEL	2,185,075	8/25/98
ASIAN HERB-O-PAUSE	2,250,182	6/1/99
GIBRALTAR CARDIO BLEND	2,202,573	11/10/98
FEED THE MACHINE	2,393,807	10/10/00
FEELIN' FIT FITNESS PAKS	1,714,945	9/15/92
METABOLIFT	1,782,018	7/13/93
MEMORY FUEL	1,827,161	3/22/94
Moistique	1,325,343	3/19/85
NEUROVITES	1,347,147	7/9/85
NUTRABOLICS	1,511,454	11/8/88
NUVEG	1,806,666	11/23/93
OCUGUARD	1,784,724	7/27/93

		resent de la maintaine de la maintaine de la company de la
TRADEMARK		PER RECEDITED
GENTLE C-250	1,777,621	6/22/93
OPTI FUEL (STYLIZED)	1,584,280	2/27/90
OPTI-CAL	1,763,989	4/13/93
Oxyguard	1,402,801	7/29/86
Oxy-Quenchers	1,247,535	8/9/83
Phos Fuel	1,698,746	7/7/92
Power Fuel	1,709,127	8/18/92
RADICAL FIGHTERS	1,209,619	12/18/84
NAC FUEL	2,141,107	3/3/98
TRUHERBS	2,206,215	11/24/98
ALC FUEL	2,350,510	5/16/00
GROWTH FUEL	2,183,051	8/18/98
Mega Glutamine Fuel	2,186,898	9/1/98
HEALTHLAB	2,220,273	1/26/99
GLUCOSAMINE FUEL	2,144,791	3/17/98
PYRUVATE FUEL	2,142,919	3/10/98
TRIBULUS FUEL STACK	2,144,793	3/17/98
TRIBULUS FUEL	2,144,792	3/17/98
Tri-Boron	1,500,151	8/16/88
XTRA FUEL	2,217,212	1/12/99
The Herbal Village	2400369	10/31/00
Catechin 90-Power	2341186	04/11/00
Aqua Slim	2367558	07/18/00
PMS Power	2367558	09/28/99
Bitter Melon Power	2276023	09/07/99
Mediterranean Phytonutrient Power	2344587	04/25/00
Maitake Power	2249770	06/01/99
Modified Citrus Pectin Power	2288620	10/26/99
Super Phenolic Fruit Power	2288619	10/26/99
Black Cohosh-Power	2303387	12/28/99
Olive Leaf-Power	2256685	6/29/99
Herbal Power	2527788	01/08/02
Nature's Herbs	2284606	10/12/99
Celery Seed-Power	2238233	04/13/99
Horse Chestnut Power	2245469	05/18/99
Flower Power	2406599	11/21/00
DHEA Power	2114188	11/18/97
Stevia-Power	2093223	09/02/97
Super Prostate Power	2144192	03/17/98
Beta Sitosterol-Power	2144191	03/17/98
Yohimbe-Power	2093222	09/02/97
Phosphatidylserine-Power	2091354	08/26/97

TRADEMARK TO THE		
Mexican Wild Yam-Power	2144190	03/17/98
Grape Seed-Power	2177821	08/04/98
Nettle Power	2069880	06/10/97
Kava Kava Power	2071861	06/17/97
Curcumin Power	2261309	07/13/99
Carotene Power	2355498	06/06/00
Propolis Power	2385847	09/12/00
Nature's Goodness	2576369	06/04/02
Put Your Faith in Nature and Trust in	2449354	05/08/01
Nature's Herbs	2449334	05,00,02
	2037034	02/11/97
Elderberry Power	2022205	12/10/96
Skin-Power	2024032	12/17/96
Phyto Estrogen Power	2293834	11/23/99
Cat's Claw Power	2066480	06/03/97
Dong-Quai Power	2221040	02/02/99
Power Herbs		10/24/95
Joint Power	1929844	09/19/95
Echinacea Golden Seal Power	1919220	08/29/95
Garlic Power	1914499	10/31/95
Primrose Power	1931234	09/12/95
Golden Seal Power	1917588	08/29/95
Lysine Power	1914497	11/21/95
Cranberry Power	1936823	08/15/95
What Nature Had in Mind for You	1911051	
Shitake-Power	1837333	05/30/00
Pygeum-Power	1837697	05/30/00
Women's Care	1841637	06/28/94
Reishi Power	1833634	05/03/94
DGL Power	1833633	05/03/94
Saw Palmetto-Power	1833632	05/03/94
Hawthorn-Power	1833631	05/26/00
Black Currant-Power	1833630	05/03/94
Artichoke Power	1833629	05/03/94
Phytopower	1840272	06/21/94
Anti-Catabolic Fuel	1920669	09/19/95
Power Extract	1818718	02/01/94
Fever Few Power	1783832	07/27/93
Herbal Male Vite	1794479	09/21/93
Herba Choline	1818699	02/01/94
Herbal Detox	1808760	12/07/93
Daily Fatigue	1799043	10/19/93
Gymnesyl	1771072	05/18/93
Phyllanthol	1831022	04/19/94

TRADEMARK		
I RADEMARK I FILL	1760400	04/06/93
Borage Power	1762400	12/22/92
Green Tea Power	1740831	03/15/94
Herba-Vites	1826809	
Veno Care	1740817	12/22/92
Inflam-aid	1740816	12/22/92
Superior Chinese Resist	1740814	12/22/92
Nature's Herbs The Perfect Balance of Tradition and Science	1892540	05/02/95
Nature's Herbs the Herbal Experts in Sports Nutrition	1853997	09/13/94
Astrex F-3	1746742	01/19/93
Advanced Women's Care	.1788745	08/17/93
Pollen Hearts	1770898	05/11/93
Milk-Thistle Power	1635260	02/19/91
Leading The Way To Better Health	1635258	02/19/91
Willowprin	1636639	03/05/91
The Power to Feel Your Best	1635257	02/19/91
Bilberry-Power	1632126	02/22/91
Echinacea-Power	1643312	05/07/91
Ginkgo Power	1632125	01/22/91
Migracin	1638118	03/19/91
Ginger Power	1635934	02/26/91
Skinny Formula	1635933	02/26/91
Femcleanse	1635931	02/26/91
Silica-Power	1635930	02/26/91
More of Nature's Goodness	1636638	03/05/91
Valerian Power	1635929	02/26/91
Desert Herb Combination	1632124	02/22/91
Citrus Power	1632890	02/29/91
St. John's Power	1652870	08/06/91
Certified Potency logo	1651283	07/23/91

TRADEMARK	APPLICATION OF THE STATE OF THE	APPLICATION ELL.
Soy Power	76/120966	9/1/00
The Herbal Millennium	75/141430	05/13/03

TRADEMAR	Z su za	
Nutussin	1483019	04/05/88
Allerin	1471921	01/12/88
Caprinol	1461620	10/20/87

KL3:2308550.2

TRADEMARK		DATE
Coldrin	1457970	09/22/87
Safe-Cap	1435268	04/07/87
Heart-Life	1536033	04/25/89
PMS. O.S.	1433521	03/24/87
PMS-Syn	1466895	12/01/87
Agua Trim	1551571	08/15/89
Quietime	1432678	03/17/87
Stresstop	1441740	06/09/87
Cardio-Care	1432677	03/17/87
Digestab	1441739	06/09/87
Imune-Gard	1408176	09/09/86
Natur-Pharma	1555434	09/12/89
Congest-Ease	1400070	07/08/86
Stress-Release	1364129	10/08/85
Ever-Slender	1374544	12/10/85
Gentle-Cleanse	1364128	10/08/85
Potency Power	1364127	10/08/85
Passion-Power	1364126	10/08/85
Good-Nite	1364125	10/08/85
Nature's Power	1364124	10/08/85
Power	1400866	07/15/86
Healthcare Naturals	1378779	01/21/86
Cold-Control	1402630	07/22/86
Allerelief	1331599	04/23/85
Power-Herb	1351273	07/30/85
Senna-Lax	1460560	10/13/87
Herbtrim	1305702	11/20/84
Vege Lax	1315339	01/22/85
Travel-case	1306793	11/27/84
Zest-Aid	1347126	07/09/85
Herbal Calm	1317572	02/05/85
Menstrual Ease	1345516	07/02/85
Diurtab	1304664	11/13/84
Arth Plus	1382959	02/18/86
Bronc Ease	1304663	11/13/84
Femchange	1304662	11/13/84
Winterspice	1234246	04/12/83
The Herb Master's Original Formulas	1234245	04/12/83
Nature's Author	1234244	04/12/83
Willowtab	1235143	04/19/83
Andro Fuel	2293618	11/16/99
Beyond Cholesterol	2443239	4/10/01
Birch Bark-Power	2377505	8/15/00

TRADEMARK	RECTAR SOLUTION	RES TO THE
	Logo (CO)	10/1 <b>2/9</b> 9
Blackberry Moon	2286698	3/20/01
Cardio Results	2437591	5/20/03
Carp-L Care	2718053	
Cholesterol-Power	2298211	12/7/99
Cholesterol Success	2652592	11/19/02
CLA Fuel	2476268	8/7/01
Creatine Fizz Fuel	2448371	5/1/01
Energy Fuel	2595723	7/16/02
FastBurn	2373811	8/1/00
FlexiLicious	2633220	10/8/02
Fuel Plex	2729521	6/24/03
Herbal Orchard	2308003	1/11/00
Joint Food	2594565	7/16/02
Lazy Lemontime	2375521	8/8/00
Lycosoy	2443427	4/10/01
Mega Soy	2293620	11/16/99
Nettle Root Power	251308	11/27/01
Oregano-Power	2493123	9/25/01
Peach Blush	2286700	10/12/99
Peppermint Breeze	2286699	10/12/99
Pro-Complex Formula	2381842	8/29/00
Raspberry Melody	2286701	10/12/99
Red Clover Power	2500727	10/23/01
Seastatin	2443374	4/10/01
Sharp Thinking	2317252	2/8/00
Strawberry Journey	2333398	3/21/00
Strength Fuel	2513423	11/27/01
Thermo 1	2507635	11/13/01
Today's Woman	2441580	4/3/01
Tomato-Power	2520644	12/18/01
Triple Whey Fuel	2293619	11/16/99
Twinlab Skin Fitness	2443320	4/10/01
Ultra3 Growth Fuel	2744188	7/29/03
Vita-Lok	2443031	4/10/01

TO THE	ATRIKANIO	TERRET APPLICATION TO BE TO SERVE THE SERVE TH
AM Protein Fuel	76/525807	5/4/03
Amino Complex	76/505739	4/11/03
Blood Pressure Control	76/532419	7/25/03
Cran Tract	76/519434	5/27/03
Diabalance	76/432885	Notice of Allowance issued 3/11/03

KL3:2308550.2

	Z ZPROTECTION Z	LAPRICATION
Ecdyburn	76/442812	Statement of Use filed 6/20/03
Ecdycut	76/442811	Statement of Use filed 6/20/03
MRP Nitrate 3 Fuel	. 76/505737	4/11/03
Carb Defense	76/373670	Notice of Allowance issued 2/17/02
Thermo Sport	76/261238	filed 5/24/01 - suspended
Spiru-Licious	76/519435	5/27/03
The Science Behind the Size	76/382263	Notice of Allowance issued 12/31/02
TwinSorb	76/526266	6/27/03
Vital Memory	76/525806	5/14/03
Xtreme Power Fuel	76/505738	4/11/03

# STATE REGISTRATIONS

STATE	MARK	REGISTRATION DATE	REGISTRATION NUMBER
Puerto Rico	Alvita	1/31/98	39032
Puerto Rico	Alvita	1/31/98	39031
Puerto Rico	Nature's Herbs	1/31/98	39935
Puerto Rico	Twinlab	8/9/93	32128

KL3:2308550.2

UNITED	STATES	BANK!	RUPTCY	<b>COURT</b>
SOUTHE	RN DIS	TRICT (	OF NEW	YORK

100

In re	:	
·	:	Chapter 11 Case No.
	:	
TWINLAB CORPORATION, et al.,	:	03-15564 (CB)
	_	

Debtors. : (Jointly Administered) :

ORDER PURSUANT TO SECTIONS 105, 363(b), (f), AND (m), 365, AND 1146 OF THE BANKRUPTCY CODE AND BANKRUPTCY RULE 2002 AUTHORIZING (I) THE SALE OF SUBSTANTIALLY ALL OF THE DEBTORS' ASSETS FREE AND CLEAR OF LIENS, CLAIMS AND ENCUMBRANCES TO IDEASPHERE, INC. AND TL ACQUISITION CORP., (II) THE ASSUMPTION AND ASSIGNMENT OF CERTAIN EXECUTORY CONTRACTS, AND (III) THE ASSUMPTION OF CERTAIN LIABILITIES

This matter is before the Court on the motion of Twinlab Corporation and its debtor subsidiaries in the above-captioned cases (collectively, the "Debtors"), dated September 4, 2003 (the "Motion"), for an order pursuant to sections 105, 363, 365, and 1146 of title 11 of the United States Code (11 U.S.C. §§ 101 et seq., the "Bankruptcy Code") (a) approving that certain Asset Purchase Agreement by and among Ideasphere, Inc., TL Acquisition Corp. (collectively, "Ideasphere"), and the Debtors, dated as of September 3, 2003 (the "Purchase Agreement"), a copy of which is annexed hereto as Exhibit A; (b) authorizing the sale of the Purchased Assets¹ free and clear of all liens, claims and encumbrances; (c) authorizing the assumption and assignment of the Purchased Contracts; and (d) establishing certain cure amounts; and objections to the Motion (the "Objections") having been filed; and the Court having heard the statements

A:\TWINLABSALE ORDER.DOC

<sup>&</sup>lt;sup>1</sup> Capitalized terms not otherwise defined herein shall have the meanings ascribed to such terms in the Purchase Agreement.

of counsel and the evidence presented in support of the relief requested by the Debtors in the Motion at a hearing before the Court on October 27, 2003 (the "Hearing"); and it appearing that the Court has jurisdiction over this matter; and it further appearing that the legal and factual bases set forth in the Motion and at the Hearing establish just cause for the relief granted herein; and it further appearing that the relief requested in the Motion is in the best interests of the Debtors, their creditors, and all other parties in interest in the Debtors' chapter 11 cases;

### THE COURT HEREBY FINDS AND DETERMINES THAT:

- A. The findings and conclusions set forth herein constitute the Court's findings of fact and conclusions of law pursuant to Fed. R. Bankr. P. 7052, made applicable to this proceeding pursuant to Fed. R. Bankr. P. 9014.
- B. To the extent any of the following findings of fact constitute conclusions of law, they are adopted as such. To the extent any of the following conclusions of law constitute findings of fact, they are adopted as such.
- C. The court has jurisdiction over the Motion and the transactions contemplated by the Purchase Agreement pursuant to 28 U.S.C. §§ 157 and 1334, and this matter is a core proceeding pursuant to 28 U.S.C. § 157(b)(2)(A) and (N). Venue of these cases and the Motion in this district is proper under 28 U.S.C. §§ 1408 and 1409.
- D. The statutory predicates for the relief sought in the Motion are sections 105(a), 363, 365, and 1146(c) of the Bankruptcy Code as supplemented by Rules 2002, 6004, and 6006 of the Federal Rules of Bankruptcy Procedure (the 'Bankruptcy Rules').

2

- E. As evidenced by the affidavits of service and publication previously filed with the Court, and based on the representations of counsel at the Hearing, (i) proper, timely, adequate, and sufficient notice of the Motion, the Hearing, the Sale (as herein defined) and the assumption and assignment of the Purchased Contracts has been provided in accordance with Bankruptcy Rules 2002(a), 6004(a) and 6006(c) and in compliance with the order of the Court entered September 25, 2003 (the "Bidding Procedures Order") establishing bidding procedures for the auction (the "Auction") of the Purchased Assets, (ii) such notice was good and sufficient, and appropriate under the particular circumstances, and reasonably calculated to reach and apprise all holders of Interests (as hereafter defined) about the Sale thereof, the Auction and the Bidding Procedures, the Sale, and the assumption of the Purchased Contracts and (iii) no other or further notice of the Motion, the Hearing, the Sale, or the assumption and assignment of the Purchased Contracts is or shall be required.
- F. As demonstrated by (i) the testimony and other evidence proffered or adduced at the Hearing, and (ii) the representations of counsel made on the record at the Hearing, the Debtors have marketed the Purchased Assets and conducted the sale process in compliance with the Bidding Procedures Order, the Auction was duly noticed, and a reasonable opportunity has been given to any interested party to make a higher and better offer for the Purchased Assets and the Purchased Contracts, although no such offer was made.
- G. Each Debtor (i) has full corporate power and authority to execute the Purchase Agreement and all other documents contemplated thereby, and the sale of the Purchased Assets by the Debtors has been duly and validly authorized by all

3

A: TWINLAB SALE ORDER. DOC

necessary corporate action of each of the Debtors, (ii) has all of the corporate power and authority necessary to consummate the transactions contemplated by the Purchase Agreement, (iii) has taken all corporate action necessary to authorize and approve the Purchase Agreement and the consummation by the Debtors of the transactions contemplated thereby, and (iv) no consents or approvals, other than those expressly provided for in the Purchase Agreement, are required for the Debtors to consummate such transactions.

- H. Approval of the Purchase Agreement and consummation of the transactions contemplated thereby (the "Sale") at this time is in the best interests of the Debtors, their creditors, their estates, and other parties in interest.
- I. The Debtors have demonstrated compelling circumstances and a good, sufficient, and sound business purpose and justification for the Sale prior to, and outside of, a plan of reorganization.
- J. The Purchase Agreement was negotiated, proposed and entered into by the Debtors and Ideasphere without collusion, in good faith, and from arm's length bargaining positions. Neither the Debtors nor Ideasphere has engaged in any conduct that would cause or permit the Agreement to be avoided under 11 U.S.C. § 363(n).
- K. Ideasphere is a good faith purchaser under section 363(m) of the
   Bankruptcy Code and, as such, is entitled to all of the protections afforded thereby.
- L. Ideasphere is not an "insider" of any of the Debtors, as that term is defined in section 101 of the Bankruptcy Code.

- M. The consideration provided by Ideasphere pursuant to the Purchase Agreement (i) is fair and reasonable, (ii) is the highest and best offer for the Purchased Assets, (iii) will provide a greater recovery to the Debtors' estates than would be provided by any other available alternative, and (iv) constitutes reasonably equivalent value and fair consideration under the Bankruptcy Code and under the laws of the United States, any state, territory, possession, or the District of Columbia.
- N. The Sale must be approved and consummated promptly in order to preserve the viability of the business to which the Purchased Assets relate as a going concern.
- O. The transfer of the Purchased Assets to Ideasphere will be a legal, valid, and effective transfer of the Purchased Assets and, except for the liabilities being assumed by Ideasphere (as defined in section 2.3 of the Purchase Agreement, the "Assumed Liabilities"<sup>2</sup>), will vest Ideasphere with all right, title, and interest of the Debtors to the Purchased Assets free and clear of interests, Claims (as that term is defined in the Bankruptcy Code), or liens, including, but not limited to (i) those that purport to give to any party a right or option to effect any forfeiture, modification, right of first refusal, or termination of the Debtors' or Ideasphere's interest in the Purchased Assets, or any similar rights, (ii) those relating to taxes arising under or out of, in connection with, or in any way relating to the operation of the Purchased Assets prior to the consummation of the Sale (the "Closing"), and (iii) (a) those arising under all mortgages, deeds of trust, security interests, conditional sale or other title retention agreements, pledges, liens, judgments, demands, encumbrances, rights of first refusal or charges of any kind or

<sup>&</sup>lt;sup>2</sup> The Assumed Liabilities do not include the Zions Mortgage.

nature, if any, including, but not limited to, any restriction on the use, voting, transfer, receipt of income or other exercise of any attributes of ownership and (b) all debts arising in any way in connection with any agreements, acts, or failures to act, of any of the Debtors or any of the Debtors' predecessors or affiliates, Claims, obligations, liabilities, demands, guaranties, options, rights, contractual or other commitments, indemnities, indemnity obligations and warranties relating to any acts, omissions or circumstances arising prior to the Closing, including those constituting Excluded Liabilities (as defined in Section 2.4 of the Purchase Agreement), restrictions, interests and matters of any kind and nature, whether known or unknown, contingent or otherwise, whether arising prior to or subsequent to the commencement of these bankruptcy cases, and whether imposed by agreement, understanding, law, equity or otherwise, including but not limited to Claims otherwise arising under doctrines of successor liability or any Claims or Interests related to the Excluded Assets (as defined in Section 2.2 of the Purchase Agreement) or the Excluded Liabilities (collectively, the "Interests").

P. Ideasphere would not have entered into the Purchase Agreement and would not consummate the transactions contemplated thereby, thus adversely affecting the Debtors, their estates, and their creditors, if the sale of the Purchased Assets to Ideasphere, the assignment of the Purchased Contracts to Ideasphere, and the assumption of the Assumed Liabilities by Ideasphere were not, except as otherwise provided for in the Purchase Agreement, free and clear of all Interests of any kind or nature whatsoever, or if Ideasphere would, or in the future could, be liable for any of the Interests including, but not limited to (i) any employment or labor agreements; (ii) any pension, welfare, compensation, or other employee benefit plans, agreements, practices,

6

A::TWINLAB SALE ORDER, DOC

and programs, including, without limitation, any pension plan of the Debtors; (iii) any other employee, worker's compensation, occupational disease, or unemployment or temporary disability related Claim, including without limitation Claims that might otherwise arise under or pursuant to (a) the Employee Retirement, Income, Security Act of 1974, as amended, (b) the Fair Labor Standards Act, (c) Title VII of the Civil Rights Act of 1964, (d) the Coal Industry Retiree Health Benefit Act of 1992, (e) the Federal Rehabilitation Act of 1973, (f) the National Labor Relations Act, (G) the Worker Adjustment and Retraining Act of 1988, (h) the Age Discrimination and Employee Act of 1967, or (i) the Consolidated Omnibus Budget Reconciliation Act of 1985; (iv) any products liability or similar Claims, whether pursuant to any state or federal laws or otherwise, including without limitation, any Claims related to the Excluded Assets; (v) environmental Claims or Liens arising from conditions first existing on or prior to the Closing (including, without limitation, the presence of hazardous, toxic, polluting, or contaminating substances or waste) that may be asserted on any basis, including, without limitation, under the Comprehensive Environmental Response, Compensation, and Liability Act, 42 U.S.C. § 9601 et seq. or similar state statute; (vi) any bulk sales or similar law; and (vii) any tax statutes or ordinances, including, without limitation, the Internal Revenue Code of 1986, as amended.

Q. The Debtors may sell the Purchased Assets free and clear of all Interests of any kind or nature whatsoever because, in each case, one or more of the standards set forth in section 363(f)(1)-(5) of the Bankruptcy Code has been satisfied.

Those (i) holders of Interests and (ii) non-Debtor parties to Purchased Contracts who did not object, or who withdrew their objections, to the Sale or the Motion are deemed to

7

A::TWINLABSALE ORDER, DOC

have consented pursuant to section 363(f)(2) of the Bankruptcy Code. Those (i) holders of Interests and (ii) non-Debtor parties to Purchased Contracts who did object fall within one or more of the other subsections of section 363(f) of the Bankruptcy Code and are adequately protected by having their Interests, if any, attach to the cash proceeds of the Sale ultimately attributable to the property against or in which they claim an Interest.

- R. The Sale is a prerequisite to the Debtors' ability to confirm and consummate a chapter 11 plan or plans, and is made in contemplation of such a plan or plans. Accordingly, the Sale is a transfer pursuant to section 1146(c) of the Bankruptcy Code, which shall not be taxed under any law imposing a stamp, transfer, or any other similar tax.
- S. The Debtors have demonstrated that it is an exercise of their sound business judgment to assume and assign the Purchased Contracts to Ideasphere in connection with the consummation of the Sale, and the assumption and assignment of the Purchased Contracts, subject to the limitations described in Paragraphs O and P of this Order, is in the best interests of the Debtors, their estates, and their creditors. The Purchased Contracts being assigned to, and the liabilities being assumed by, Ideasphere are an integral part of the Purchased Assets being purchased by Ideasphere and, accordingly, such assumption and assignment of the Purchased Contracts and Assumed Liabilities are reasonable and enhance the value of the Debtors' estates.
- T. The Debtors have (i) cured and/or provided adequate assurance of cure of any default existing prior to the date hereof under any of the Purchased Contracts, within the meaning of section 365(b)(1)(A) of the Bankruptcy Code, and (ii) provided compensation or adequate assurance of compensation to any party for any actual

pecuniary loss to such party resulting from a default prior to the date hereof under any of the Purchased Contracts, within the meaning of section 365(b)(1)(B) of the Bankruptcy Code, and Ideasphere has provided adequate assurance of future performance under the Purchased Contracts, within the meaning of section 365(b)(1)(C) of the Bankruptcy Code.

NOW THEREFORE, THE COURT HEREBY ORDERS, ADJUDGES, AND DECREES AS FOLLOWS:

### **General Provisions**

- The Motion is granted, as further described herein.
- 2. The Objections to the Motion or the relief requested therein that have not been withdrawn, waived, or settled, and all reservations of rights included in such Objections, are overruled on the merits, with the exception of certain Objections to cure amounts identified on the record of the Hearing which, if still unresolved, will be the subject of a further hearing to be scheduled.

#### Approval of the Purchase Agreement

- The Purchase Agreement and documents ancillary thereto, and all
  of the terms and conditions thereof, is hereby approved.
- 4. Pursuant to 11 U.S.C. § 363(b), the Debtors are authorized to perform their obligations under and comply with the terms of the Purchase Agreement, and consummate the Sale, pursuant to and in accordance with the terms and conditions of the Purchase Agreement.
- 5. The Debtors are authorized and directed to execute and deliver, and empowered to perform under, consummate and implement, the Purchase Agreement, together with all additional instruments and documents that the Debtors or Ideasphere

9

A:TWINLABSALE ORDER.DOC

deem necessary or appropriate to implement the Purchase Agreement and effectuate the Sale and to take all further actions as may be reasonably requested by Ideasphere for the purpose of assigning, transferring, granting, conveying and conferring to Ideasphere or reducing to possession, the Purchased Assets, or as may be necessary or appropriate to the performance of the obligations as contemplated by the Purchase Agreement.

6. This Order and the Purchase Agreement shall be binding in all respects upon all creditors, claimants of, and holders of equity interests in any Debtor and their successors and assigns (whether known or unknown), any holders of Interests, all non-Debtor parties to the Purchased Contracts, all successors and assigns of Ideasphere, the Debtors and their affiliates and subsidiaries, the Purchased Assets, and any subsequent trustees appointed in the Debtors' chapter 11 cases or upon a conversion to chapter 7 under the Bankruptcy Code and shall not be subject to rejection. Nothing contained in any chapter 11 plan confirmed in these bankruptcy cases or the confirmation order confirming any such chapter 11 plan shall conflict with or derogate from the provisions of the Purchase Agreement or this Order.

### Transfer of Assets

7. Except for the Assumed Liabilities, pursuant to sections 105(a) and 363(f) of the Bankruptcy Code, the Purchased Assets shall be transferred to Ideasphere, and upon the Closing shall be, free and clear of all Interests of any kind or nature whatsoever (including, but not limited to, those described in paragraphs O and P of this Order), and all such Interests of any kind or nature whatsoever shall attach to the net proceeds of the Sale in the order of their priority, with the same validity, force, and effect

which they now have as against the Purchased Assets, subject to any claims and defenses the Debtors may possess with respect thereto.

- 8. Except as expressly permitted or otherwise specifically provided by the Purchase Agreement or this Order, all persons and entities, including, but not limited to, all debt security holders, equity security holders, governmental, tax, and regulatory authorities, lenders, tort claimants, trade and other creditors, holding Interests of any kind or nature whatsoever against or in any of the Debtors, the Excluded Assets, Excluded Liabilities or the Purchased Assets (whether legal or equitable, secured or unsecured, matured or unmatured, contingent or non-contingent, senior or subordinated), arising under or out of, in connection with, or in any way relating to, the Debtors, the Excluded Assets, the Excluded Liabilities, the Purchased Assets, the operation of the Purchased Assets prior to the Closing, or the Sale, are forever barred, estopped, and permanently enjoined from asserting against Ideasphere, its successors or assigns, its property, or the Purchased Assets, such persons' or entities' Interests.
- 9. Nothing in this Order or the Purchase Agreement shall be construed to release or nullify any liability to any governmental entity under police or regulatory requirements that any entity would be subject to as the owner or operator of property after the Closing except to the extent otherwise compromised.
- 10. The transfer of the Purchased Assets to Ideasphere pursuant to the Purchase Agreement constitutes a legal, valid, and effective transfer of the Purchased Assets, and shall vest Ideasphere with all right, title, and interest of the Debtors in and to the Purchased Assets free and clear of all Interests of any kind or nature whatsoever

(including those arising out of or related to the Excluded Assets and the Excluded Liabilities) other than the Assumed Liabilities.

- If any person or entity that has filed financing statements, mortgages, mechanic's liens, lis pendens, or other documents or agreements evidencing Interests in the Debtors or the Purchased Assets shall not have delivered to the Debtors prior to the Closing, in proper form for filing and executed by the appropriate parties, termination statements, instruments of satisfaction, releases of all Interests which the person or entity has with respect to the Debtors or the Purchased Assets or otherwise, then (a) the Debtors are hereby authorized and directed to execute and file such statements, instruments, releases and other documents on behalf of the person or entity with respect to the Debtors or the Purchased Assets and (b) Ideasphere is hereby authorized to file, register, or otherwise record a certified copy of this Order, which shall constitute conclusive evidence of the release of all Interests in the Debtors or the Purchased Assets of any kind or nature whatsoever.
- 12. On the Closing of the Sale, each of the Debtors' creditors and any other holder of an Interest is authorized and directed to execute such documents and take all other actions as may be necessary to release its Interests in the Purchased Assets, if any, as such Interests may have been recorded or may otherwise exist.
- 13. It is a condition to the Closing of the Sale that, on the Closing of the Sale, the Debtors shall, subject only to the provisions and undertakings set forth in the Stipulation and Agreed Order Regarding Preservation of Certain Possible Claims of the Official Committee of Unsecured Creditors approved and "so ordered" by this Court on October \_\_\_, 2003, (i) utilize a portion of the proceeds derived from the Sale to

indefeasibly pay and satisfy in full all outstanding indebtedness and obligations owing under the Amended and Restated Financing Agreement, dated as of September 3, 2003 (as amended, the "DIP Credit Agreement") and (ii) terminate the DIP Credit Agreement and the commitments thereunder. Until the occurrence of the actions described in this paragraph 13 (whether upon the Closing or otherwise), the CIT Group/Business Credit, Inc. (the "DIP Agent"), as agent for the lenders under the DIP Credit Agreement (the "DIP Lenders"), and the DIP Lenders expressly reserve all of their rights and protections under (i) the Final Order (A) Authorizing Postpetition Financing and Granting Security Interests and Superpriority Administrative Expense Status Pursuant to 11 U.S.C. §§ 361, 362, and 364; (B) Modifying the Automatic Stay Pursuant to 11 U.S.C. § 362; and (C) Granting Adequate Protection Pursuant to 11 U.S.C. §§ 361, 362, 363, and 364, (ii) the DIP Credit Agreement, (iii) the Bankruptcy Code and (iv) all other applicable law.

14. The Debtors are authorized to pay all allowed secured claims in full in cash out of the proceeds of the Sale on the Closing Date or thereafter in accordance with the terms and provisions of the operative loan or other documentation.

# Assumption and Assignment to Ideasphere of Purchased Contracts

15. Pursuant to section 365 of the Bankruptcy Code and subject to and conditioned upon the Closing of the Sale, the Debtors' assumption and assignment to Ideasphere of the Purchased Contracts, and Ideasphere's assumption of such contracts on the terms set forth in the Purchase Agreement and including the limitations described in Paragraphs O, P and S of this Order, is hereby approved, and the requirements of section 365(b)(1) of the Bankruptcy Code with respect thereto are hereby deemed satisfied.

13

A: TWINLABSALE ORDER DOC

TRADEMARK REEL: 005482 FRAME: 0891

- 16. The Debtors are hereby authorized and directed in accordance with sections 105(a) and 365 of the Bankruptcy Code to (a) assume and assign to Ideasphere, effective upon the Closing of the Sale, the Purchased Contracts free and clear of all Interests of any kind or nature whatsoever other than the Assumed Liabilities and subject to the limitations described in Paragraphs O, P and S of this Order, and (b) execute and deliver to Ideasphere such documents or other instruments as Ideasphere deems may be necessary to assign and transfer the Purchased Contracts and Assumed Liabilities to Ideasphere.
- 17. With respect to the Purchased Contracts: (a) each Purchased Contract is an executory contract under section 365 of the Bankruptcy Code; (b) the Debtors may assume each of the Purchased Contracts in accordance with section 365 of the Bankruptcy Code; (c) the Debtors may assign each Purchased Contract in accordance with sections 363 and 365 of the Bankruptcy Code, and any provisions in any Purchased Contract that prohibit or condition the assignment of such Purchased Contract or allow the party to such Purchased Contract to terminate, recapture, impose any penalty, condition renewal or extension, or modify any term or condition upon the assignment of such Purchased Contract, constitute unenforceable anti-assignment provisions which are void and of no force and effect; (d) all other requirements and conditions under sections 363 and 365 of the Bankruptcy Code for the assumption by the Debtors and assignment to Ideasphere of each Purchased Contract have been satisfied; (e) the Purchased Contracts shall be transferred and assigned to, and following the closing of the Sale remain in full force and effect for the benefit of, Ideasphere, notwithstanding any provision in any such Purchased Contract (including those of the type described in sections 365(b)(2) and (f) of

14

A:TWINLABSALE ORDER.DOC

the Bankruptcy Code) that prohibits, restricts, or conditions such assignment or transfer and, pursuant to section 365(k) of the Bankruptcy Code, the Debtors shall be relieved from any further liability with respect to the Purchased Contracts after such assignment to and assumption by Ideasphere; and (f) upon Closing, in accordance with sections 363 and 365 of the Bankruptcy Code, Ideasphere shall be fully and irrevocably vested in all right, title and interest of each Purchased Contract.

Purchased Contracts arising or accruing prior to the Closing (without giving effect to any acceleration clauses or any default provisions of the kind specified in section 365(b)(2) of the Bankruptcy Code) shall be cured by the Debtors at the Closing or as soon thereafter as practicable by payment of the Cure Amounts (as hereinafter defined), and, Ideasphere shall have no liability or obligation related to any act or occurrence arising or accruing prior to the date of the Closing, except as otherwise expressly provided in the Purchase Agreement. The schedule attached to the Notice of Auction sent to the non-Debtor parties to the Purchased Contracts (the "Executory Contracts Schedule"), attached in substantial form hereto as Exhibit B, 3 reflects the sole amounts necessary under section 365(b) of the Bankruptcy Code to cure all monetary defaults under the Purchased

<sup>&</sup>lt;sup>3</sup> Following service of the Executory Contracts Schedule, the Debtors received, among others, objections from Damage Recovery Systems, Inc. ("DRS"), Inter-Cal Neutraceuticals, n/k/a Zila Neutraceuticals, Inc. ("Inter-Cal"), and Original Creatine Patent Company, Ltd. ("OCPC") regarding the Debtors' proposed cure amounts of their respective agreements listed on the schedule. Following negotiations between the parties, the Debtors have reached agreement with DRS, Inter-Cal, and OCPC regarding the Cure Amounts and the Executory Contracts Schedule has been revised to reflect the agreed upon cure amount. Pursuant to the consensual resolution with OCPC, OCPC reserves the right to audit the royalties under the assumed licensing agreement and the Debtors agree to adjust the cure amount accordingly, if necessary, based on the results of the audit. The Debtors have determined not to assume certain agreements that were originally attached to the Notice of Auction and such agreements are not included in Exhibit B.

Contracts (collectively, the "Cure Amounts"), and no other amounts are or shall be due to the non-Debtor parties in connection with the assumption by the Debtors and the assignment to Ideasphere of the Purchased Contracts. Notwithstanding the foregoing, to the extent Objections to the Cure Amounts are not resolved consensually, the Debtors will schedule a hearing to consider such objections subsequent to the Hearing and, pending resolution of such Objections, shall establish a post-Closing escrow in an amount equal to the differential between the cure amounts alleged in the Objections to the Cure Amounts and the respective Cure Amounts set forth in the Executory Contracts Schedule. In the event one or more agreements is added to or deleted from Exhibit B on or before the Closing, the Debtors will serve a notice of the amended Exhibit B on the non-Debtor parties to such Contracts. To the extent such non-Debtor parties are added to the amended Exhibit B, such non-Debtor parties shall have thirteen (13) days thereafter to object to the Cure Amount, unless such period is shortened by the Court.

barred, estopped, and permanently enjoined from (i) asserting against the Debtors or Ideasphere, or the property of any of them, any default or Claim arising out of any indemnity obligation or warranties for acts or occurrences arising prior to or existing as of the Closing, including those constituting Excluded Liabilities, or, against Ideasphere, any counterclaim, defense, setoff or any other Claim asserted or assertable against the Debtors; and (ii) imposing or charging against Ideasphere or its affiliates any rent accelerations, assignment fees, increases or any other fees as a result of the Debtors' assumption and assignments to Ideasphere of the Purchased Contracts. The validity of such assumption and assignments of the Purchased Contracts shall not be affected by any

16

A:\TWINLABSALE ORDER.DOC

dispute between the Debtors and any non-Debtor party to a Purchased Contract relating to such contract's respective Cure Amount.

- 20. Except as provided in the Purchase Agreement or this Order, after the Closing, the Debtors and their estates shall have no further liabilities or obligations with respect to any Assumed Liabilities and all holders of such Claims are forever barred and estopped from asserting such Claims against the Debtors, their successors or assigns, their property or their assets or estates.
- 21. The failure of the Debtors or Ideasphere to enforce at any time one or more terms or conditions of any Purchased Contract shall not be a waiver of such terms or conditions, or of the Debtors' and Ideasphere's rights to enforce every term and condition of the Purchased Contracts.

### Additional Provisions

- 22. The consideration provided by Ideasphere for the Purchased Assets under the Purchase Agreement is deemed to constitute reasonably equivalent value and fair consideration under the Bankruptcy Code and under the laws of the United States, any state, territory, possession, or the District of Columbia.
- 23. The consideration provided by Ideasphere for the Purchased Assets under the Purchase Agreement is fair and reasonable and the Sale may not be avoided under section 363(n) of the Bankruptcy Code.
- 24. This Order (a) shall be effective as a determination that, except for the Assumed Liabilities, at Closing, all Interests of any kind or nature whatsoever existing as to the Debtors and the Purchased Assets prior to the Closing have been unconditionally released, discharged and terminated, and that the conveyances described herein have been effected, and (b) shall be binding upon and shall govern the acts of all

17

A: TWINLABSALE ORDER.DOC

entities including without limitation, all filing agents, filing officers, title agents, title companies, recorders of mortgages, recorders of deeds, registrars of deeds, administrative agencies, governmental departments, secretaries of state, federal, state, and local officials, and all other persons and entities who may be required by operation of law, the duties of their office, or contract, to accept, file, register or otherwise record or release any documents or instruments, or who may be required to report or insure any title or state of title in or to any of the Purchased Assets.

- 25. Each and every federal, state, and local governmental agency or department is hereby directed to accept any and all documents, instruments, and permits necessary and appropriate to consummate the transactions contemplated by the Agreement.
- Agreement, Ideasphere shall have no obligation to pay wages, bonuses, severance pay, benefits (including, without limitation, contributions or payments on account of any under-funding with respect to any and all pension plans) or any other payment with respect to employees or former employees of the Debtors. Except as otherwise expressly provided in the Purchase Agreement, Ideasphere shall have no liability with respect to any collective bargaining agreement, employee pension plan, employee welfare or retention, benefit and/or incentive plan to which any Debtor is a party and relating to the Purchased Assets (including, without limitation, arising from or related to the rejection or other termination of any such agreement), and Ideasphere shall in no way be deemed a party to or assignee of any such agreement, and no employee of Ideasphere shall be deemed in any way covered by or a party to any such agreement, and except for Assumed

18

A: TWINLAB SALE ORDER. DOC

Liabilities, all parties to any such agreement are hereby enjoined from asserting against Ideasphere any and all Claims or Interests arising from or relating to such agreement. All notices, if any, required to be given to the Debtors' employees pursuant to the Workers Adjustment and Relocation Adjustment Act, or any similar federal or state law, shall be the sole responsibility and obligation of the Debtors, and, notwithstanding anything set forth in this Order, Ideasphere shall have no duties, responsibility, or liability therefor.

- 27. All entities who are in possession of some or all of the Purchased Assets on the Closing are hereby directed to surrender possession of the Purchased Assets to Ideasphere at Closing.
- otherwise specifically provided for in the Purchase Agreement or this Order, Ideasphere shall have no liability or responsibility for any liability or other obligation of the Debtors arising under or related to the Purchased Assets other than for the Assumed Liabilities, and shall have no obligations arising out of or related to the Excluded Assets or Excluded Liabilities. Without limiting the generality of the foregoing, and except as otherwise specifically provided herein and in the Purchase Agreement, Ideasphere shall not be liable for any Claims or Interests in or against the Debtors or any of their predecessors or affiliates, and Ideasphere shall have no successor or vicarious liabilities of any kind or character including, but not limited to, any theory of antitrust, environmental, successor or transferee liability, labor law, de facto merger, or substantial continuity, whether known or unknown as of the Closing, now existing or hereafter arising, whether fixed or contingent, with respect to the Debtors or any obligations of the Debtors arising prior to the Closing, including, but not limited to, products liability and liabilities on account of

any Taxes arising, accruing, or payable under, out of, in connection with, or in any way relating to the operation of the Purchased Assets prior to the Closing or the Excluded Assets and Excluded Liabilities prior to or after the Closing. Ideasphere has given substantial consideration under the Purchase Agreement for the benefit of holders of Interests. The consideration given by Ideasphere shall constitute valid and valuable consideration for the releases of any potential Claims of successor liability of Ideasphere, releases which the Court holds shall be deemed to have been given in favor of Ideasphere by all holders of Interests against the Debtors or their respective assets.

29. Under no circumstances shall Ideasphere be deemed a successor of or to the Debtors for any Interest against or in the Debtors, the Purchased Assets, the Excluded Assets or the Excluded Liabilities of any kind or nature whatsoever. Except for the Assumed Liabilities, the sale, transfer, assignment and delivery of the Purchased Assets shall not be subject to any Interests, and Interests of any kind or nature whatsoever shall remain with, and continue to be obligations of, the Debtors. Except for the Assumed Liabilities, all persons holding Interests against or in the Debtors, the Purchased Assets, the Excluded Assets or the Excluded Liabilities of any kind or nature whatsoever (including but not limited to, the Sellers and/or their respective successors, including any trustees thereof, creditors, employees, unions, former employees and shareholders, administrative agencies, governmental units, secretaries of state, federal, state and local officials, maintaining any authority relating to any environmental, health and safety laws, and their respective successors or assigns) shall be, and hereby are, forever barred, estopped, and permanently enjoined from asserting, prosecuting, or otherwise pursuing such Interests of any kind or nature whatsoever against Ideasphere, its property, its

20

ASTWINLABSALE ORDER DOC

successors and assigns, or the Purchased Assets, as an alleged successor or otherwise, with respect to any Interest of any kind or nature whatsoever such person or entity had, has, or may have against or in the Debtors, the Debtors' estates, their respective officers, directors, shareholders, the Purchased Assets, the Excluded Assets or the Excluded Liabilities. Following the Closing, no holder of an Interest in the Debtors shall interfere with Ideasphere's title to or use and enjoyment of the Purchased Assets based on or related to such Interest, or any actions that the Debtors may take in their Chapter 11 cases.

- shall be deemed to sell, transfer, assign, or convey the Excluded Assets (as defined in section 2.2 of the Purchase Agreement) to Ideasphere and the Debtors shall retain all right, title and interest to, in and under the Excluded Assets or (ii) shall be deemed to be an assumption by Ideasphere of the Excluded Liabilities (as defined in section 2.4 of the Purchase Agreement). Under no circumstances shall Ideasphere be deemed a successor of or to the Debtors for any liabilities of the Debtors relating to the Excluded Assets or Excluded Liabilities.
- 31. This Court retains jurisdiction to enforce and implement the terms and provisions of this Order, the Purchase Agreement, all amendments thereto, any waivers and consents thereunder, and of each of the agreements executed in connection therewith in all respects, including, but not limited to, retaining jurisdiction to (a) compel delivery of the Purchased Assets to Ideasphere, (b) compel delivery of the purchase price or performance of other obligations owed by or to the Debtors, (c) resolve any disputes arising under or related to the Purchase Agreement, except as otherwise provided therein,

21

A::TWINLABSALE ORDER.DOC

- (d) interpret, implement, and enforce the provisions of this Order, and (e) protect

  Ideasphere against (i) any of the Excluded Liabilities, (ii) the assertion of any Interests

  against the Purchased Assets, of any kind or nature whatsoever, or (iii) the assertion of

  any Claims or Interests arising out of the Excluded Assets or the Excluded Liabilities.
- 32. The transactions contemplated by the Purchase Agreement are undertaken by Ideasphere without collusion and in good faith, as that term is used in section 363(m) of the Bankruptcy Code, and accordingly, the reversal or modification on appeal of the authorization provided herein to consummate the Sale shall not affect the validity of the Sale (including the assumption and assignment of any of the Purchased Contracts), unless such authorization is duly stayed pending such appeal. Ideasphere is a purchaser in good faith of the Purchased Assets, and is entitled to all of the protections afforded by section 363(m) of the Bankruptcy Code.
- 33. The terms and provisions of the Purchase Agreement and this Order shall be binding in all respects upon, and shall inure to the benefit of, the Debtors, their estates, and their creditors, Ideasphere and its respective affiliates, successors, and assigns, and any affected third parties including, but not limited to, all persons asserting an Interest in the Purchased Assets, notwithstanding any subsequent appointment of any trustee(s) under any chapter of the Bankruptcy Code, as to which trustee(s) such terms and provisions likewise shall be binding.
- 34. Nothing contained in any plan of reorganization confirmed in these cases or any order of this Court confirming such plan shall conflict with or derogate from the provisions of the Purchase Agreement or the terms of this Order.

- 35. The failure specifically to include any particular provisions of the Purchase Agreement in this Order shall not diminish or impair the effectiveness of such provision, it being the intent of the Court that the Purchase Agreement be authorized and approved in its entirety.
- 36. The Purchase Agreement and any related agreements, documents or other instruments may be modified, amended or supplemented by the parties thereto and in accordance with the terms thereof, without further order of the Court, provided that any such modification, amendment, or supplement does not have a material adverse effect on the Debtors' estates or an adverse effect on the rights or interests of the DIP Agent and/or DIP Lenders. Unless consented to in writing by the DIP Agent, any modification, amendment or supplement that adversely affects the rights or interests of the DIP Agent and/or DIP Lenders shall, in the absence of a further order of this Court on notice to such party, be null and void and without effect..
- 37. The transfer of the Purchased Assets pursuant to the Sale is a transfer pursuant to section 1146(c) of the Bankruptcy Code, and accordingly shall not be taxed under any law imposing a stamp tax or a sale, transfer, or any other similar tax.

  Each and every federal, state and local government agency or department is hereby directed to accept any and all documents and instruments necessary and appropriate to consummate the transfer of any of the Purchased Assets, all without imposition or payment of any stamp tax, transfer tax, or similar tax.
- 38. The provisions of this Order are non-severable and mutually dependent and, pursuant to Bankruptcy Rules 6004(g) and 6006(d), this Order shall not be stayed for 10 days and shall be effective immediately upon entry.

23

A::TWINLAB SALE ORDER.DOC

Dated: October 30, 2003 New York, New York

> /s/ Cornelius Blackshear UNITED STATES BANKRUPTCY JUDGE

# EXHIBIT A

PURCHASE AGREEMENT

A:STWINLABSALE ORDER.DOC

TRADEMARK REEL: 005482 FRAME: 0903

# EXHIBIT B

EXECUTORY CONTRACTS SCHEDULE

A::TWINLABSALE ORDER.DOC

TRADEMARK REEL: 005482 FRAME: 0904 December 18, 2003

Idea Sphere, Inc.
TL Acquisition Corp.
3133 Orchards Vista Drive, SE
Grand Rapids, MI 49546

Highbridge/Zwim Special Opportunities Fund, L.P., as Agent 9 West 57th Street 27th Floor
New York, NY 10019

The undersigned (individually and collectively, referred to herein as the "Blechman Parties") acknowledge, confirm and agree that upon the transfer and sale of the Purchased Assets (as defined in that certain Asset Purchase Agreement by and among TL, Twinlab, Twin Laboratories (UK) Ltd, Idea Sphere, Inc. ("ISI") and TL Acquisition Corp. ("TLAC"), dated as of September 4, 2003, the "Asset Purchase Agreement") to ISI and TLAC:

- (a) the Purchased Assets will be free and clear of any interest, claim, security interest, lien or any other encumbrance of the Blechman Parties;
- (b) any interest, claim, security interest, lien or any other encumbrance of the Blechman Parties in or to the Purchased Assets shall be released; and
- (c) any interest, claim, security interest, lien or any other encumbrance of the Blechman Parties in or to the Purchased Assets shall attach to the net proceeds of the sale of the Purchased Assets

pursuant to that certain "Order Pursuant to Sections 105, 363(b), (f) and (m), 365, and 1146 of the Bankruptcy Code and Bankruptcy Rule 2002 Authorizing (I) the Sale of Substantially All of the Debtors' Assets Free and Clear of Liens, Claims, Encumbrances to Idea Sphere Inc. and TL Acquisition Corp., (II) the Assumption and Assignment of certain Executory Contracts, and (III) the Assumption of Certain Liabilities" entered by the Bankruptcy Court on October 31, 2003.

[Signature page follows]

KI,1:2242570.3

FROM : BRIAN BLECHMAN FRX NO. : 6315470484

Dec. 19 2003 10:100M P3

Brian Blechunan

Neil Blechman

Dean Blechman

Ross Blechman

Steve Blechman

Linda Blechman

Sharon Blechman

Robin Blechman

Elyse Blechman

Helena Blechman

KI.2:2242570.2

Brian Blechman

Dean Blechman

Neil Blechman

Ross Blechman

Steve Blechman

Linda Blechman

Sharon Blechman

Robin Blechman

Elyse Blechman

Helena Blechman

KL2:22-2576.3

Brian Blechman

Dean Blechman

Neil Blechman

Ross Blechman

Steve Blechman

Linda Blechman

Sharon Blechman

Robin Blechman

Elyse Blechman

**TRADEMARK** 

REEL: 005482 FRAME: 0908

Bri:	an B	lechn	าลก

Dean Blechman

Nail Blochman

Ross Blechman

Steve Blechman

Linda Blechman

Sharon Blechman

Robin Blechman

Hlyse Blechman

Helena Blechman

10.2:2342570.3

TRADEMARK REEL: 005482 FRAME: 0909

## ASSET PURCHASE AGREEMENT

ASSET PURCHASE AGREEMENT, dated as of September 4, 2003 (this "Agreement"), among Twinlab Corporation, a Delaware corporation, Twin Laboratories Inc., a Utah corporation, and Twin Laboratories (UK) Ltd., a company organized under the laws of the United Kingdom (each, a "Seller," and collectively, "Sellers"), and IdeaSphere, Inc., a Michigan corporation ("IdeaSphere"), and TL Acquisition Corp., a Delaware corporation ("Acquisition Sub," each of IdeaSphere and Acquisition Sub being individually referred to as a "Purchaser", and together as "Purchasers").

### WITNESSETH:

WHEREAS, Sellers have agreed that, subject to the terms of this Agreement, Sellers shall file voluntary petitions for reorganization relief to be jointly administered for procedural purposes (the "Bankruptcy Case") pursuant to Chapter 11 of Title 11 of the United States Code, 11 U.S.C. § 101 et seq. (the "Bankruptcy Code") on the Petition Date, and in concert with such filing, shall simultaneously seek the entry of an order by the United States Bankruptcy Court for the Southern District of New York (the "Bankruptcy Court") approving this Agreement and authorizing Sellers to consummate the transactions contemplated hereby and by the Ancillary Agreements;

WHEREAS, Sellers presently conduct the Business;

WHEREAS, Sellers desire to sell, transfer and assign to Purchasers, and Purchasers desires to acquire and assume from Sellers, pursuant to Sections 363 and 365 of the Bankruptcy Code, the Purchased Assets and Assumed Liabilities, free and clear of all Liens, to the maximum extent permissible under Sections 363 and 365 of the Bankruptcy Code, as more specifically provided herein; and

WHEREAS, certain terms used in this Agreement are defined in Section 1.1;

NOW, THEREFORE, in consideration of the premises and the mutual covenants and agreements hereinafter contained, the parties hereby agree as follows:

### ARTICLE I

#### **DEFINITIONS**

### 1.1 Certain Definitions.

For purposes of this Agreement, the following terms shall have the meanings specified in this <u>Section 1.1</u>:

NY2: 1309423\04\S2CV04!.DOC\78300.0003

"Affiliate" means, with respect to any Person, any other Person that, directly or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, such Person, and the term "control" (including the terms "controlled by" and "under common control with") means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through ownership of voting securities, by contract or otherwise.

"Ancillary Agreements" means the Escrow Agreement, the Assignment And Assumption Agreement, the Bill of Sale, the Real Property Conveyance Documents and the Intellectual Property Assignments.

"Balance Sheet Date" means June 30, 2003.

"Bidding Procedures" means the procedures set forth in the Bidding Procedures Order pursuant to which Alternative Transactions may be solicited, made and accepted and containing the applicable terms and conditions set forth in Section 7.2 hereof.

"Bidding Procedures Order" means the order of the Bankruptcy Court, the form and substance of which shall be subject to the approval of Sellers and Purchasers, and substantially in the form of Exhibit A, which shall, among other things (a) set a deadline for the filing of objections to the entry of the Sale Order; (b) schedule a hearing on the motion to approve the Sale Order and notice procedures in connection therewith; (c) approve the Bidding Procedures and payment of the Break-Up Fee and the Expense Reimbursement; (d) establish notice procedures with respect to the assumption and assignment of the Purchased Contracts; and (e) establish the date of the Auction.

"Business" means the business of Sellers, including the manufacturing and marketing of nutritional products, including vitamins, minerals, supplements, herbals and tea, and diet, energy and sports nutrition products including powders, bars, drinks and other health products sold through health and natural food stores, national and regional drug store chains, supermarkets, mass merchandise retailers, military post exchanges and other channels; provided, that for all purposes of this Agreement, the term Business shall exclude any and all business of Sellers relating to the Excluded Products.

"Business Day" means any day of the year on which national banking institutions in New York are open to the public for conducting business and are not required or authorized to close.

"CERCLA" means the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended.

"CERCLIS" means the Comprehensive Environmental Response, Compensation and Liability Information System maintained by the United States Environmental Protection Agency.

"Code" means the Internal Revenue Code of 1986, as amended.

"Contract" means any oral or written contract, indenture, note, bond, lease instrument, commitment, obligation, undertaking or other agreement.

"Documents" means all files, documents, instruments, papers, books, reports, records, tapes, microfilms, photographs, letters, budgets, forecasts, ledgers, journals, title policies, customer lists, regulatory filings, operating data and plans, technical documentation (design specifications, functional requirements, operating instructions, logic manuals, flow charts, etc.), user documentation (installation guides, user manuals, training materials, release notes, working papers, etc.), marketing documentation (sales brochures, flyers, pamphlets, web pages, etc.), and other similar materials related primarily to the Business and the Purchased Assets in each case whether or not in electronic form.

"Employees" means all individuals who are employed by Sellers in connection with the Business as of the Closing.

"Environmental Action" means any action, suit, demand letter, claim by any Person or entity, including any Governmental Body, notice of non-compliance or violation, notice of liability or potential liability, investigation, proceeding, consent order or consent agreement relating to any Environmental Law, Environmental Permit or Hazardous Materials or arising from alleged injury or threat to health and safety or the environment relating to any Environmental Law, including (a) by any such Person, entity or Governmental Body for enforcement, cleanup, removal, response, remedial or other actions or damages and (b) by any Governmental Body or third party for damages, contribution, indemnification, cost recovery, compensation or injunctive relief.

"Environmental Law" means any foreign, federal, state or local statute, regulation, ordinance, or rule of common law currently in effect relating to the protection of human health and safety or the environment including the Comprehensive Environmental Response, Compensation and Liability Act (42 U.S.C. § 9601 et seq.), the Hazardous Materials Transportation Act (49 U.S.C. App. § 1801 et seq.), the Resource Conservation and Recovery Act (42 U.S.C. § 6901 et seq.), the Clean Water Act (33 U.S.C. § 1251 et seq.), the Clean Air Act (42 U.S.C. § 7401 et seq.) the Toxic Substances Control Act (15 U.S.C. § 2601 et seq.), the Federal Insecticide, Fungicide, and Rodenticide Act (7 U.S.C. § 136 et seq.), and the Occupational Safety and Health Act (29 U.S.C. § 651 et seq.), and the regulations promulgated pursuant thereto.

"Environmental Permit" means any Permit required under any Environmental Law.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"Escrow Agreement" means the Escrow Agreement in customary form as agreed between Purchasers and Sellers.

"Excluded Contracts" means all Contracts to which any Seller is a party that is not a Purchased Contract.

"Excluded Products" means any and all products that include (i) alkaloids from the herb known as "Ma Huang," also known as ephedra and (ii) Kava Kava, including the products set forth on Schedule 1.1(b).

"FTC" means Federal Trade Commission.

"Furniture and Equipment" means all furniture, fixtures, furnishings, machinery, tools, manufacturing and other equipment, vehicles, leasehold improvements, and other tangible personal property owned or used by any Seller in the conduct of the Business, including all artwork, desks, chairs, tables, computer and computer-related hardware (including, computers, file servers, facsimile servers, scanners, color printers, laser printers and networks), copiers, telephone lines and numbers, telecopy machines and other telecommunication equipment, cubicles and miscellaneous office furnishings and supplies.

"GAAP" means generally accepted accounting principles in the United States as of the date hereof.

"Governmental Body" means any government or governmental or regulatory body thereof, or political subdivision thereof, whether foreign, federal, state, or local, or any agency, instrumentality or authority thereof, or any court or arbitrator (public or private).

"Hazardous Material" means any substance, material or waste which is regulated by any Governmental Body including petroleum and its by-products, asbestos and any material or substance which is defined as a "hazardous waste," "hazardous substance," "hazardous material," "restricted hazardous waste," "industrial waste," "solid waste," "contaminant," "pollutant," "toxic waste" or "toxic substance" under any provision of Environmental Law.

"HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.

"Indebtedness" of any Person means, without duplication, (i) the principal of and premium (if any) in respect of (A) indebtedness of such Person for money borrowed and (B) indebtedness evidenced by notes, debentures, bonds or other similar instruments for the payment of which such Person is responsible or liable; (ii) all obligations of such Person issued or assumed as the deferred purchase price of property, all conditional sale obligations of such Person and all obligations of such Person under any title retention agreement (but excluding trade accounts payable and other accrued current liabilities arising in the Ordinary Course of Business); (iii) all obligations of such Person under leases required to be capitalized in accordance with GAAP; (iv) all obligations of such Person for the reimbursement of any obligor on any letter of credit, banker's acceptance or similar credit transaction; (v) all obligations of the type referred to

in clauses (i) through (iv) of other Persons for the payment of which such Person is responsible or liable, directly or indirectly, as obligor, guarantor or otherwise, including guarantees of such obligations; and (vi) all obligations of the type referred to in clauses (i) through (v) of other Persons secured by any Lien on any property or asset of such Person (whether or not such obligation is assumed by such Person).

"Intellectual Property Licenses" means (i) any grant by Sellers to a third Person of any right to make, use, have made, sell, offer to sell, or grant sublicenses to do the foregoing for any of the Purchased Intellectual Property in connection with the Business, and (ii) any grant to Sellers of a right to make, use, have made, sell, offer to sell, or grant sublicenses to do the foregoing for a third Person's Intellectual Property Rights in connection with the Business.

"Intellectual Property Rights" means (i) all patents and applications therefor, including continuations, divisionals, continuations-in-part, or reissue patent applications and patents issuing thereon including any foreign counterparts thereof (collectively, "Patents"), (ii) all trademarks, service marks, trade names, service names, brand names, all trade dress rights, logos, Internet domain names and corporate names and general intangibles of a like nature, together with the goodwill associated with any of the foregoing, and all applications, registrations and renewals thereof (collectively, "Marks"), (iii) copyrights and registrations and applications therefor and works of authorship, and mask work rights (collectively, "Copyrights") and (iv) discoveries, concepts, ideas, research and development, know-how, formulae, inventions (whether patentable or unpatentable and whether or not reduced to practice), invention disclosures, apparatus, creations, improvements, works of authorship and other similar materials, and all recordings, graphs, drawings, reports, analyses, compositions, manufacturing and production processes and techniques, technical data, procedures, designs, drawings, specifications, databases, and other proprietary and confidential information, including customer lists, supplier lists, pricing and cost information, and business and marketing plans and proposals of Sellers, and other writings, and other tangible embodiments of the foregoing, in any form whether or not specifically listed herein and, in each case, all related technology (collectively, "Trade Secrets"), whether presently existing or created or acquired anywhere in the world between the date of this Agreement and the Closing Date.

"IRS" means the Internal Revenue Service.

"Knowledge of Sellers" means the actual knowledge of those officers and directors of Sellers listed on Schedule 1.1(c).

"<u>Law</u>" means any federal, state, local or foreign law, statute, code, ordinance, rule or regulation.

"Legal Proceeding" means any judicial, administrative or arbitral actions, suits, proceedings (public or private) or claims or any proceedings by or before a Governmental Body.

"Liabilities" means debts, losses, liabilities, claims (including claims as defined in the Bankruptcy Code), damages, expenses, fines, costs, royalties, proceedings, deficiencies, indebtedness, commitments or obligations (including those arising out of any Legal Proceeding, such as any settlement or compromise thereof or judgment or award therein), of any nature, whether known or unknown, absolute or contingent, accrued or unaccrued, disputed or undisputed, liquidated or unliquidated, secured or unsecured, joint or several, direct or indirect, due or to become due, vested or unvested, executory, determined, determinable or otherwise, and whether or not resulting from third party claims, and any reasonable out-of-pocket costs and expenses (including reasonable legal counsels', accountants', or other fees and expenses incurred in defending any Legal Proceeding or in investigating any of the same, either prior to or after the commencement thereof, or in asserting any rights hereunder), but not including consequential, exemplary, special and punitive damages and lost profits.

"Lien" means any lien, encumbrance, hypothecation, pledge, mortgage, deed of trust, security interest, claim (including but not limited to any claim arising out of or related to product liability claims), lease, charge, option, right of first refusal, easement, encroachment, servitude, proxy, voting trust or agreement, transfer restriction under any shareholder or similar agreement or encumbrance.

"Material Adverse Effect" means (i) a material adverse effect on the business, assets, results of operations or financial condition of Sellers and their subsidiaries taken as a whole, or (ii) a material adverse effect on the ability of Sellers taken as a whole to consummate the transactions contemplated by this Agreement or perform their obligations under this Agreement, other than an effect resulting from an Excluded Matter. "Excluded Matter" means any one or more of the following: (i) the effect of any change in the United States or foreign economies or securities or financial markets in general; (ii) the effect of any change that generally affects any industry in which Sellers operate; (iii) the effect of any change arising in connection with earthquakes, hostilities, acts of war, sabotage or terrorism or military actions or any escalation or material worsening of any such hostilities, acts of war, sabotage or terrorism or military actions existing or underway as of the date hereof; (iv) any effect resulting from the public announcement of this Agreement, compliance with terms of this Agreement or the consummation of the transactions contemplated by this Agreement; (v) any effect resulting from a change in GAAP following the date of this Agreement; or (vi) any effect resulting from the filing of the Bankruptcy Case and reasonably anticipated effects thereof.

"NPL" means the National Priorities List under CERCLA.

"Order" means any order, injunction, judgment, decree, ruling, writ, assessment or arbitration award of a Governmental Body.

"Ordinary Course of Business" means the ordinary and usual course of normal day-to-day operations of the Business through the date hereof consistent with past practice.

"Permits" means any approvals, authorizations, consents, licenses, permits or certificates of a Governmental Body.

"Permitted Exceptions" means (i) all defects, exceptions, restrictions, easements, rights of way and encumbrances disclosed in policies of title insurance which have been made available to Purchasers and disclosed on Schedule 1.1(d); (ii) statutory Liens for current Taxes, assessments or other governmental charges not yet delinquent or the amount or validity of which is being contested in good faith by appropriate proceedings provided an appropriate reserve is established therefor; (iii) mechanics', carriers', workers', repairers' and similar Liens arising or incurred in the Ordinary Course of Business for obligations that are not yet delinquent (in the case of clauses (i) through (iii), which are not individually or in the aggregate material); (iv) zoning, entitlement and other land use and environmental regulations by any Governmental Body provided that such regulations have not been violated; (v) title of a lessor under a capital or operating lease that has been entered into in the Ordinary Course of Business and not in anticipation of the transactions contemplated by this Agreement; and (vi) other immaterial imperfections in title, charges, easements, restrictions and encumbrances.

"Person" means any individual, corporation, partnership, firm, joint venture, association, joint-stock company, trust, unincorporated organization, Governmental Body or other entity.

"Petition Date" means September 4, 2003.

"Products" means any and all products developed, manufactured, marketed or sold by Sellers, whether work in progress or in final form, except for the Excluded Products.

"Purchased Contracts" means the Contracts set forth on Schedule 1.1(e) provided that in no event shall Purchased Contracts include Contracts related to the Excluded Products.

"Purchased Intellectual Property" means all Intellectual Property Rights owned, licensed or used by Sellers in connection with the Business or the Purchased Assets, and including all Software owned, licensed or used by Seller.

"Real Property Conveyance Documents" means the documents in customary form to be agreed upon between Purchasers and Sellers prior to Closing, pursuant to which Sellers will, at Closing, sell and assign all of their right, title and interest to the Real Property to Purchasers in accordance with this Agreement.

"Release" means any release, spill, emission, leaking, pumping, injection, deposit, disposal, discharge, dispersal, or leaching into the indoor or outdoor environment, or into or out of any property.

"Sale Order" means an order of the Bankruptcy Court (satisfactory in form and substance to Purchasers and Sellers), which order shall not have been stayed, vacated

or otherwise rendered ineffective, authorizing, among other things, (i) the sale of the Purchased Assets to Purchasers free and clear of all Liens, including any claims that might arise out of any product liability suit that might otherwise be brought against the Purchasers under the doctrine of successor liability or similar doctrine or theory; (ii) the assignment of the Purchased Contracts to and the assumption of the Purchased Contracts by Purchasers; and (iii) the consummation of the transactions contemplated by this Agreement and all other transactions and agreements contemplated hereby and containing a finding that each Purchaser is a good faith purchaser for value and is purchasing the Purchased Assets in good faith within the meaning of Section 363(m) of the Bankruptcy Code.

"Software" means any and all (i) computer programs, including any and all software implementations of algorithms, models and methodologies, whether in source code or object code, (ii) databases and compilations, including any and all data and collections of data, whether machine readable or otherwise, (iii) descriptions, flow-charts and other work product used to design, plan, organize and develop any of the foregoing, screens, user interfaces, report formats, firmware, development tools, templates, menus, buttons and icons, and (iv) all documentation including user manuals and other training documentation related to any of the foregoing, whether presently existing or acquired or created anywhere in the world between the date of this Agreement and the Closing Date.

"Specified Assets" means any assets of Sellers (including Sellers' product held as inventory) that relate solely to the Excluded Products.

"Specified Liabilities" means any Liabilities of Sellers relating to or arising out of the manufacture, sale or distribution by Sellers of Excluded Products (including sales by any Seller to Division Sales or any of their Affiliates), including any Liabilities arising out of or related to the Legal Proceedings described on Schedule 1.1(f).

"<u>Tax Authority</u>" means any state or local government, or agency, instrumentality or employee thereof, charged with administration of any law or regulation relating to Taxes.

"<u>Tax Return</u>" means any report, return, information filing, claim for refund or other information, including any schedules or attachments thereto, and any amendments to any of the foregoing required to be supplied to a Tax Authority in connection with Taxes.

"Taxes" means (i) all federal, state, local and foreign taxes, charges, fees, imposts, levies or other assessments, including all net income, gross receipts, capital, excise, employment, sales, use, ad valorem, value added, transfer, license, payroll, franchise, profits, inventory, capital stock, severance, stamp, withholding, Social Security, unemployment, occupation, disability, real property, personal property, registration, alternative or add-on minimum, estimated or other taxes, charges, fees, imposts, levies or other assessments, including any interest, penalties or additions thereto,

whether disputed or not, and (ii) any liability for any items described in clause (i) payable by reason of contract, transferee liability, operation of law (including Treasury Regulation 1.1502-6) or otherwise.

"<u>Utah Facility</u>" means that certain property located at 600 East Quality Drive, American Fork, Utah 84003, including all buildings, facilities and improvements thereon.

"WARN" means the Worker Adjustment and Retraining Notification Act of 1988, as amended.

"Zions Mortgage" means that certain Mortgage Agreement, dated as of May 7, 1996, between Sellers and Zions First National Bank relating to the Utah Facility.

"Zions Mortgage Balance" means the total amount payable to Zions First National Bank under the Zions Mortgage, as of the Closing Date.

1.2 <u>Terms Defined Elsewhere in this Agreement</u>. For purposes of this Agreement, the following terms have meanings set forth in the sections indicated:

<u>Term</u>	Section
Alternative Transaction	7.1
Antitrust Laws	8.3(b)
Assignment and Assumption	4.2(b)
Agreement	
Assumed Liabilities	2.3
Auction	7.2(c)
Bankruptcy Case	Recitals
Bankruptcy Code	Recitals
Bankruptcy Court	Recitals
Bid	7.2(a)
Bid Deadline	7.2(a)
Break-Up Fee	7.1
Business Permits	5.14(b)
Cash Purchase Price	3.1
Closing	4.1
Closing Date	4.1
COBRA	9.2(d)
COBRA Beneficiaries	9.2(d)
Confidential Information	8.5(b)
Copyrights	1.1 (in Purchased Intellectual Property definition)
Cure Amount	2.5 .
Current Assets	5.26
December 31 Balance Sheet	5.4
December Financial Statements	5.4

<u>Term</u>	Section
Definitive Sale Documentation	7.2(a)
Deposit Amount	3.2
DIP Budget	8.2(a)
Employee and Related Liabilities	2.3
Employee Benefit Plans	5.11(a)
ERISA Affiliate	5.11(a)
Escrow Agent	3.2
Excluded Assets	2.2
Excluded Liabilities	2.4
Excluded Matter	1.1 (in Material Adverse Effect definition)
Expense Reimbursement	7.1
Financial Statements	5.4
HIPAA	9.2(d)
Indemnity Account	11.3
Indemnity Escrow Amount	11.3
Intellectual Property Assignments	4.2(g)
Interim Financial Statements	5.4
Inventory	2.1(c)
Key Employees	9.2(e)
Loss	11.2(a)
Marks	1.1 (in Purchased Intellectual Property definition)
Material Contract	5.10(a)
Multiemployer Plans	5.11(a)
Owned Property	5.7
Patents	1.1 (in Purchased Intellectual Property definition)
Pension Plan	5.11(a)
Personal Property Leases	5.8
Purchase Price	3.1
Purchased Assets	2.1
Purchaser	Recitals
Purchasers	Recitals
Purchaser Documents	6.2
Purchaser Indemnified Parties	11.2(a)
Purchaser Plans	9.2(b)
Qualified Bid	7.2(b)
Qualified Bidder	7.2(a)
Real Property	5.7
Real Property Lease	5.7
Related Persons	5.17
Sale Hearing	7.2
SEC Reports	5.19
Seller	Recitals Recitals
Sellers	5.2
Seller Documents	J.L

<u>Тетт</u>	Section
Seller Indemnified Parties Successful Bid Termination Date Title IV Plans Trade Secrets Transfer Taxes Transferred Employees Utility Equipment	11.2(b) 7.2(c)(iii) 4.4(a) 5.11 1.1 (in Purchased Intellectual Property definition) 12.1 9.1(a) 5.7

# 1.3 Other Definitional and Interpretive Matters.

(a) Unless otherwise expressly provided, for purposes of this Agreement, the following rules of interpretation shall apply:

<u>Calculation of Time Period</u>. When calculating the period of time before which, within which or following which any act is to be done or step taken pursuant to this Agreement, the date that is the reference date in calculating such period shall be excluded. If the last day of such period is a non-Business Day, the period in question shall end on the next succeeding Business Day.

Dollars. Any reference in this Agreement to "\$" shall mean U.S. dollars.

Exhibits/Schedules. The Exhibits and Schedules to this Agreement are hereby incorporated and made a part hereof and are an integral part of this Agreement. All Exhibits and Schedules annexed hereto or referred to herein are hereby incorporated in and made a part of this Agreement as if set forth in full herein. Any capitalized terms used in any Schedule or Exhibit but not otherwise defined therein shall be defined as set forth in this Agreement.

Gender. Any reference in this Agreement to gender shall include all genders.

Headings. The provision of a Table of Contents, the division of this Agreement into Articles, Sections and other subdivisions and the insertion of headings are for convenience of reference only and shall not affect or be utilized in construing or interpreting this Agreement. All references in this Agreement to any "Section" are to the corresponding Section of this Agreement unless otherwise specified.

Herein. The words such as "herein," "hereinafter," "hereof," and "hereunder" refer to this Agreement as a whole and not merely to a subdivision in which such words appear unless the context otherwise requires.

Including. The word "including" or any variation thereof means "including, without limitation" and shall not be construed to limit any general statement that it follows to the specific or similar items or matters immediately following it.

(b) The parties hereto have participated jointly in the negotiation and drafting of this Agreement and, in the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as jointly drafted by the parties hereto and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provision of this Agreement.

# ARTICLE II

# PURCHASE AND SALE OF ASSETS; ASSUMPTION OF LIABILITIES

- 2.1 Purchase and Sale of Assets. On the terms and subject to the conditions set forth in this Agreement, at the Closing, Purchasers shall purchase, acquire and accept from Sellers, and each Seller shall and shall cause each other Seller to, sell, transfer, assign, convey and deliver to Purchasers or one or more of their designated Affiliates, all the right, title and interest in, to and under the Purchased Assets, free and clear of all Liens, to the maximum extent permissible under Sections 363 and 365 of the Bankruptcy Code. "Purchased Assets" shall mean all the assets, properties, rights and claims of every kind and description of each of the Sellers used in the Business (other than Excluded Assets), whether real or personal, tangible or intangible, vested or unvested, contingent or otherwise, wherever located and whether existing now or hereinafter acquired, as the same shall exist on the Closing Date, whether or not any of such assets, properties, rights or claims have any value for accounting purposes or are carried or reflected on or specifically referred to in books or financial statements of Sellers, including the following:
- (a) all accounts receivable other than any accounts receivable and any other rights to receive payments arising out of or in connection with any Excluded Asset or the sale or distribution of Excluded Products;
- (b) any agreements with the State of Utah providing for payments and subsidies to Sellers in connection with Sellers' operations in the State of Utah, including any rights to payments relating to periods prior to and after the Closing;
- (c) all raw material, supplies, manufactured goods, purchased goods, work in process, goods in process and finished goods, other than Excluded Products, owned by Sellers relating to the Business ("Inventory");
- (d) all deposits (including customer deposits and security deposits for rent, electricity, telephone or otherwise) and prepaid charges and expenses, or portions thereof, of Sellers relating to the Business other than any deposits or prepaid charges and expenses paid in connection with or relating to any Excluded Assets or sales of Excluded Products;
- (e) all rights of Sellers in the Real Property, together with all facilities, buildings, improvements, fixtures and other appurtenances thereto and rights in respect thereof;

- (f) the Furniture and Equipment to the extent owned by Sollers;
- (g) all Purchased Intellectual Property;
- (h) the Purchased Contracts;
- (i) all Documents that are used in, held for use in or intended to be used in, or that arise primarily out of, the Business or the Purchased Assets, including Documents relating to Products, services, marketing, advertising, promotional materials, Purchased Intellectual Property, personnel files for Transferred Employees and all files, customer files and documents (including credit information), supplier lists, records, literature and correspondence, whether or not physically located on any of the premises referred to in clause (e) above, but excluding personnel files for Employees who are not Transferred Employees and any Documents solely related to or which are required to realize the benefits solely of any Excluded Assets;
  - (j) the Business Permits to the extent transferable;
- (k) all supplies owned by Sellers and used in connection with the Business;
- (l) all rights of Sellers under non-disclosure or confidentiality, noncompete, or non-solicitation agreements with employees and agents of Sellers or with third parties to the extent relating to the Business or the Purchased Assets (or any portion thereof);
- (m) all rights of Sellers under or pursuant to all warranties, representations and guarantees made by suppliers, manufacturers and contractors to the extent relating to Products sold, or services provided, to Sellers or to the extent affecting any Purchased Assets other than any warranties, representations and guarantees pertaining to any Excluded Assets;
  - (n) all information management systems of Sellers;
- (o) all goodwill and other intangible assets associated with the Business, including customer and supplier lists and the goodwill associated with the Purchased Intellectual Property; and
- (p) any rights, claims or causes of action of Sellers against third parties relating to the Business or the Purchased Assets arising out of events occurring on or prior to the Closing Date, but excluding those related to the Excluded Assets or the Excluded Products and those specified in <u>Section 2.2(j)</u>.
- 2.2 Excluded Assets. Nothing herein contained shall be deemed to sell, transfer, assign or convey the Excluded Assets to any Purchaser, and Sellers shall retain all right, title and interest to, in and under the Excluded Assets. "Excluded Assets" shall

mean all assets, properties, interests and rights of Sellers other than the Purchased Assets, including each of the following assets:

- (a) all cash, cash equivalents, bank deposits or similar cash items of Sellers;
- (b) all of Sellers' deposits or prepaid charges and expenses, or portions thereof, paid in connection with or relating to any Excluded Assets or the Excluded Products;
  - (c) the Excluded Contracts;
  - (d) the Specified Assets;
- (e) all intercompany obligations, liabilities and Indebtedness, including any note Indebtedness, owed to or by any Seller to or by any other Seller;
- (f) any Intellectual Property Rights of Sellers other than the Purchased Intellectual Property;
- (g) any (i) Tax Returns of Sellers books and records that Sellers are required by Law to retain or that relate to Tax Returns of Sellers; provided, however, that Purchasers shall have the right to make copies of any portions of such retained books and records that relate to the Business or any of the Purchased Assets; (ii) minute books; and (iii) documents relating to proposals to acquire the Business by Persons other than Purchasers;
- (h) any claim, right or interest of Sellers in or to any refund, rebate, abatement or other recovery for Taxes, together with any interest due thereon or penalty rebate arising therefrom (except that Purchasers shall be entitled to fifty percent (50%) of any refunds, rebates, abatements or other recovery for Taxes described in Section 12.1);
- (i) all insurance policies or rights to proceeds thereof relating to the assets, properties, business or operations of Sellers, except as otherwise agreed among Purchasers and Sellers prior to the Closing;
- (j) any avoidance actions of the Chapter 11 estates of the Sellers pursuant to Sections 540-549 of the Bankruptcy Code other than such causes of action that may exist against vendors, suppliers, or customers of the Sellers, which causes of action shall be Purchased Assets; and
  - (k) Sellers' equity interest in Twinlab Passion Ltd.
- 2.3 <u>Assumption of Liabilities</u>. On the terms and subject to the conditions set forth in this Agreement, from and after the Closing, Purchasers, or one or more of their designated Affiliates, will assume and pay, perform, discharge and be responsible for only the following Liabilities (the "<u>Assumed Liabilities</u>"):

- (a) the Liabilities set forth on Schedule 2.3(a) relating to (i) Transferred Employees and (ii) earned but unpaid broker's commissions relating to sales of Sellers' Products, all as described in Schedule 2.3(a), up to an aggregate amount of three million seven hundred thousand dollars (\$3,700,000), it being understood that such limit assumes that there are four hundred thirty (430) Transferred Employees and in the event that there are fewer than four hundred thirty (430) Transferred Employees, there shall be a proportionate reduction in such limit of Liabilities (the amounts described in this paragraph (a) of this Section 2.3 are referred to as the "Employee and Related Liabilities");
- (b) all executory Liabilities of Sellers under the Purchased Contracts to the extent such Liabilities relate to the period from and after the Closing;
- (c) Liabilities of the Sellers under "retail voucher programs" to reimburse retailers for the cost of advertisements, replacement products or similar incentive reimbursements that run or occur on or after the Closing Date;
  - (d) fifty percent (50%) of all Transfer Taxes; and
- (e) all Liabilities relating to amounts required to be paid by Purchasers hereunder.
- 2.4 <u>Excluded Liabilities</u>. None of the Purchasers will assume or be liable for any Liabilities of Sellers other than the Assumed Liabilities. All Liabilities of Sellers other than the Assumed Liabilities are referred to as "<u>Excluded Liabilities</u>." For the avoidance of doubt, the Excluded Liabilities shall include:
- (a) any Liability of any Sellers, except as specifically set forth in <u>Section 2.3</u>, relating to or arising out of the Business or any Purchased Asset and based upon, arising out of or resulting from any fact, circumstance, occurrence, condition, act or omission existing on or occurring on or prior to the Closing;
- (b) all Liabilities arising out of Excluded Assets, including the Specified Liabilities and Liabilities relating to Excluded Contracts;
- (c) all Liabilities for Taxes of Sellers other than as set forth in <u>Section 2.3(d)</u> above or Article XII below;
- (d) all Liabilities incurred in the Ordinary Course of Business existing prior to the filing of the Bankruptcy Case that are subject to compromise under the Bankruptcy Case;
- (e) all intercompany obligations, liabilities and Indebtedness, including any note Indebtedness, owed to any Seller by any other Seller;
- (f) all Liabilities relating to amounts required to be paid by Sellers hereunder;

- (g) any Liability of any Seller arising out of (A) any Legal Proceeding pending or threatened as of the date of the Closing, (B) any Legal Proceeding filed after the Closing Date relating to events occurring prior to such date, or (C) any actual or alleged violation by any Seller of any applicable Law prior to, or on account of, the Closing;
- (h) any trade payable of any Seller for goods or services received prior to the Closing Date or for expenses related to the transactions contemplated hereby;
- (i) except as set forth in <u>Schedule 2.3(a)</u>, any Liability of any Seller arising under any Employee Benefit Plan;
  - (j) any Indebtedness of Sellers;
- (k) any Liability of any Seller, whether now existing or hereafter arising, based upon, arising out of or otherwise in respect of (A) any violation of any Environmental Law to the extent existing on or prior to the Closing Date or (B) any other storage, use, sale, disposal or Release into the environment of any Hazardous Material on or prior to the Closing Date, or any exposure to any Hazardous Materials existing on or prior to the date Closing Date, including in each case costs of any clean-up of any environmental contamination or other condition and related governmental oversight cost;
- (I) any Liability of any Seller that relates to, or that arises out of, products manufactured, shipped or sold by or on behalf of such Seller on or prior to the Closing Date (including claims of negligence, personal injury, product damage, product liability, product warranties, promotional obligations, strict liability, product recall, safety or regulatory compliance, infringement of Intellectual Property Rights of a Person or any other claims (including workers' compensation, employer's liability or otherwise)), whether such Liability, obligation or commitment relates to or arises out of accidents, injuries or losses occurring on or prior to or after the Closing Date, including any Liability of any Seller arising out of or related to the Specified Liabilities;
- (m) any Liability, whether now existing or hereafter arising, in respect of claims for personal injury arising out of or related to the employment by any Seller of any Transferred Employee on or prior to the Closing Date, or to exposure or alleged exposure on or prior to the Closing Date to any materials or chemicals in the work place by any Transferred Employee or any other employees heretofore employed in the Business; and
- (n) any Liability under WARN for terminations of employees by Sellers prior to the Closing.
  - 2.5 Purchased Contracts, Personal Property Leases and Real Property Leases.
- (a) <u>Purchased Contracts</u>. At Closing and pursuant to Section 365 of the Bankruptcy Code, Sellers shall assign to Purchasers or one or more of their designated

Affiliates and Purchasers or one or more of their designated Affiliates shall assume from the applicable Sellers the Purchased Contracts.

(b) <u>Cure Amounts.</u> From and after the Closing, Purchasers will assume and pay, perform, discharge and be responsible for only those obligations and liabilities of Sellers under or related to the Purchased Contracts which accrue from and after the Closing Date, except to the extent any such obligation or liability is retained by Sellers pursuant to the terms hereof. Prior to or at the Closing, Sellers shall pay all cure amounts due with respect to the Purchased Contracts (collectively, the "<u>Cure Amounts</u>"). Except as expressly provided for herein, none of the Purchasers shall assume or pay, perform, discharge or be responsible for any of the obligations or liabilities of Sellers under the Purchased Contracts.

### 2.6 Further Conveyances and Assumptions.

- (a) From time to time following the Closing, Sellers shall, or shall cause their Affiliates to, make available to Purchasers such data in personnel records of Transferred Employees as is reasonably necessary for Purchasers to transition such employees into Purchasers' records.
- (b) From time to time following the Closing, Sellers and Purchasers shall, and shall cause their respective Affiliates to, execute, acknowledge and deliver all such further conveyances, notices, assumptions, releases and acquaintances and such other instruments, and shall take such further actions, as may be reasonably necessary or appropriate to assure fully to Purchasers and their respective successors or assigns, all of the properties, rights, titles, interests, estates, remedies, powers and privileges intended to be conveyed to Purchasers under this Agreement and the Seller Documents and to assure fully to Sellers and its Affiliates and their successors and assigns, the assumption of the liabilities and obligations intended to be assumed by any Purchaser under this Agreement and the Seller Documents, and to otherwise make effective the transactions contemplated hereby and thereby.
- 2.7 <u>Bulk Sales Law.</u> Purchasers hereby waive compliance by Sellers with the requirements and provisions of any "bulk-transfer" Laws of any jurisdiction that may otherwise be applicable with respect to the sale and transfer of any or all of the Purchased Assets to Purchasers.

#### ARTICLE III

### CONSIDERATION

3.1 <u>Consideration.</u> The aggregate cash consideration for the Purchased Assets shall be an amount in cash equal to sixty five million dollars (\$65,000,000) <u>less</u> the Zions Mortgage Balance if the Zions Mortgage is assumed by Purchasers (the "<u>Cash Purchase Price</u>") <u>plus</u> the Employee and Related Liabilities in an amount up to three million seven hundred thousand dollars (\$3,700,000) (the "<u>Purchase Price</u>"). The aggregate

17

consideration for the Purchased Assets shall also include the assumption of the other Assumed Liabilities.

- 3.2 <u>Purchase Price Deposit</u>. Upon entry of the Bidding Procedures Order, Purchasers shall deposit with a mutually acceptable escrow agent (the "<u>Escrow Agent</u>"), pursuant to the terms of the Escrow Agreement, by certified check or wire transfer of immediately available funds, the aggregate sum of two million dollars (\$2,000,000) (the "<u>Deposit Amount</u>"). Pursuant to the Escrow Agreement, the Deposit Amount, plus all accrued interest and investment income thereon, shall be either:
- (a) if the Closing shall occur, the Deposit Amount plus all accrued interest and investment income thereon shall be (i) applied towards the Purchase Price payable by Purchasers to Sellers under Section 3.3 hereof; and (ii) delivered to Sellers at Closing;
- (b) if this Agreement is validly terminated by Sellers pursuant to Section 4.4(e), the Deposit Amount, plus all accrued interest and investment income thereon, shall be delivered to Sellers;
- (c) if this Agreement is validly terminated (i) by Purchasers pursuant to Section 4.4(d), 4.4(f), 4.4(g), 4.4(i) or 4.4(j), (ii) by Purchasers or Sellers pursuant to Section 4.4(a) or 4.4(c), or (iii) by Purchasers and Sellers pursuant to Section 4.4(b), the Deposit Amount, together with all accrued interest and investment income thereon, shall in each case be returned to Purchasers; provided, however, in the event that this Agreement is terminated by Purchasers pursuant to Section 4.4(d), 4.4(f), 4.4(g), 4.4(i), or 4.4(j), then (x) the Deposit Amount, together with all accrued interest and investment income thereon shall be returned to Purchasers and (y) Sellers shall pay to Purchasers the Expense Reimbursement; and provided further, in the event this Agreement is terminated by Purchasers pursuant to Section 4.4(h), Sellers shall pay to Purchasers the Expense Reimbursement.

Notwithstanding the foregoing, nothing in this Section 3.2 shall be deemed to release any party from liability for (A) fraud or (B) willful breach of its obligations under this Agreement. In no event shall Sellers have any obligation to pay to Purchasers any amounts in respect of the Deposit Amount prior to payment by Purchasers of the Deposit Amount pursuant to Section 3.2; provided, that this provision shall not limit Sellers' obligations under this Agreement other than with respect to the return of the Deposit Amount.

3.3 Payment of Purchase Price. On the Closing Date, (i) Purchasers shall pay the Cash Purchase Price (less (x) the Deposit Amount, together with all accrued interest and investment income thereon, and (y) the Indemnity Escrow Amount) to Sellers, which shall be paid by wire transfer of immediately available funds into an account designated by Sellers, (ii) Purchasers shall deposit the Indemnity Escrow Amount with the Escrow Agent pursuant to the terms of the Escrow Agreement, and (iii) the Deposit Amount, together with all accrued interest and investment income thereon, shall be released in accordance with Section 3.2 and the terms of Escrow Agreement. The Indemnity Escrow

Amount shall be released in accordance with <u>Section 11.3</u> below and the terms of the Escrow Agreement.

3.4 Allocation of Purchase Price. Prior to the Closing Date, Purchasers shall prepare and deliver to Sellers a statement setting forth the allocation of the Purchase Price among Purchasers and the Purchased Assets. The parties agree that, for the purpose of reporting Taxes, they (i) shall report the transactions contemplated by this Agreement in accordance with such allocation statement, (ii) shall not take a position for the purpose of reporting Taxes inconsistent therewith, and (iii) shall cooperate in the preparation of any Tax Return filed in connection therewith.

### ARTICLE IV

### CLOSING AND TERMINATION

- 4.1 Closing Date. Subject to the satisfaction of the conditions set forth in Sections 10.1, 10.2 and 10.3 hereof (or the waiver thereof by the party entitled to waive such conditions), the closing of the purchase and sale of the Purchased Assets and the assumption of the Assumed Liabilities provided for in Article II hereof (the "Closing") shall take place at the offices of Weil, Gotshal & Manges LLP located at 767 Fifth Avenue, New York, New York (or at such other place as the parties may designate in writing) at 10:00 a.m. (Eastern time) on the date that is two (2) Business Days following the satisfaction or waiver of the conditions set forth in Article X (other than conditions that by their nature are to be satisfied at the Closing, but subject to the satisfaction or waiver of such conditions) and a date that is no more than ten (10) Business Days after the entry by the Bankruptcy Court of the Sale Order, unless another time or date, or both, are agreed to in writing by the parties hereto. The date on which the Closing shall be held is referred to in this Agreement as the "Closing Date."
  - 4.2 <u>Deliveries by Sellers</u>. At the Closing, Sellers shall deliver to Purchasers:
- (a) a duly executed Bill of Sale in customary form as agreed between Purchasers and Sellers prior to the Closing;
- (b) an Assignment and Assumption Agreement in customary form as agreed between Purchasers and Sellers prior to the Closing (the "Assignment and Assumption Agreement");
- (c) the duly executed and acknowledged Real Property Conveyance Documents;
- (d) certified resolutions of the Board of Directors or equivalent governing body of each of the Sellers approving and authorizing the transactions contemplated by this Agreement and the Ancillary Agreements;
- (e) officer's certificates, executed by a duly authorized officer of each of the Sellers (i) attaching true and complete copies of each Seller's articles of incorporation

19

and bylaws or equivalent organizational documents and (ii) as to the incumbency and signature of the officers of each Seller executing any applicable agreements or documents;

- (f) certificate of the Secretary of State of the state of incorporation for each Seller as to the due incorporation and good standing or equivalent certifications as to formation and standing of each Seller;
- (g) duly executed assignments of the Purchased Intellectual Property, in customary form to be agreed between Purchasers and Sellers prior to the Closing (the "Intellectual Property Assignments");
- (h) the officer's certificate required to be delivered pursuant to Sections 10.1(a) and 10.1(d);
- (i) a certified copy of the Sale Order and case docket reflecting that the Sale Order is in effect; and
- (j) all other instruments of conveyance and transfer, in form and substance reasonably acceptable to Purchasers, as may be necessary to convey the Purchased Assets to Purchasers.
- 4.3 <u>Deliveries by Purchasers</u>. At the Closing, Purchasers shall deliver to Sellers:
- (a) the Cash Purchase Price (less the Deposit Amount, together with all accrued interest and investment income thereon (which will be released from the Escrow Account to the Sellers in accordance with the terms of the Escrow Agreement), and the Indemnity Escrow Amount), in immediately available funds, as set forth in Section 3.3 hereof; and
- (b) the officer's certificate required to be delivered pursuant to Sections 10.2(a) and 10.2(c).
- 4.4 <u>Termination of Agreement</u>. This Agreement may be terminated prior to the Closing as follows:
- (a) by Purchasers or Sellers, if the Closing shall not have occurred by the close of business on November 14, 2003 (the "Termination Date"); provided, however, that, if the Closing shall not have occurred due to any continuation of the notice periods under the HSR Act or any other Antitrust Laws with respect to the transactions contemplated hereby and if all other conditions to the respective obligations of the parties to close hereunder that are capable of being fulfilled by the Termination Date shall have been so fulfilled or waived, then no party may terminate this Agreement prior to December 31, 2003; provided, that if the Closing shall not have occurred on or before the Termination Date due to a material breach of any representations, warranties, covenants

or agreements contained in this Agreement by Purchasers or Sellers, then the breaching parties may not terminate this Agreement pursuant to this Section 4.4(a);

- (b) by mutual written consent of Sellers and Purchasers:
- (c) by either Sellers or Purchasers, if there shall be any applicable Law that makes consummation of the transactions contemplated by this Agreement illegal or otherwise prohibited or if any final, nonappealable Order is entered by a Governmental Body of competent jurisdiction having valid enforcement authority permanently restraining, prohibiting or enjoining the consummation of, or imposing conditions upon, the transactions contemplated by this Agreement, or materially interfering with Purchasers' ability to operate the Business or a material portion thereof;
- (d) by Purchasers, so long as Purchasers are not in breach of any representation or warranty, or any covenant or agreement contained herein in any material respect, if there shall be a breach by Sellers of any representation or warranty, or any covenant or agreement contained in this Agreement which would result in a failure of a condition set forth in Section 10.1 or 10.3, and which breach cannot be cured or has not been cured within ten (10) Business Days after the giving of written notice by Purchasers to Sellers of such breach:
- (e) by Sellers, so long as no Seller is in breach of any representation or warranty, or any covenant or agreement contained herein in any material respect, if there shall be a breach by Purchasers of any representation or warranty, or any covenant or agreement contained in this Agreement which would result in a failure of a condition set forth in Section 10.2 or 10.3, and which breach cannot be cured or has not been cured within ten (10) Business Days after the giving of written notice by Sellers to Purchasers of such breach:
- (f) by Purchasers, if the Bankruptcy Case of any Seller is converted to Chapter 7 and a Chapter 7 trustee is appointed with respect to any Seller;
- (g) by Purchasers, (i) if any Seller executes and delivers a definitive agreement with respect to an Alternative Transaction, (ii) if the Bankruptcy Court enters an order approving any Alternative Transaction or (iii) five (5) business days after any Seller determines that a Bid other than that of Purchasers is the Successful Bid;
- (h) by Purchasers, if the Bankruptcy Court shall not have entered the Bidding Procedures Order by the later of (x) September 29, 2003 or (y) ten days after the formation of an official committee of unsecured creditors in the Sellers' Bankruptcy Casc;
  - (i) by Purchasers, in the event of a Material Adverse Effect; or
- (j) by Purchasers, if the Bankruptcy Court shall not have entered the Sale Order by the later of (x) October 28, 2003 or (y) five days after the Auction.

4.5 Procedure Upon Termination. In the event of termination of this Agreement and abandonment hereof by Purchasers or Sellers, or all parties, pursuant to Section 4.4 hereof, written notice thereof shall forthwith be given to each of the other parties, and this Agreement shall terminate, and the purchase of the Purchased Assets hereunder shall be abandoned, without further action by Purchasers or Sellers; provided, however, that the Escrow Agreement shall continue to remain in full force and effect until the Escrow Agreement is terminated pursuant to the terms thereof. If this Agreement is terminated as provided herein each party shall redeliver all documents, work papers and other material of any other party relating to the transactions contemplated hereby, whether so obtained before or after the execution hereof, to the party furnishing the same.

### 4.6 <u>Effect of Termination</u>.

- (a) In the event that this Agreement is validly terminated in accordance with Section 4.4, then each of the parties shall be relieved of their duties and obligations arising under this Agreement after the date of such termination; provided, however, that the obligations of the parties set forth in Section 3.2, Section 7.1, Section 8.5 and Article XIII hereof shall survive any such termination and shall be enforceable hereunder.
- (b) Nothing in this Section 4.6 shall relieve Purchasers or Sellers of any liability for a willful breach of this Agreement prior to the date of termination. The damages recoverable by the non-breaching party shall include all attorneys' fees reasonably incurred by such party in connection with the transactions contemplated hereby.

### ARTICLE V

# REPRESENTATIONS AND WARRANTIES OF SELLERS

The Sellers hereby represent and warrant, jointly and severally, to Purchasers that:

- Organization and Good Standing. Except as a result of the commencement of the Bankruptcy Case, each Seller is duly organized, validly existing and in good standing under the laws of the state of its organization and has all requisite corporate power and authority to own, lease and operate its properties and to carry on its business as now conducted. Each Seller is duly qualified or authorized to do business as a foreign entity and in good standing under the laws of each jurisdiction in which it owns or leases its real property and each other jurisdiction in which the conduct of its businesses or the ownership of its properties requires such qualification or authorization, except where the failure to be so qualified, authorized or in good standing would not have a Material Adverse Effect.
- 5.2 <u>Authorization of Agreement</u>. Subject to obtaining Bankruptcy Court approval pursuant to the Sale Order, each Seller has all requisite power, authority and

22

legal capacity to execute and deliver this Agreement and each Seller has all requisite power, authority and legal capacity to execute and deliver each Ancillary Agreement and each other agreement, document, or instrument or certificate contemplated by this Agreement or to be executed by any Seller in connection with the consummation of the transactions contemplated by this Agreement (the "Seller Documents"), to perform its obligations hereunder and thereunder and to consummate the transactions contemplated hereby and thereby. The execution, delivery and performance by each Seller of this Agreement and each other Seller Document have been duly authorized by all necessary corporate action on behalf of each such Seller. This Agreement has been, and each other Seller Document will be at or prior to the Closing, duly and validly executed and delivered by Sellers and (assuming the due authorization, execution and delivery by the other parties hereto and thereto) this Agreement constitutes, and each other Seller Document when so executed and delivered will constitute, legal, valid and binding obligations of Sellers, enforceable against Sellers in accordance with their respective terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium and similar laws affecting creditors' rights and remedies generally, and subject, as to enforceability, to general principles of equity, including principles of commercial reasonableness, good faith and fair dealing (regardless of whether enforcement is sought in a proceeding at law or in equity).

- 5.3 Conflicts. Except as set forth on Schedule 5.3, none of the execution and delivery by Sellers of this Agreement and the other Seller Documents, the consummation of the transactions contemplated hereby or thereby, or compliance by Sellers with any of the provisions hereof or thereof will conflict with, or result in any violation of or default (with or without notice or lapse of time, or both) under, or give rise to a right of termination, cancellation or acceleration of any obligation or to loss of a material benefit under, or give rise to any obligation of Sellers to make any payment under, or to the increased, additional, accelerated or guaranteed rights or entitlements of any Person under any provision of (i) the certificate of incorporation and bylaws of any Seller; (ii) any Contract or Permit to which any Seller is a party or by which any of the properties or assets of any Seller are bound, except to the extent any of the foregoing is not enforceable due to operation of applicable bankruptcy law or the Sale Order; (iii) any Order of any court, Governmental Body or arbitrator applicable to any Seller or any of the properties or assets of any Seller; or (iv) other than, in the case of clauses (ii), (iii) and (iv), such conflicts, violations, defaults, terminations, cancellations or accelerations that would not have a Material Adverse Effect.
- 5.4 Financial Information. Attached hereto as Schedule 5.4 are true and complete copies of (i) the audited consolidated balance sheet of Sellers as of December 31, 2002 (the "December 31 Balance Sheet"), (ii) the respective related consolidated statement of operations and statements of change in shareholders' equity and of cash flows for the twelve months ended December 31, 2002, together with the report thereon of Deloitte & Touche LLP, independent auditors (together with the December 31 Balance Sheet, the "December Financial Statements"), (iii) the unaudited consolidated balance sheet of Sellers (the "Balance Sheet") as of June 30, 2003 (the "Balance Sheet Date"), and (iv) the respective related consolidated statement of income and cash flows for the

six months ended June 30, 2003 (the "Interim Financial Statements"), which financial statements have been prepared on the same basis as the December Financial Statements. The December Financial Statements and the Interim Financial Statements are collectively referred to as the "Financial Statements.") Except as set forth on Schedule 5.4, the Financial Statements have been prepared in accordance with GAAP, and fairly present in all material respects the financial position and the results of operations, changes in shareholders' equity and cash flows of each of the Sellers, as applicable, as of the dates and for the periods indicated. Sellers have no Liabilities (whether absolute, accrued, contingent or otherwise), claims, obligations or other Liens relating to the Business or the Purchased Assets, except as disclosed in the Financial Statements, including the notes thereto, or Schedule 5.4 and except such Liabilities which, individually or in the aggregate, would not have a Material Adverse Effect.

5.5 Absence of Certain Developments. Since the Balance Sheet Date, except as set forth in Schedule 5.5, there has been no (a) Material Adverse Effect, (b) amendment or termination of, or default by any Seller or, to the Knowledge of Sellers, any other party, under any material contract or agreement relating to the Business to which any Seller is a party or by which any Seller is bound, (c) material destruction, material damage or other material loss to any of the Purchased Assets or any other material properties or assets of the Business, whether or not covered by insurance, (d) sale, lease, or other disposition of any properties or assets related to the Business except in the Ordinary Course of Business, (e) purchase, lease or other acquisition of any properties or assets related to the Business or other capital expenditures related to the Business other than in the Ordinary Course of Business, (f) increase in the compensation payable or to become payable to any of the employees or other personnel of any Seller that are employed by or render services to the Business except for normal periodic increases in the Ordinary Course of Business, (g) entering into or amendment of any employment agreement by any Seller, (h) change in accounting methods, unless required by GAAP, or principle or cost allocation procedures that affect the financial statements of the Sellers, (i) any material adverse change in any Seller's relations with any of the employees or other personnel of such Seller, or agents, customers or suppliers of the Business, or (i) agreement or commitment to take any action described in this Section <u>5.5</u>.

### 5.6 Ownership of Purchased Assets.

(a) Sellers have good title to all of the Purchased Assets free and clear of any Lien, other than Permitted Exceptions, except with respect to leased assets, as to which Sellers have good and marketable title to the leasehold interest. Upon consummation of the transactions contemplated hereby, Purchasers will have acquired good title in and to, or, in the case of assets which are leased or licensed, a valid, binding and enforceable leasehold interest or license in, each of the Purchased Assets, free and clear of all Liens, to the maximum extent provided in Sections 363 and 365 of the Bankruptcy Code. The Purchased Assets are in adequate condition and working order, reasonable wear and tear excepted.

- (b) The Purchased Assets constitute all the properties and assets relating to, used or held for use in connection with the Business, other than the Excluded Assets. There are no material assets or properties used primarily in the operation of the Business and owned by a third party that will not be leased or licensed to Purchasers under current leases or license agreements.
- Real Property. Schedule 5.7 sets forth a complete list of (i) all real property and interests in real property owned in fee by Sellers (individually, an "Owned Property" and collectively, the "Owned Properties"), and (ii) all real property and interests in real property leased by Sellers (individually, a "Real Property Lease," collectively, the "Real Property Leases," and together with the Owned Properties, the "Real Property") as lessee or lessor. Sellers have good, marketable and valid fee title to all Owned Property, free and clear of all Liens of any nature whatsoever except (A) the Zions Mortgage, and (B) Permitted Exceptions. Sellers have a valid and enforceable leasehold interest under each of the Real Property Leases, subject to applicable bankruptcy, insolvency, reorganization, moratorium and similar laws affecting creditors' rights and remedies generally and subject, as to enforceability, to general principles of equity (regardless of whether enforcement is sought in a proceeding at law or in equity). No Seller has received or given any notice of any default or event that with notice or lapse of time, or both, would constitute a default by any Seller under any of the Real Property Leases, except for defaults which Sellers are obligated to cure as of the Closing Date pursuant to Section 2.5 hereof which shall be so cured, or are unenforceable due to the Bankruptcy Case or the Sale Order, or which are not individually or in the aggregate, likely to result in a Material Adverse Effect. Seller has no knowledge of any default by any other party to a Real Property Lease. Sellers enjoy peaceful and undisturbed possession of the Real Property. All water, storm sewer, sanitary sewer, gas, electricity, telephone and other utility lines, pipes and other equipment necessary in order to operate the Real Property in the manner that such Real Property is presently being operated presently serve the Real Property and are in good working order (collectively, the "Utility Equipment"). All of the Utility Equipment is lawfully located on, under or at (a) the Real Property, (b) other property with respect to which the applicable Seller (or the applicable utility) is the beneficiary of a perpetual and irrevocable easement to use such property to run the applicable Utility Equipment and/or (c) other property owned by a Governmental Body on or under which the applicable Seller (or the applicable utility) is legally permitted to run the applicable Utility Equipment. There are no violations of any law, regulations or orders (including parking ordinances, site plan approvals, zoning and subdivision rules and regulations and land use development plans) issued by any Governmental Body having jurisdiction against or affecting the Real Property (or any portion thereof), the Business or any Seller which have not been cured, except for such violations which would not, individually or in the aggregate, result in a Material Adverse Effect. There are no applications currently pending by Seller for any change in, or otherwise pertaining to, (x) the zoning applicable to the Real Property or (y) the use of the Real Property. There are no leases or other occupancy agreements in effect with respect to all or any portion of the Owned Property. All material HVAC, electric, gas, fire-safety, plumbing, mechanical and other systems at the Real Property are in good, working condition which will allow the Business to be continued in the manner presently

being operated. There are no pending condemnation or eminent domain proceedings affecting the Real Property and no Seller has received any written notices of any, and to the Knowledge of Sellers there are no, threatened condemnation or eminent domain proceedings affecting the Real Property or any portion thereof. The Real Property has not suffered any casualty or other material damage that has not been repaired.

5.8 Tangible Personal Property. Schedule 5.8 sets forth all leases of personal property ("Personal Property Leases") relating to personal property used by Sellers in the Business or to which any Seller is a party or by which the properties or assets of any Seller are bound. Each Seller has a valid and enforceable leasehold interest under each Personal Property Lease under which it is a lessee, subject to applicable bankruptcy, insolvency, reorganization, moratorium and similar laws affecting creditors' rights and remedies generally and subject, as to enforceability, to general principles of equity (regardless of whether enforcement is sought in a proceeding at law or in equity). No Seller has received or given any notice of any default or event that with notice or lapse of time or both would constitute a default by any Seller under any Personal Property Lease, except for defaults which Sellers are obligated to cure as of the Closing Date pursuant to Section 2.5 hereof which shall be so cured, or are unenforceable due to the Bankruptcy Case or the Sale Order or which would not, individually or in the aggregate, result in a Material Adverse Effect.

# 5.9 <u>Intellectual Property</u>

- (a) To the Knowledge of Sellers and based on the Purchased Intellectual Property identified to the Sellers as of the date of this Agreement, Schedule 5.9(a) sets forth an accurate and complete list of all Patents, registered Marks, pending applications for registrations of any Marks and unregistered Marks, registered Copyrights and pending applications for registration of Copyrights owned or filed by Sellers and included in the Purchased Intellectual Property. Schedule 5.9(a) also identifies the owner of each such item of Purchased Intellectual Property and the jurisdictions in which each such item of Purchased Intellectual Property has been issued or registered or in which any such application for such issuance or registration has been filed.
- (b) Except as disclosed in Schedule 5.9(b), Sellers are the sole and exclusive owners of all right, title and interest in and to all of the Purchased Intellectual Property identified in Schedule 5.9(a). Except as set forth on Schedule 5.9(b), Sellers are the sole and exclusive owners of, or have valid and continuing rights to make, have made, use, sell, offer to sell and grant sublicenses to do any of the foregoing, as the case may be, of all Purchased Intellectual Property made, used, sold, offered for sale or licensed by Sellers in the Business as presently conducted, free and clear of all material Liens or obligations to others (it being understood between Sellers and Purchasers that the Security Agreement, dated May 25, 2001, between Twin Laboratories Inc. and Ross Blechman et al., is a material Lien). Sellers have made all fillings and recordations reasonable and in the ordinary course of business to protect and maintain their interest in the Patents, registered Marks and registered Copyrights, and applications therefor identified in Schedule 5.9(a).

- (c) To the Knowledge of Sellers, the sale, offer for sale, manufacture, import or license of Products of the Business, as the Business is presently conducted, and the present business practices and methods of Sellers, have not and do not infringe or constitute an unauthorized use, violation or misappropriation of any Intellectual Property Right or similar right, of any Person (including pursuant to any non-disclosure agreements or obligations to which Sellers or Sellers and any of their present or former employees is a party). Except as set forth on Schedule 5.9(c), the Purchased Intellectual Property owned by or licensed to Sellers includes all of the Intellectual Property Rights necessary to enable Sellers to conduct the Business in the manner in which such Business is currently being conducted.
- (d) Except with respect to licenses of commercial off-the-shelf Software, and except pursuant to the contracts, agreements and instruments listed in Schedule 5.9(d), Sellers are not required, obligated, or under any liability whatsoever, to make any payments by way of royalties, fees or otherwise to any owner, licensor of, or other claimant to any Purchased Intellectual Property with respect to the use thereof in connection with the conduct of the Business as currently conducted.
- (e) Except with respect to licenses of commercial off-the-shelf Software, Schedule 5.9(e) sets forth a complete and accurate list of all contracts, agreements and instruments to which any of the Sellers is a party (i) with respect to any Intellectual Property Licenses, or (ii) containing a covenant limiting a Seller's ability to exploit fully any of the Purchased Intellectual Property. To the Knowledge of Sellers, each of the agreements identified in Schedule 5.9(e) is currently in force.
- (f) To the Knowledge of Sellers and within the two (2) years prior to the date of this Agreement, neither any Trade Secret of Sellers nor any other non-public proprietary information, material to the Business as presently conducted, has been authorized to be disclosed or, has been actually disclosed by Sellers to any employee or any third Person other than pursuant to a confidentiality or non-disclosure agreement restricting its disclosure and use. Sellers have taken security measures reasonable in the industry to protect the secrecy, confidentiality and value of all the Purchased Intellectual Property. Also within the two years prior to the date of this Agreement, each employee, consultant and independent contractor of any Seller who has access to material non-public information of any Seller has entered into a written non-disclosure and invention assignment agreement with such Seller.
- (g) Except as set forth on Schedule 5.9(g), there are no pending or, to the Knowledge of Sellers, threatened actions, causes of action, claims, suits, proceedings, orders, writs, injunctions, or decrees which involve a claim of infringement, unauthorized use, or violation of the Purchased Intellectual Property by any Person against any Seller, or challenging Sellers' ownership or use, validity or enforceability of, any Purchased Intellectual Property. Except as set forth on Schedule 5.9(g), Sellers have not received written (including, without limitation, by electronic mail) notice of any such threatened claim and, to the Knowledge of Sellers, there are no facts or circumstances that would form the basis for any valid claim of infringement, unauthorized use, or violation by any

Person against any Seller, or challenging the ownership, use, validity or enforceability of any Purchased Intellectual Property.

(h) Except as set forth on <u>Schedule 5.9(h)</u>, Sellers have not made any claims that a Person is infringing, violating, misusing or misappropriating any Purchased Intellectual Property, and to the Knowledge of Sellers, no Person is infringing, violating, misusing or misappropriating any Purchased Intellectual Property owned by Sellers.

### 5.10 Material Contracts.

- (a) Schedule 5.10(a) sets forth all of the Contracts to which any Seller is a party or by which it is bound and that are related to the Business or by which the Purchased Assets may be bound or affected (collectively, the "Material Contracts"):
  - (i) all Contracts with any Affiliate or current or former officer or director of Sellers;
  - (ii) all Contracts for the sale of any of the assets of Sellers, other than in the Ordinary Course of Business;
  - (iii) all Contracts relating to the acquisition by Sellers of any operating business or the capital stock of any other Person;
  - (iv) all Contracts relating to incurrence of Indebtedness, guarantees or loans;
  - (v) all Contracts or related series of Contracts which involve the expenditure of more than \$50,000 in the aggregate or require performance by any party more than one year from the date hereof that, in either case, are not terminable by Sellers without penalty on less than 180 days' notice;
  - (vi) all Contracts with the twenty (20) largest customers (as measured by revenue for the six-month period ending on June 30, 2003) of the Business;
  - (vii) all Contracts with the twenty (20) largest suppliers (as measured by expenditures for the six-month period ending on June 30, 2003) of the Business:
  - (viii) all employment or consulting Contracts for employees, officers, directors or consultants:
  - (ix) all Contracts that grant a right of first refusal or similar right with respect to any of the Purchased Assets or any direct or indirect economic interest in any of the Purchased Assets in favor of any third party;

- (x) all joint venture, partnership or similar Contracts affecting the Business;
- (xi) all Contracts that restrict the ability of Sellers to conduct the Business in any place or in any manner or to solicit or retain any Persons as customers, employees or independent contractors;
- (xii) all Contracts affecting the ownership of, leasing of, title to, use of, or any leasehold or other interest in Real Property;
  - (xiii) all Contracts relating to the Purchased Intellectual Property;
- (xiv) all Contracts which result in the imposition of a Lien on the Purchased Assets;
- (xv) all Contracts requiring the consent of one or more third Persons with respect to the consummation of the transactions contemplated by this Agreement;
- (xvi) all Contracts entered into outside of the Ordinary Course of Business; and
  - (xvii) all other Contracts material to the Business.
- (b) There are no written or oral contracts that are material to the Business other than as set forth on Schedule 5.10(a). True and complete copies (including all modifications, amendments and supplements) of the Purchased Contracts listed on Schedule 5.10(a) have been provided or made available by Sellers to Purchasers. Sellers have not assigned, delegated or otherwise transferred to any third party any of their rights or obligations with respect to any Purchased Contract.
- (c) Subject to entry of the Sale Order and the assumption and assignment of such Material Contracts, and except as set forth on Schedule 5.10(c), each of the Material Contracts is, and will be as of the Closing, in full force and effect and is the legal, valid and binding obligation of Sellers, enforceable against them in accordance with its terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium and similar laws affecting creditors' rights and remedies generally and subject, as to enforceability, to general principles of equity (regardless of whether enforcement is sought in a proceeding at law or in equity). Except as set forth on Schedule 5.10(c), Sellers have not received or given any written notice of any default or event that with notice or lapse of time or both would constitute a default by any Seller under any Material Contract, other than defaults which Sellers are obligated to cure as of the Closing Date pursuant to Section 2.5 hereof and are listed on Schedule 5.10(c), or are unenforceable due to the Bankruptcy Case or the Sale Order.
  - 5.11 Employee Benefits.

- (a) Schedule 5.11(a) lists: (i) all material "employee benefit plans," as defined in Section 3(3) of ERISA, and all other material employee benefit arrangements or payroll practices, including bonus plans, employment, consulting or other compensation agreements, incentive, equity or equity-based compensation, deferred compensation arrangements, stock purchase, severance pay, sick leave, vacation pay, salary continuation, disability, hospitalization, medical insurance, life insurance, scholarship programs maintained by Sellers or to which Sellers contributed or is obligated to contribute thereunder for current or former employees of Sellers (the "Employee Benefit Plans") and (ii) all "employee pension plans," as defined in Section 3(2) of ERISA ("Pension Plan"), subject to Title IV of ERISA or Section 412 of the Code, maintained by any Seller and any trade or business (whether or not incorporated) which are or have ever been under common control, or which are or have ever been treated as a single employer, with Sellers under Section 414(b), (c), (m) or (o) of the Code ("ERISA Affiliate") or to which Sellers and any ERISA Affiliate contributed or has ever been obligated to contribute thereunder (the "Title IV Plans"). Schedule 5,11(a) separately sets forth each Title IV Plan which is a multiemployer plan as defined in Section 3(37) of ERISA ("Multiemployer Plans"), or has been subject to Sections 4063 or 4064 of ERISA ("Multiple Employer Plans").
- (b) Except for non-compliance which would not, individually or in the aggregate, have a Material Adverse Effect, all Employee Benefit Plans are in compliance with ERISA and the Code. Each Pension Plan which is intended to be qualified under Section 401(a) of the Code has received a favorable determination letter from the IRS and, to the Knowledge of Sellers, there are no circumstances that could result in revocation of any such favorable determination letter. There is no pending or, to the Knowledge of Sellers, threatened litigation relating to the Employee Benefit Plans. No Seller has engaged in a transaction with respect to any Employee Benefit Plan that could subject Sellers, the Business, the Purchased Assets or any Employee Benefit Plan to any Taxes or penalty imposed by either Section 4975 of the Code or Section 502(i) of ERISA. No action has been taken with respect to any Employee Benefit Plan to either terminate such Employee Benefit Plan or to cause distributions, other than in the Ordinary Course of Business to participants under such Employee Benefit Plan.
- (c) No Employee Benefit Plan is subject to Section 302 or Title IV of ERISA or Section 412 of the Code and no facts exist pursuant to which any Liability or obligation under Title IV of ERISA could be incurred with respect to the Business or the Purchased Assets.
- (d) All contributions required to be made under the terms of any Employee Benefit Plan have been timely made when due.
- (e) No Seller has any obligations for post-employment health or life benefits, except as set forth on Schedule 5.11(e). There are no restrictions on the rights of Sellers to amend or terminate any such Employee Benefit Plan without incurring any liability or obligation thereunder.

(f) Other than one employee employed by Twin Laboratories (UK) Ltd., no Employee Benefit Plan covers any employees employed outside the United States.

### 5.12 Labor.

- (a) Sellers will provide to the Purchasers, prior to the Closing, a complete and accurate list of all the employees or other personnel employed in or rendering services to the Business as of the most recent practicable date showing for each such employee or other personnel (i) the position(s) held, such employee or other personnel's service recognized by Sellers for purposes of any Employee Benefit Plans (including service with predecessor employers, if applicable, and any prior unabridged service with Sellers), such employee's or other personnel's eligible vacation, and any employment management, retention, severance, independent contractor, or other similar agreement to which such employee or other personnel is a party, (ii) whether such employee or other personnel is actively at work as of such date, and (iii) if such employee or other personnel is not actively at work as of such date, the nature of his or her absence (e.g., vacation, illness, short-term disability or leave of absence under the Family and Medical Leave Act of 1993) and his or her expected or required date of return to active service.
- (b) Except as set forth on Schedule 5.12(b), no Seller is party to, and none of the employees or other personnel employed in or rendering services to the Business are covered by any union, collective bargaining or other similar labor agreements and to the Knowledge of Sellers there is no indication that such employees or personnel desire to be covered by a collective bargaining agreement. As related to the Business, there is not presently pending or existing, and to the Knowledge of Sellers there is not threatened, (i) any strike, or slowdown, picketing, lockout, work stoppage or other material labor dispute, or (ii) any application for certification of a collective bargaining agent. As related to the Business, (A) no labor organization or group of employees of Sellers has made a pending demand for recognition or certification, and there are and have been no representation or certification proceedings or petitions seeking a representation proceeding, with the United States National Labor Relations Board or any other labor relations tribunal or authority, nor have any such demands, proceedings or petitions been brought or filed or threatened to be brought or filed within the past five years, (B) there have not been at any time within the last five years, any actual or, to the Knowledge of Sellers, any threatened organizing activities, strikes, picketing, work stoppages, slowdowns, lockouts, material arbitrations or material grievances, or other material labor disputes against or involving Sellers and (C) Sellers are in compliance in all material respects and with all applicable law with respect to employment and employment practices, terms and conditions of employment, wages and hours, independent contracting, immigration and occupational safety and health.
- (c) Except as set forth on <u>Schedule 5.12(c)</u>, there is not pending or, to the Knowledge of Sellers, threatened any action, cause of action, claim, suit, proceeding, order, writ, injunction or decree against Sellers or any of their Affiliates at law or in equity relating to the alleged violation of any law with respect to the subjects listed in <u>Section 5.12(b)</u> above. Except as set forth on <u>Schedule 5.12(c)</u>, to the Knowledge of

Sellers, there is no pending or threatened investigation of the Sellers by any federal, state, local or foreign agency responsible for the enforcement of labor or employment laws. As related to the Business, all material levies, assessments, and penalties made against Sellers pursuant to all applicable workers compensation legislation as of the Balance Sheet Date have been paid or have been reserved for or accrued on the Balance Sheet by Sellers and Sellers have not, as of the Closing Date, been reassessed under any such legislation.

- (d) Except as set forth on <u>Schedule 5.12(d)</u>, no employee has a written employment contract which promises such employee continued employment for any period of time.
- 5.13 <u>Litigation</u>. Except as set forth in <u>Schedule 5.13</u> and the Bankruptcy Case, there are no Legal Proceedings pending or, to the Knowledge of Sellers, threatened against Sellers, or to which any Seller is otherwise a party before any Governmental Body, which, if adversely determined, would reasonably be expected to result in a Material Adverse Effect. Sellers are not subject to any Order of any Governmental Body which would result in a Material Adverse Effect.

# 5.14 Compliance with Laws; Permits.

- (a) Each Seller and the Purchased Assets are in compliance with and the Business has been operated in compliance with all Laws, regulations and Orders relating to the Purchased Assets and the Business, including all Environmental Laws, and all rules and regulations of any Governmental Body or other Person which regulates any goods produced or services rendered by such Seller, except for such noncompliance as could not reasonably be expected to have a Material Adverse Effect. Except as set forth on Schedule 5.14(a), no Seller has received notice of any violation of any Law, regulation or Order, and neither Seller nor any of the Purchased Assets are in default under any order, writ, judgment, award, injunction or decree of any Governmental Body, applicable to the Purchased Assets or the Business, except for defaults which, individually or in the aggregate, would not have a Material Adverse Effect.
- (b) All material approvals, Permits, qualifications, franchises, Orders and registrations, together with all modifications, amendments, supplements and extensions thereof, of all Governmental Bodies and any other governmental entities or other Person that are necessary for Sellers to own the Purchased Assets, conduct the Business and use and occupy the Real Property (collectively, the "Business Permits") are valid and in full force and effect and are fully and freely transferable by Sellers to Purchasers. None of the Business Permits will be terminated or become terminable as a result of the transactions contemplated hereby. A true, complete and correct list of all Business Permits will be attached to this Agreement as Schedule 5.14(b) prior to the Closing.
  - 5.15 Intentionally omitted.

- 5.16 Insurance. Schedule 5.16 lists all policies of insurance owned or held by Sellers relating to the Business, or insuring the Purchased Assets, including the type and amount of coverage and the expiration dates of the policies. Except as set forth on Schedule 5.16, (i) current premiums and any other obligations under such insurance have been paid and all such policies are valid and enforceable and in full force and effect on the date hereof, and (ii) Sellers have not received any notice within the last ninety (90) days threatening suspension, revocation, modification or cancellation of any insurance policy or a material increase in any premium in connection therewith or informing Sellers that any coverage listed on Schedule 5.16 will or may not be available in the future on substantially the same terms as now in effect other than notices with respect to items that have been cured.
- Related Person Transactions No Affiliate, director, officer, employee, 5.17 agent, representative or other personnel of any Seller, nor any spouse or child or other family member of any of them, or any Person affiliated with any of them (each of the foregoing, a "Related Person"), has any interest in any of the Purchased Assets or any other assets or properties relating to the Business. No Related Person has owned, directly or indirectly, and whether on an individual, joint or other basis, any equity interest (other than ownership by a Person of less than 1% of the outstanding capital stock of a company as long as the Person is not otherwise involved in the management of such company) or any other financial or profit interest in a Person that has (i) had business dealings with the Business or (ii) engaged in competition with the Business. Except with respect to employment and compensation arrangements with employees in the Ordinary Course of Business described in Schedule 5.17, no Related Person is a party to any contract, agreement or arrangement with, or has any claim or right against, or owes any amounts to, Sellers in connection with the Business. All loans, payables and other amounts due between a Related Person and any Seller or any of their respective affiliates in connection with the Business are listed in Schedule 5.17.
- 5.18 <u>Financial Advisors</u>. Except as set forth on <u>Schedule 5.18</u>, no Person has acted, directly or indirectly, as a broker, finder or financial advisor for Sellers in connection with the transactions contemplated by this Agreement and no Person is entitled to any fee or commission or like payment from Purchasers in respect thereof. Sellers are jointly and severally responsible for the payment of any and all fees and disbursements of any such financial advisors.
- 5.19 Public Filings. The SEC Reports (as defined below), when filed with the Securities and Exchange Commission, complied as to form in all material respects with the requirements of the Securities Exchange Act of 1934, as amended, and the rules promulgated thereunder. As of their respective dates, the SEC Reports did not contain an untrue statement of material fact or omit to state a material fact required to be stated therein. The "SEC Reports" shall mean: (i) the Annual Report on Form 10-K for the fiscal year ended December 31, 2002; (ii) the Quarterly Report on Form 10-Q for the fiscal quarter ended March 31, 2003; (iii) the Current Reports on Form 8-K filed on May 15, 2003, June 27, 2003, August 15, 2003 and August 27, 2003; (iv) the Form 12b-25 filed on August 15, 2003; and (v) the Schedule 14A filed on April 25, 2003.

- 5.20 Product Liability. Except as set forth on Schedule 5.20, none of the Sellers have any material Liability that relates to, or that arises out of, products manufactured, shipped or sold by or on behalf of any Seller on or prior to the Closing date (including claims of negligence, personal injury, product damage, product liability, product warranties, promotional obligations, strict liability, product recall, safety or regulatory compliance, infringement of intellectual property or any other similar or related claims, whether such liability, obligation or commitment relates to or arises out of accidents, injuries or losses occurring on or prior to or after the Closing Date, including any Liability of any Seller arising out of or related to the Specified Liabilities. The reserves of six million five hundred thousand dollars (\$6,500,000) as of the Balance Sheet Date established with respect to certain product liability indemnity claims relating to the sale of products containing ephedra represent management's best estimate of the Sellers' contingent Liabilities with respect to such Liabilities.
- Taxes. Except as set forth on Schedule 5.21, all material Tax Returns required to be filed by or on behalf of Sellers relating to the Business have been timely filed, and all such Tax Returns are complete and accurate in all material respects; all Taxes due from Sellers for periods through the Closing Date, whether or not shown on any Tax Return, have been paid in full or an adequate provision has been made for any such Taxes on the Financial Statements (in accordance with GAAP); no Seller is currently the beneficiary of any extension of time within which to file any Tax Return relating to the Business; there are no pending claims made by any authority in any jurisdiction where Sellers do not file Tax Returns that any Seller is or may be subject to taxation by that jurisdiction based on the conduct or operation of the Business or the use or ownership of any of the Purchased Assets; Sellers have withheld and paid all Taxes relating to the Business required to have been withheld and paid in connection with amounts paid or owing to any third party; there is no audit, examination, litigation pending or, to the Knowledge of Sellers, threatened with respect to any Taxes of Sellers relating to the Business; no Seller has executed an extension or waiver of any statute of limitations on the assessment or collection of any Tax relating to the Business that is currently in effect; no rulings have been issued by or agreements entered into with any relevant governmental tax authority with respect to any Seller relating to the Business; no Seller is a party to any Tax allocation or sharing agreement with any other Person who is not a Seller or has any liability for the Taxes of any other Person who is not a Seller in connection with the Business; and there are no liens for Taxes on the Purchased Assets except for statutory liens for current Taxes not yet due and payable.
- 5.22 <u>Environmental</u>. Except as set forth on <u>Schedule 5.22</u> or as could not reasonably be expected, in the aggregate, to have a Material Adverse Effect:
- (a) Sellers' operations and properties, and the operations and properties of each of their respective subsidiaries, relating to the Business comply with all applicable Environmental Laws and Environmental Permits. Any past non-compliance with such Environmental Laws and Environmental Permits has been resolved without ongoing obligations or costs, and no circumstances exist that could (A) form the basis of an Environmental Action against Sellers, any of their respective subsidiaries or any of

their properties or (B) cause any of the Purchased Assets or the Business to be subject to any restrictions on ownership, occupancy, use or transferability under any Environmental Law.

- (b) None of the Purchased Assets or other property relating to the Business currently owned or operated by Sellers or any of their respective subsidiaries is, and, to the Knowledge of Sellers, no property formerly owned or operated by Sellers or any of their respective subsidiaries, is listed or proposed for listing on the NPL or on the CERCLIS or any analogous foreign, state or local list; there are no underground or aboveground storage tanks or any surface impoundments, septic tanks, pits, sumps or lagoons in which Hazardous Materials are being or have been treated, stored or disposed on any of the Purchased Assets or other property relating to the Business currently owned or operated by Sellers or any of their respective subsidiaries; and Hazardous Materials have not been released, discharged or disposed of on, under or from any of the Purchased Assets or other property currently owned or operated by Sellers or any of their respective subsidiaries.
- (c) Neither Sellers nor any of their respective subsidiaries has received a request for information or has been identified as a potentially responsible party relating to any actual or threatened Release of Hazardous Materials at any site, location or operation relating to the operations of the Business; neither Sellers nor any of their respective subsidiaries is undertaking, and has not completed, either individually or together with other potentially responsible parties, any investigation or assessment or remedial or response action relating to any actual or threatened Release of Hazardous Materials at any site, location or operation relating to the operations of the Business, either voluntarily or pursuant to the order of any Governmental Body or the requirements of any Environmental Law; and all Hazardous Materials generated, used, treated, handled or stored at, or transported to or from, any of the Purchased Assets or other property relating to the Business currently owned or operated by Sellers or any of their respective subsidiaries have been used, sold or disposed of in compliance with Environmental Laws.
- 5.23 <u>Consents and Approvals</u>. Subject to entry of the Sale Order, the execution, delivery and performance by each Seller of this Agreement and the Ancillary Agreements and the consummation of the transactions contemplated hereby and thereby do not require the consent or approval of, or filing with, any Governmental Body or other Person and no consent or waiver of any party to any Purchased Contract is required except for such consents, approvals and filings, the failure to obtain or make which would not, individually or in the aggregate, have a Material Adverse Effect.
- 5.24 Receivables and Payables. Sellers have disclosed to Purchasers a true, complete and correct list of all accounts receivable reflected on the books and records of Sellers as of the Balance Sheet Date and all accounts receivable created since the Balance Sheet Date, other than receivables collected, sold, transferred, assigned or otherwise settled in the Ordinary Course of Business. All such accounts receivable represent usual, customary and reasonable charges for services actually rendered or goods, products, equipment and supplies sold and delivered. All such accounts receivables arose from

bona fide transactions in the Ordinary Course of Business and were or are recorded, billed and collected in the Ordinary Course of Business. Sellers have invoiced all customers and business counterparties the full amount of all accounts receivable.

- 5.25 Inventory. The Inventory, taken as a whole, including Inventory purchased following the Balance Sheet Date, consists of a quality and quantity usable and salable in the Ordinary Course of Business, except for obsolete items and items of below-standard quality, all of which have been written off or written down to net realizable value on the Balance Sheet or on the accounting records of the Business as of the Closing Date, as the case may be. The Inventory levels, taken as a whole, have been maintained at such volumes, quantities and amounts as required for the conduct of the Business as conducted since the Balance Sheet Date. Inventory now on hand that was purchased after the Balance Sheet Date, taken as a whole, was purchased in the Ordinary Course of Business at a cost not exceeding market prices prevailing at the time of purchase. The Inventory, taken as a whole, is of such quality as to meet the quality control standards of the Business and any applicable quality control standards of any Governmental Body or material customer of the Business.
- 5.26 <u>Current Assets</u>. The accounts receivable and Inventory (collectively, the "<u>Current Assets</u>") of Sellers as of the Balance Sheet Date was fifty one million three hundred ninety one thousand dollars (\$51,391,000), calculated in accordance with GAAP. As of the Closing, the Sellers' Current Assets, net of any Inventory and accounts receivable relating to Excluded Products will be no less than forty seven million dollars (\$47,000,000), calculated in accordance with GAAP.
- 5.27 Suppliers and Customers. To the Knowledge of Sellers and except as disclosed to the Purchasers: (i) none of the twenty (20) largest suppliers of the Business has notified Sellers in writing of its intention to stop, or decrease the rate of, supplying materials, products or services to the Business and (ii) none of the twenty (20) largest customers of the Business has notified Sellers in writing of its intention to stop, or decrease the rate of, purchasing materials, products or services from the Business.
- 5.28 <u>Subsidiaries</u>. None of the Sellers owns any equity interests in any entity, other than (i) equity interests in corporations that are in the process of dissolving, all of which own no assets and have no material Liabilities and (ii) the equity interest in Twinlab Passion Ltd.
- 5.29 No Other Representations or Warranties; Schedules. Except for the representations and warranties contained in this Agreement (as modified by the Schedules hereto), (i) none of the Sellers or any other Person makes any other express or implied representation or warranty with respect to Sellers, the Business, the Purchased Assets, the Assumed Liabilities or the transactions contemplated by this Agreement, and Sellers disclaim any other representations or warranties, whether made by Seller or any Affiliate of Sellers or any of their respective officers, directors, employees, agents or representatives and (ii) the Purchased Assets are being transferred on a "where is" and, as to condition, "as is" basis. Except for the representations and warranties contained in this

Agreement (as modified by the Schedules hereto), each Seller (i) expressly disclaims and negates any representation or warranty, expressed or implied, at common law, by statute, or otherwise, relating to the condition of the Purchased Assets (including any implied or expressed warranty of merchantability or fitness for a particular purpose, or of conformity to models or samples of materials) and (ii) disclaims all liability and responsibility for any representation, warranty, projection, forecast, statement, or information made, communicated, or furnished (orally or in writing) to Purchasers or its Affiliates or representatives (including any opinion, information, projection, or advice that may have been or may be provided to Purchasers by any director, officer, employee, agent, consultant, or representative of Sellers or any of its Affiliates). Sellers make no representations or warranties to Purchasers regarding the probable success or profitability of the Business.

### ARTICLE VI

# REPRESENTATIONS AND WARRANTIES OF PURCHASERS

Purchasers hereby represent and warrant, jointly and severally, to Sellers that:

- 6.1 <u>Organization and Good Standing</u>. Each Purchaser is a corporation duly organized, validly existing and in good standing under the laws of its state of organization and has all requisite corporate power and authority to own, lease and operate its properties and to carry on its business as now conducted.
- Authorization of Agreement. Each Purchaser has full corporate power and 6.2 authority to execute and deliver this Agreement and each other agreement, document, instrument or certificate contemplated by this Agreement or to be executed by such Purchaser in connection with the consummation of the transactions contemplated hereby and thereby (the "Purchaser Documents"), and to consummate the transactions contemplated hereby and thereby. The execution, delivery and performance by each Purchaser of this Agreement and each other Purchaser Document have been duly authorized by all necessary corporate action on behalf of each such Purchaser. This Agreement has been, and each other Purchaser Document will be at or prior to the Closing, duly executed and delivered by each Purchaser and (assuming the due authorization, execution and delivery by the other parties hereto and thereto) this Agreement constitutes, and each other Purchaser Document when so executed and delivered will constitute, legal, valid and binding obligations of such Purchaser, enforceable against such Purchaser in accordance with their respective terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium and similar laws affecting creditors' rights and remedies generally, and subject, as to enforceability, to general principles of equity, including principles of commercial reasonableness, good faith and fair dealing (regardless of whether enforcement is sought in a proceeding at law or in equity).
  - 6.3 Conflicts; Consents of Third Parties.

37

- by Purchasers of this Agreement and the other Purchaser Documents, the consummation of the transactions contemplated hereby or thereby, or the compliance by Purchasers with any of the provisions hereof or thereof will (i) conflict with, or result in a violation of, any provision of the certificate of incorporation or by-laws of any Purchaser, (ii) conflict with, violate, result in the breach of, or constitute a default under, or result in the termination, cancellation or acceleration (whether after the filing of notice or lapse of time or both) of any material right or obligation of any Purchaser under, any Contract or Permit to which such Purchaser is a party or by which such Purchaser or its properties or assets are bound or (iii) violate any Order of any Governmental Body applicable to any Purchaser or any of the properties or assets of such Purchaser or (iv) any applicable Law, except, in the case of clauses (ii), (iii) and (iv), for such violations, breaches, defaults, terminations, cancellations or accelerations as would not have a material adverse effect on the ability of any Purchaser to perform its obligations under this Agreement or to consummate the transactions contemplated by this Agreement.
- (b) No consent, waiver, approval, Order, Permit or authorization of, or declaration or filing with, or notification to, any Person or Governmental Body is required on the part of Purchasers in connection with the execution and delivery of this Agreement or Purchaser Documents, the compliance by Purchasers with any of the provisions hereof or thereof, the consummation of the transactions contemplated hereby or the taking by Purchasers of any other action contemplated hereby, or for Purchasers to conduct the Business, except for compliance with the applicable requirements of the HSR Act.
- of Purchasers, threatened against Purchasers, or to which any Purchaser is otherwise a party before any Governmental Body, which, if adversely determined, would reasonably be expected to have a material adverse effect on the ability of Purchasers to perform their obligations under this Agreement or to consummate the transactions hereby. Purchasers are not subject to any Order of any Governmental Body which would reasonably be expected to have a material adverse effect on the ability of Purchasers to perform their obligations under this Agreement or to consummate the transactions contemplated hereby.
- 6.5 <u>Financial Advisors</u>. No Person has acted, directly or indirectly, as a broker or finder for Purchasers in connection with the transactions contemplated by this Agreement and no person is entitled to any fee or commission or like payment in respect thereof.
- 6.6 <u>Financial Capability</u>. At the Closing, Purchasers will have sufficient internal funds (without giving effect to any unfunded financing regardless of whether any such financing is committed) available to pay the Purchase Price and any expenses incurred by Purchasers in connection with the transactions contemplated by this Agreement, and will have the resources and capabilities (financial or otherwise) to perform their obligations hereunder and have not incurred any obligation, commitment,

restriction or Liability of any kind, which would have a material adverse effect on such resources and capabilities.

#### ARTICLE VII

### BANKRUPTCY COURT APPROVAL

- Break-up Fee. In consideration for Purchasers having expended 7.1 considerable time and expense in connection with this Agreement and the negotiation thereof and the identification and quantification of the assets of the Sellers, in the event that any of the Sellers (i) consummates a transaction in respect of a Bid, other than that of Purchasers, or (ii) sells, transfers, leases or otherwise disposes directly or indirectly, including through an asset sale, stock sale, merger, reorganization or other similar transaction, of all or substantially all or a material portion of the Purchased Assets (or agrees to do any of the foregoing) in a transaction or series of transactions within 12 months from the date hereof (either of clause (i) or (ii) being an "Alternative Transaction"), Sellers shall pay in cash in immediately available funds to Purchasers a break-up fee in an amount equal to three million four hundred thirty five thousand dollars (\$3,435,000) (the "Break-Up Fee") plus expense reimbursement equal to the reasonable costs and reasonable out-of-pocket fees and expenses of counsel, accountants, financial and other advisors incurred by Purchasers in connection with their legal, financial and business due diligence and the preparation and negotiation of this Agreement and any Ancillary Agreements (the "Expense Reimbursement"); provided that in no event shall the Break-Up Fee or Expense Reimbursement be payable to Purchasers if this Agreement is terminated by Sellers pursuant to Section 4.4(e). Sellers' obligation to pay the Break-Up Fee and Expense Reimbursement shall survive termination of this Agreement, and on the Petition Date, Sellers shall make a motion seeking an order that provides that the Break-Up Fee and Expense Reimbursement shall be paid as administrative expense claims in the Bankruptcy Case under Sections 503(b) and 507(a)(1) of the Bankruptcy Code and that (i) the Break-Up Fee shall be payable upon the earlier to occur of the closing of the Alternative Transaction and consummation of a plan of reorganization or liquidating plan of reorganization and (ii) the Expense Reimbursement shall be paid in accordance with this Agreement as soon as possible upon termination of this Agreement.
- 7.2 <u>Bidding Procedures Order.</u> Sellers acknowledge that this Agreement is the culmination of an extensive process undertaken by Sellers to identify and negotiate a transaction with a bidder who was prepared to pay the highest and best purchase price for the Purchased Assets while assuming or otherwise satisfying certain liabilities in order to maximize the value for Sellers' constituents. Set forth below are the bidding procedures (the "<u>Bidding Procedures</u>") to be employed with respect to this Agreement concerning the sale of the Purchased Assets of Sellers to Purchasers. The sale is subject to competitive bidding as set forth herein and approval by the Bankruptcy Court at a hearing under Sections 363 and 365 of the Bankruptcy Code (the "<u>Sale Hearing</u>"). The following overbid provisions and related bid protections are designed to compensate Purchasers for their efforts and agreements to date and to facilitate a full and fair process designed to maximize the value of the Purchased Assets for the benefits of Sellers' stakeholders.

- (a) <u>Bid Deadline</u>. All bids must be submitted to and actually received by Sellers, with a copy to Purchasers, not later than 11:00 a.m. on a date that is three days prior to the date scheduled by the Bankruptcy Court for the Auction (the "<u>Bid Deadline</u>"). For purposes of this Agreement, "<u>Bid</u>" shall mean a letter from a person who the Boards of Directors of Sellers has determined in the exercise of its fiduciary duty is financially able to consummate the purchase of the Purchased Assets (a "<u>Qualified Bidder</u>") stating that (i) such Qualified Bidder offers to purchase the Purchased Assets upon the terms and conditions set forth in a copy of this Agreement, together with all Exhibits and Schedules hereto (the "<u>Definitive Sale Documentation</u>"), marked to show those amendments and modifications to the Definitive Sale Documentation, including, but not limited, to, price and the time of closing, that such Qualified Bidder proposes; (ii) such Qualified Bidder is prepared to enter into and consummate the transaction within not more than ten days after approval by the Bankruptcy Court of the Sale Order; and (iii) such Qualified Bidder's offer is irrevocable until the closing of a purchase of the Purchased Assets.
- (b) <u>Qualified Bid</u>. Only Qualified Bids will qualify for consideration at the Auction (defined below). For purposes of this Agreement, a "<u>Qualified Bid</u>" is a Bid that:
  - (i) complies in all respects with 7.2(a);
  - (ii) has a cash component of at least an amount equal to the sum of (a) the cash component of Purchasers' Bid (including the Deposit Amount and the Indemnity Escrow Amount) plus (b) the Break-Up Fee plus (c) the Expense Reimbursement (for purposes of this subsection only, the Expense Reimbursement shall be estimated at one million dollars (\$1,000,000)) plus (d) such portion of the three million seven hundred thousand dollars (\$3,700,000) of Liabilities constituting Employee and Related Liabilities to the extent the Qualified Bid does not propose to assume all or any portion of such Liabilities plus (e) in the case of the initial Qualified Bid, two million dollars (\$2,000,000), and in the case of any subsequent Qualified Bids, one million dollars (\$1,000,000) over the preceding Qualified Bid;
  - (iii) is a proposal that Sellers determine, in the good faith opinion of the Boards of Directors of Sellers, and after taking into account the benefits of this Agreement to its constituencies (as well as the potential costs of pursuing other transactions) is not materially more burdensome or conditional than the terms of this Agreement and is in the best interest of the estate and its constituencies;
  - (iv) is substantially on the same or better terms and conditions as set forth in a copy of the Definitive Sale Documentation;
  - (v) is accompanied by satisfactory evidence of committed financing or other ability to perform and is not subject to any due diligence, financing or other contingencies;

- (vi) is accompanied by a cash deposit of two million dollars (\$2,000,000) (the "Deposit"), which shall be refunded by the Sellers in the event that such Qualified Bid is not a Successful Bid and which shall be credited towards the Purchase Price in the event such Bid is the Successful Bid, provided that Purchasers shall pay the Deposit upon the entry of an order approving the Bidding Procedures in accordance with Section 3.2 above; and
- (vii) does not entitle or seek to entitle the Qualified Bidder to any expense reimbursement or termination or similar fee.

If Sellers do not receive any Qualified Bids, Sellers will report the same to the Bankruptcy Court and will proceed with a sale and assignment of the Purchased Assets to Purchasers. This Agreement executed by Purchasers shall constitute a Qualified Bid for all Purposes.

### (c) Auction, Bidding Increments, and Bids Remaining Open.

- If Sellers receive a Qualified Bid or Qualified Bids, Sellers (i) will conduct an auction (the "Auction") in New York City at a location specified by Sellers on the date that is no later than the later of (i) October 23, 2003 or (ii) twenty (20) days after entry of the Bidding Procedures Order. Only Purchasers, Sellers, any representative of any official committee appointed in the Bankruptcy Case, and any Qualified Bidders who have timely submitted Oualified Bids shall be entitled to attend the Auction, and only Purchasers and Qualified Bidders will be entitled to make any subsequent Qualified Bids at the Auction. Bidding at the Auction will continue until such time as the highest and best offer is determined, provided that Qualified Bidders shall be required to bid in cash (i.e. no party shall be entitled to "credit bid" claims that they may have against the estate). Sellers may announce at the Auction additional procedural rules that are reasonable under the circumstances (e.g., the amount of time allotted to make subsequent overbids) for the Auction so long as such rules are not inconsistent with these Bidding Procedures.
- (ii) Sellers shall provide Purchasers with copies of all Qualified Bids. In addition, Sellers will inform Purchasers and each Qualified Bidder who has expressed its intent to participate in the Auction of the identity of all Qualified Bidders that may participate at the Auction.
- (iii) Prior to the conclusion of the Auction, Sellers shall in their business judgment determine the highest and best Qualified Bid (the "Successful Bid") and shall seek Bankruptcy Court authority to enter into the transaction evidenced by the Successful Bid.
- 7.3 Bankruptcy Court Filings. On the Petition Date, Sellers shall file with the Bankruptcy Court a motion seeking entry of the Sale Order and the Bidding Procedures Order. Purchasers agree that they will promptly take such actions as are reasonably

requested by Sellers to assist in obtaining the Sale Order and the Bidding Procedures Order for the purposes, among others, of providing necessary assurances of performance by Purchasers under this Agreement and demonstrating that each Purchaser is a "good faith" purchaser under Section 363(m) of the Bankruptcy Code. In the event the entry of the Sale Order or the Bidding Procedures Order shall be appealed, Sellers and Purchasers shall use their respective reasonable efforts to defend such appeal.

- 7.4 Section 363(m) Protection. Sellers shall use commercially reasonable efforts to cause the Bankruptcy Court to enter a Sale Order which provides, in either the findings of fact or conclusions of law to be made by the Bankruptcy Court as part of the Sale Order, that the transactions contemplated by this Agreement are undertaken by Purchasers and Sellers at arm's length, without collusion and in good faith within the meaning of Section 363(m) of the Bankruptcy Code, and such parties are entitled to the protections of Section 363(m) of the Bankruptcy Code.
- No Shop. Prior to the entry of the Bidding Procedures Order on the Bankruptcy Court's docket, and except as otherwise required by this Agreement or by the Bankruptcy Code, Sellers shall not, directly or indirectly, through any officer, director, employee, agent or adviser, solicit any Alternative Transaction or participate in any negotiations with respect to any Alternative Transaction, regardless of whether any proposal or offer for any Alternative Transaction was unsolicited. Subsequent to the entry of the Bidding Procedures Order on the Bankruptcy Court's docket, Sellers shall not, directly or indirectly, through any officer, director, employee, agent or adviser, solicit any Alternative Transaction or participate in any negotiations with respect to any Alternative Transaction other than as expressly permitted under the Bidding Procedures Order; provided, however, that nothing herein shall prohibit Sellers from seeking to sell, pursuant to a Qualified Bid in connection with the Auction established in the Bidding Procedures Order, the Purchased Assets.

#### ARTICLE VIII

### **COVENANTS**

8.1 Access to Information. Sellers agrees that, prior to the Closing Date, Purchasers shall be entitled, through their respective officers, employees and representatives (including its legal advisors and accountants), to make such investigation (including review of documentation and inspections of Real Property, but only if such inspection and testing does not interfere with Seller's business and operations at such property and Purchasers provide Sellers with proof of Purchasers' insurance covering such inspections and testing) of the properties, businesses and operations of the Business and such examination of the books, records and financial condition of the Business, the Purchased Assets and the Assumed Liabilities as they reasonably request and to make extracts and copies of such books and records. Any such investigation and examination shall be conducted during regular business hours upon reasonable advance notice and under reasonable circumstances and shall be subject to restrictions under applicable Law. Sellers shall cause their officers, employees, consultants, agents, accountants, attorneys

and other representatives to cooperate with Purchasers and Purchasers' representatives in connection with such investigation and examination, and Purchasers and their representatives shall cooperate with Sellers and their representatives and shall use their reasonable efforts to minimize any disruption to the Business. Notwithstanding anything herein to the contrary, no such investigation or examination shall be permitted to the extent that it would require Sellers or any Affiliate to disclose information subject to attorney-client privilege or conflict with any confidentiality obligations to which Sellers or any Affiliates are bound, from which Sellers will use commercially reasonable efforts to be released.

# 8.2 Conduct of the Business Pending the Closing.

- (a) Prior to the Closing, except (1) as otherwise expressly contemplated by this Agreement or (2) with the prior written consent of Purchasers (which consent shall not be unreasonably withheld or delayed) or (3) as may be required to comply with the debtor in possession budget agreed upon in connection with the debtor in possession credit facility as such budget may be amended from time to time with the consent of the Purchasers (provided, however, that Purchasers shall not unreasonably withhold or delay their consent to such amendment) (the "DIP Budget"), Sellers shall:
  - (i) operate the Business in a manner consistent with the DIP Budget;
  - (ii) maintain all of the Purchased Assets in good repair, condition and working order, reasonable wear and tear excepted, for the uses in which they are currently used in the Business;
  - (iii) maintain in full force and effect all policies of insurance covering the Business and the Purchased Assets;
  - (iv) conduct the Business only in the Ordinary Course of Business; and
  - (v) use their reasonable efforts to (A) preserve the present business operations, organization and goodwill of the Business, and (B) preserve the present relationships with customers and suppliers of the Business.
- (b) Except (1) as may be required by the DIP Budget, (2) as otherwise expressly contemplated by this Agreement or (3) with the prior written consent of Purchaser (which consent shall not be unreasonably withheld or delayed), solely as it relates to the Business, Sellers shall not:
  - (i) (A) materially increase the annual level of compensation of any employee, (B) increase the annual level of compensation payable or to become payable by any Seller to any of its respective executive officers, (C) grant any unusual or extraordinary bonus, benefit or other direct or indirect compensation to any employee, director or consultant, (D) increase the coverage

or benefits available under any (or create any new) Employee Benefit Plan or (E) enter into or amend any employment, deferred compensation, severance, consulting, non-competition or similar agreement (or amend any such agreement) to which any Seller is a party or involving a director, officer or employee of any Seller in his or her capacity as a director, officer or employee of such Seller, except, in each case, in the Ordinary Course of Business, or as required by applicable Law from time to time in effect or by any of the Employee Benefit Plans or Pension Plans:

- (ii) make any material change to any of their methods of accounting unless required by GAAP;
  - (iii) acquire or agree to acquire any business;
- (iv) sell (including by sale-leaseback), lease, transfer, license (whether on an exclusive or non-exclusive basis), mortgage or otherwise dispose of, encumber or subject to any Lien, any Purchased Assets or interests therein, other than sales of Inventory in the Ordinary Course of Business;
- (v) sell, transfer, or otherwise dispose of, encumber or subject to any Lien, any capital stock of any or interest in any Person;
- (vi) commence or continue any "going out of business" or inventory liquidation sales or offer any discounted sales of Inventory, other than quantity discounts heretofore offered by Sellers in the usual, regular and Ordinary Course of Business, consistent with past practice;
- (vii) enter into any contract, license or other agreement that contains any provision that, as a result of the consummation of the transactions contemplated by this Agreement or the Ancillary Agreements, would (assuming that the other party's consent or approval is not obtained, to the extent required) result in any penalty, additional payments or forfeiture that would be payable on or after the Closing Date;
- (viii) modify, amend or supplement any Purchased Contracts, or terminate any Purchased Contracts except in accordance with the terms thereof;
- (ix) make any single new commitment or related series of commitments or increase any single previous commitment or related series of commitments for capital expenditures for the Business in an amount exceeding fifty thousand dollars (\$50,000) nor make any commitments or increase any previous commitments for capital expenditures for the Business in an amount exceeding fifty thousand dollars (\$50,000) in the aggregate;
- (x) accelerate or delay the sale of the goods or services of the Business, except as may be necessary in the Ordinary Course of Business;

- (xi) sell, assign, transfer, license, abandon or convey any Purchased Intellectual Property except in the Ordinary Course of Business;
- (xii) except as would not adversely affect Purchasers, the Business and the Purchased Assets, settle any material claim regarding Taxes;
- (xiii) perform any act which would cause any representation, warranty, covenant, condition or agreement of Sellers in this Agreement or the Ancillary Agreements to be or become untrue in any respect or intentionally omit to take any action necessary to prevent any such representation, warranty, covenant, condition or agreement from being or becoming untrue in any respect, and will comply in all respects with all Laws applicable to the Business, the Purchased Assets and all of the Permits;
- (xiv) waive or compromise any material claim or right or fail to renew any Permit; and
  - (xv) agree to do anything prohibited by this Section 8.2.
- (c) Sellers will promptly notify Purchasers in writing of (i) any event known to Sellers which would render any representation or warranty of Sellers contained in this Agreement or any Ancillary Agreement, if made on or as of the date of such event or the Closing Date, untrue or inaccurate, (ii) any change, condition or event that has or could reasonably be expected to have a Material Adverse Effect or (iii) any failure of Sellers to comply with or satisfy any covenant, condition or agreement to be complied with or satisfied hereunder. Notwithstanding the disclosure to Purchasers of any matter pursuant to this Section 8.2(c), Sellers shall not be relieved of any Liability for, nor shall the provision of such information by Sellers to Purchasers be deemed a waiver of, the breach of any representation, warranty, covenant, condition or agreement of Sellers contained in this Agreement or any Ancillary Agreement.

#### 8.3 Regulatory Approvals.

- (a) If legally required, Purchasers and Sellers shall, make or cause to be made all filings required of them or any of their respective subsidiaries or Affiliates under the HSR Act or other Antitrust Laws with respect to the transactions contemplated hereby as promptly as practicable. Sellers and Purchasers may, as each deems advisable and necessary, reasonably designate any competitively sensitive material provided to the other under this Section 8.3 as "outside counsel only." Such materials and the information contained therein shall be given only to the outside legal counsel of the recipient and will not be disclosed by such outside counsel to employees, officers, or directors of the recipient, unless express written permission is obtained in advance from the source of the materials (Sellers or Purchasers, as the case may be).
- (b) If a filing under the HSR Act is legally required, each Purchaser and Seller shall use commercially reasonable efforts to resolve such objections, if any, as may be asserted by any Governmental Body with respect to the transactions contemplated by

this Agreement under the HSR Act, the Sherman Act, as amended, the Clayton Act, as amended, the Federal Trade Commission Act, as amended, and any other applicable Laws that are designed to prohibit, restrict or regulate actions having the purpose or effect of monopolization or restraint of trade (collectively, the "Antitrust Laws").

## 8.4 Further Assurances.

Each Seller and Purchaser shall use its commercially reasonable efforts to (i) take all actions necessary or appropriate to consummate the transactions contemplated by this Agreement and (ii) cause the fulfillment at the earliest practicable date of all of the conditions to their respective obligations to consummate the transactions contemplated by this Agreement.

## 8.5 Confidentiality.

- (a) Prior to the Closing, Purchasers shall keep confidential, and cause their Representatives to keep confidential, any and all Confidential Information relating to Sellers and Sellers' business and operations. Following the Closing, Purchasers shall keep confidential, and cause their respective Affiliates, directors, officers, employees, agents and controlling Persons (collectively, their "Representatives") to keep confidential, any and all Confidential Information relating to the Excluded Products, Excluded Liabilities, Excluded Assets and the business and operations of the Sellers, other than the Business. Sellers shall, at all times, keep confidential, and cause their Representatives to keep confidential, any and all Confidential Information regarding the Purchasers and the Purchasers' business and operations. Following the Closing, Sellers shall keep confidential, and cause their Affiliates to keep confidential, any and all Confidential Information regarding the Business, the Purchased Assets and the Assumed Liabilities. In the event this Agreement is terminated in accordance with Article IV, Purchasers shall, at all times, keep confidential, and cause their Representatives to keep confidential, any and all Confidential Information regarding the Sellers. The provisions of this Section 8.5 supercede any confidentiality or non-disclosure agreement entered into between any Seller and Purchasers, which agreement is hereby terminated in its entirety and of no further force or effect.
- (b) As used herein, "Confidential Information" means any and all confidential information of the other party or parties that one party either receives from another party in connection herewith or in the course of performing its obligations hereunder (except that such information may be shared, on a confidential basis, with the party's attorneys, lenders and auditors); provided, however, that Confidential Information does not include information which (A) is or becomes public other than as a result of a breach of the terms of this Section 8.5, (B) was available to the disclosing party on a non-confidential basis prior to its disclosure by such party or (C) becomes available to a disclosing party on a non-confidential basis from a Person (i) who is not a party to this agreement, (ii) who is not otherwise known to the disclosing party to be in violation of a confidentiality agreement with the party whose information is the subject of such

disclosure, or (iii) not otherwise known to be prohibited from transmitting the information to the disclosing party.

- (c) Notwithstanding anything else in this <u>Section 8.5</u>, a party hereto may disclose Confidential Information of another party hereto as required by Law or Order to the extent so required (in which case the legally obligated party shall promptly notify the other parties and give the other parties an opportunity to oppose such disclosure prior to the making of such disclosure).
- (d) Notwithstanding anything to the contrary set forth herein or in any other agreement to which the parties hereto are parties or by which they are bound, the obligations of confidentiality contained herein and therein, as they relate to the transactions described in this Agreement, shall not apply to the Tax structure or Tax treatment of the transactions described in this Agreement, and each party hereto (and any employee, representative, or agent of any party hereto) may disclose to any and all persons, without limitation of any kind, the Tax structure and Tax treatment of the transactions described in this Agreement and all materials of any kind (including opinions or other Tax analysis) that are provided to such party relating to such Tax treatment and Tax structure; provided, however, that such disclosure shall not include the name (or other identifying information not relevant to the Tax structure or Tax treatment) of any person and shall not include information for which nondisclosure is reasonably necessary in order to comply with applicable securities laws.
- Preservation of Records. Sellers and Purchasers agree that each of them 8.6 shall preserve and keep the records held by it or their respective Affiliates relating to the Business for a period of seven (7) years from the Closing Date and shall make such records and personnel available to the other as may be reasonably required by such party in connection with, among other things, administration of the Bankruptcy Case, preparation of the plan and disclosure statement for filing in the Bankruptcy Case and confirmation of the plan in the Bankruptcy Case, any insurance claims by, Legal Proceedings or Tax audits against or governmental investigations of Sellers or Purchasers or any of their Affiliates or in order to enable Sellers or Purchasers to comply with their respective obligations under this Agreement, each Ancillary Agreement and each other agreement, document or instrument contemplated hereby or thereby; provided, however, that no providing party shall be required to (i) incur any out-of-pocket expenses, (ii) provide information or records under circumstances which the providing party believes in its sole reasonable determination may expose it to Liability to any Person or may prejudice any interest, commercial, legal or otherwise, of the providing party or (iii) take any action that in the providing party's sole reasonable determination unreasonably interferes with its business. Without limiting the foregoing, Purchasers agree to make available to Sellers on a reasonable and as-needed basis Transferred Employees, including former members of management of Sellers, to assist Sellers undertaking of the foregoing, subject to reasonable compensation to be agreed upon between Purchasers and Sellers. In the event Sellers, on the one hand, or Purchasers, on the other hand, wish to destroy such records after that time, such party shall first give ninety (90) days prior written notice to the other and such other party shall have the right at its option and

expense, upon prior written notice given to such party within that ninety (90) day period, to take possession of the records within one hundred and eighty (180) days after the date of such notice.

- 8.7 Publicity. Neither Sellers, on the one hand, nor Purchasers, on the other hand, shall issue any press release or public announcement concerning this Agreement or the transactions contemplated hereby without obtaining the prior written approval of the other party hereto, which approval will not be unreasonably withheld or delayed, unless, in the sole judgment of such Purchasers or Sellers, disclosure is otherwise required by applicable Law or with respect to filings to be made with the Bankruptcy Court in connection with this Agreement or by the applicable rules of any stock exchange on which Purchasers, or Sellers lists securities, provided that the party intending to make such release shall consult with the other party with respect to the text thereof and give the other party reasonable opportunity to review such text.
- 8.8 <u>Use of Name</u>. Following the Closing Date, except as otherwise required by the Bankruptcy Court or the Bankruptcy Code and the regulations promulgated thereunder, Sellers shall not use any corporate, trade, or service name that includes Purchased Intellectual Property or, in each case, any confusingly similar words.
- 8.9 Receipt of Property Relating to Purchased Assets. If Sellers or any of their respective Affiliates, or any other Person acting for or in concert with any of the foregoing Persons, shall receive any money, check, note, draft, instrument, payment or other property relating to or as proceeds of the Purchased Assets or the Assumed Liabilities or any part thereof, each such Person shall receive all such items in trust for, and as the sole and exclusive property of, Purchasers and, immediately upon receipt thereof, shall notify Purchasers in writing of such receipt and shall remit the same (or cause the same to be remitted) in kind to Purchasers in the manner specified by Purchasers.
- 8.10 <u>Consents</u>. Sellers shall use their best efforts to obtain (a) entry of the Bidding Procedures Order and the Sale Order by the Bankruptcy Court, and (b) the consents or approvals of each other Governmental Body and each other Person necessary for the consummation of the transactions contemplated by this Agreement and the Ancillary Agreements.
- law or are immaterial to the Business, Sellers shall cause the issuance or transfer of all Permits relating to the Business to Purchasers on or before the Closing Date in form and substance the same as the Permits that were held by Sellers. Sellers shall give and make all required notices and reports to each appropriate Governmental Body and other Persons with respect to the Permits that may be necessary for the sale and purchase of the Business and the ownership, operation and use of the Purchased Assets by Purchasers following the Closing.

- 8.12 <u>Assignment of Contracts</u>. The Sale Order shall provide that effective on the Closing Date the Sellers shall assume, cure all monetary defaults with respect to and assign the Purchased Contracts to, Purchasers, subject to the Purchasers showing adequate assurance of future performance.
- 8.13 Plan of Reorganization. The Sellers and each of their Affiliates shall use their best efforts to ensure that any plan of reorganization or liquidation proposed by the Sellers or any of their affiliates in their Bankruptcy Cases shall (i) provide for a release of the Purchasers from any product liability or similar claims pursuant to Section 1141(c) of the Bankruptcy Code, (ii) enjoin any suits brought against the Purchasers in connection with such claims and (iii) provide that existing insurance policies will continue to be used to pay any product liability claims.
- 8.14 Adjustments to Schedules. Purchasers and Sellers agree to work together in good faith to make adjustments to the Schedules from the date hereof until the day prior to the hearing on the Bidding Procedures Order, provided that such changes are not material individually or in the aggregate. Any such changes so agreed upon shall be deemed to have been made as of the date of the execution of this Agreement.

#### ARTICLE IX

#### EMPLOYEES AND EMPLOYEE BENEFITS

#### 9.1 Employment.

- (a) <u>Transferred Employees</u>. Purchasers shall offer employment to all individuals employed by Sellers in connection with the Business as of the Closing Date to commence immediately following the Closing, each such offer contingent upon the issuance of the Sale Order of the Bankruptcy Court and the Closing. Each such offer shall be made in accordance with applicable Law and shall be on such terms that, when taken as a whole for each employee, are substantially similar to those currently offered by Sellers. Subject to the terms of any employment agreement entered into by the Purchasers with any of the Key Employees, Purchasers' employment of the Continuing Employees shall be on an "at-will" basis and on such other terms and conditions of employment as Purchasers shall offer in their sole discretion. Except as otherwise agreed to in writing, Purchasers shall be under no obligation to employ or continue to employ any individual for any period. The Employees who accept Purchasers' offer of employment and who commence employment with Purchasers shall be referred to herein as the "<u>Transferred Employees</u>."
- (b) Standard Procedure. Pursuant to the "Standard Procedure" provided in Section 5 of Revenue Procedure 96-60, 1996-2 C.B. 399, (i) Purchasers and Sellers shall report on a predecessor/successor basis as set forth therein, (ii) Sellers will not be relieved from filing a Form W-2 with respect to any Transferred Employees, and (iii) Purchasers will undertake to file (or cause to be filed) a Form W-2 for each such Transferred Employee with respect to the portion of the year during which such

Employees are employed by Purchasers that includes the Closing Date, excluding the portion of such year that such Employee was employed by Sellers.

- (c) With respect to each Transferred Employee, Sellers hereby waive and release each such individual from any and all contractual, common law or other restrictions enforceable by Sellers on the employment, activities or other conduct of such individuals after their termination of employment with Sellers; provided, however, that Sellers shall assign to the Purchasers their rights to all obligations not to disclose confidential information relating to the Business and all obligations not to compete with the Business owed to Sellers by such Transferred Employee.
- (d) Except as expressly provided herein, nothing herein shall be construed as transferring to Purchasers (i) any Contract or agreement with any current or former employee of Sellers or for the employment of any Person or engagement of any independent contractor by Sellers or (ii) any rights or obligations Sellers may owe to or be owed by any current or former employee, officer, director, consultant, independent contractor, or agent of Sellers.
- (e) Nothing herein, express or implied, shall confer upon any employee or former employee of Sellers any rights or remedies (including any right to employment or continued employment for any specified period) of any nature or kind whatsoever, under or by reason of this Agreement. Purchasers and Sellers agree that the provisions contained herein are not intended to be for the benefit of or otherwise be enforceable by, any third party, including any employee or former employee of Sellers.
- (f) Purchasers shall have no obligation to pay severance to any employee who does not accept an offer of employment to join Purchasers and any such obligations remain obligations of the Sellers.

## 9.2 Employee Benefits

- (a) Effective as of the Closing Date, Purchasers shall assume sponsorship of the Employee Benefit Plans agreed upon between the Purchaser and the Sellers set forth on Schedule 9.2(a) and Sellers shall assign any trusts or funding vehicles associated with such plans to Purchasers.
- (b) If Purchasers adopt any new employee benefit plans on behalf of any Transferred Employees following the Closing ("Purchaser Plans"), for purposes of eligibility and vesting (but not benefit accrual) under such Purchaser Plans, Purchasers shall credit each Transferred Employee with his or her years of service with Sellers and any predecessor entities thereof as if such service was performed for the Purchasers. The Purchaser Plans shall not deny Transferred Employees coverage on the basis of pre-existing conditions (except to the extent that a Transferred Employee was denied coverage under an Employee Benefit Plan for a pre-existing condition) and any expenses incurred by or on behalf of such Transferred Employee, on or before the Closing Date, shall be taken into account under the Purchaser Plans for purposes of satisfying

applicable deductible, coinsurance and maximum out-of-pocket provisions under the applicable Purchaser Plans.

- (c) Except as set forth in the next sentence, nothing contained in this Section 9.2 or elsewhere in this Agreement shall be construed to limit Purchasers' ability to terminate the employment of any individual Transferred Employee or amend, modify or terminate any employee benefits available to any individual Transferred Employee, it being understood and agreed that if Purchasers shall terminate the employment of any Transferred Employee within 18 months after the Closing other than for cause, then Purchasers shall provide to each such Transferred Employee (i) outplacement services valued at not less than five thousand dollars (\$5,000) and (ii) all severance benefits no less favorable than those to which such Transferred Employee would be entitled if he or she was terminated at that time by Sellers in accordance with Sellers' severance practice in effect as of the date hereof, a copy of which is attached as Schedule 9.2(c). Purchasers agree not to terminate fifty (50) or more Transferred Employees in either New York and/or Utah within the sixty (60) day period following the Closing.
- (d) Continuation and Portability of Employee Benefits. Purchasers shall be responsible for administering compliance with the continuation coverage requirements for "group health plans" under Title X of the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended ("COBRA"), and the portability requirements under the Health Insurance Portability and Accountability Act of 1996 ("HIPAA") with respect to Transferred Employees whose coverage under the Employee Benefit Plans terminates on or after the Closing Date and any other employee or former employee of the Sellers (and their eligible dependents) who had a "qualifying event" (within the meaning of Section 4980B(f)(d) of the Code on or prior to the Closing (the "COBRA Beneficiaries"). Purchasers shall be responsible for administering compliance under COBRA and HIPAA for the COBRA Beneficiaries and those Transferred Employees whose coverage terminates subsequent to their becoming participants in the Purchaser Plans.
- (e) <u>Key Employees.</u> Sellers shall afford to Purchasers and their representatives reasonable access to those employees whom Purchasers and Sellers agree are key employees (the "<u>Key Employees</u>") in order to permit Purchasers to solicit for employment the Key Employees and to negotiate employment agreements reflecting retention, confidentiality, non-competition and other terms and conditions of employment with the Key Employees.

## ARTICLE X

#### CONDITIONS TO CLOSING

10.1 <u>Conditions Precedent to Obligations of Purchasers</u>. The obligation of Purchasers to consummate the transactions contemplated by this Agreement is subject to the fulfillment, on or prior to the Closing Date, of each of the following conditions (any or all of which may be waived by Purchasers in whole or in part):

- (a) The representations and warranties of Sellers set forth in this Agreement qualified as to materiality shall be true and correct and those not so qualified shall be true and correct in all material respects, at and as of the Closing, except to the extent such representations and warranties expressly relate to an earlier date (in which case such representations and warranties qualified as to materiality shall be true and correct and those not so qualified shall be true and correct in all material respects, on and as of such earlier date); provided, however, that in the event of a breach of a representation or warranty other than a representation or warranty qualified by Material Adverse Effect, the condition set forth in this Section 10.1(a) shall be deemed satisfied unless the effect of all such breaches of representations and warranties taken together result in a Material Adverse Effect, and Purchaser shall have received a certificate signed by an authorized officer of Seller, dated the Closing Date, to the foregoing effect; it being understood, however, that nothing in this proviso shall be deemed to be a waiver by Purchasers of any remedies they may have for a breach of representation or warranty, including under Article XI below; and provided, further, that for purposes of this Section 10.1(a), clause (i) of the definition of Material Adverse Effect shall read "(x) a material adverse effect on the business, assets results of operations or financial condition of the Business or the Purchased Assets, or (y) a material adverse effect on the Specified Assets, the Specified Liabilities and Liabilities relating to or arising out of product liability claims and Environmental Laws, in each case in this clause (y) which have a material adverse effect on the business, assets, results of operation or financial condition of Sellers and their subsidiaries taken as a whole ."
- (b) Any Permits, approvals or consents required for Sellers in connection with consummation of the transactions contemplated hereby and by the Ancillary Agreements shall have been obtained and shall be in full force and effect on the Closing Date:
- (c) The Bankruptcy Court shall have entered the Bidding Procedures Order by the later of September 29, 2003 or 10 days after the appointment of an official committee of unsecured creditors, in the form and substance attached hereto as Exhibit A, or which shall otherwise have been approved of by the Purchasers in their sole and absolute discretion, and the Bidding Procedures Order, as entered by the Bankruptcy Court, shall not be stayed and shall not modify the terms and conditions of this Agreement or the Ancillary Agreements or the transactions contemplated hereby or thereby in any way that would adversely affect Purchasers;
- (d) Sellers shall have performed and complied in all material respects with all obligations and agreements required in this Agreement to be performed or complied with by it prior to the Closing Date, and Purchasers shall have received a certificate signed by an authorized officer of Sellers, dated the Closing Date, to the forgoing effect; and
- (e) Seller shall have delivered, or caused to be delivered, to Purchasers all of the items set forth in <u>Section 4.2.</u>

- (f) Since the Balance Sheet Date, there shall not have been a Material Adverse Effect;
- (g) Sellers shall have obtained the third party consents, if any, necessary for the assignment and assumption of the Purchased Contracts, except for such consents, the failure of which to obtain would not be material;
- (h) Purchaser shall have obtained, from a title company selected by Purchasers and at the then prevailing rates, a title insurance policy or irrevocable commitment for a title insurance policy for all of the Owned Property in form and substance reasonably satisfactory to Purchasers, insuring Purchasers as the owner of the Owned Property and containing no exceptions to title other than the Permitted Exceptions; and
- (i) The Real Property, together with all facilities, buildings, improvements, and fixtures, is in the same physical condition it was in on the date this Agreement was executed, subject to ordinary wear and tear. No condemnation of all or a portion of the Real Property shall have been threatened or shall have occurred between the date of this Agreement and Closing.
- 10.2 <u>Conditions Precedent to Obligations of Seller</u>. The obligations of Sellers to consummate the transactions contemplated by this Agreement are subject to the fulfillment, prior to or on the Closing Date, of each of the following conditions (any or all of which may be waived by Sellers in whole or in part):
- (a) The representations and warranties of Purchasers set forth in this Agreement qualified as to materially shall be true and correct, and those not so qualified shall be true and correct in all material respects, at and as of the Closing, except to the extent such representations and warranties expressly relate to an earlier date (in which case such representations and warranties qualified as to materially shall be true and correct, and those not so qualified shall be true and correct in all material respects, on and as of such earlier date), and Sellers shall have received a certificate signed by an authorized officer of Purchasers, dated the Closing Date, to the forgoing effect;
- (b) Any Permits, approvals or consents required for Purchasers in connection with consummation of the transactions contemplated hereby and by the Ancillary Agreements shall have been obtained and shall be in full force and effect on the Closing Date;
- (c) Purchasers shall have performed and complied in all material respects with all obligations and agreements required by this Agreement to be performed or complied with by Purchasers on or prior to the Closing Date, and Sellers shall have received a certificate signed by an authorized officer of Purchasers, dated the Closing Date, to the forgoing effect; and
- (d) Purchasers shall have delivered, or caused to be delivered, to Sellers all of the items set forth in Section 4.3.

- 10.3 <u>Conditions Precedent to Obligations of Purchasers and Sellers</u>. The respective obligations of Purchasers and Sellers to consummate the transactions contemplated by this Agreement are subject to the fulfillment, on or prior to the Closing Date, of each of the following conditions (any or all of which may be waived by Purchasers and Sellers in whole or in part):
- (a) no Governmental Body shall have enacted, issued, promulgated, enforced or entered any Law or non-appealable Order that is in effect on the Closing Date and prohibits the consummation of the Closing or imposes material limitations on Purchasers' ownership of the Purchased Assets;
- (b) there shall not be in effect any Order by a Governmental Body of competent jurisdiction restraining, enjoining or otherwise prohibiting the consummation of the transactions contemplated hereby;
- (c) the Bankruptcy Court shall have entered the Sale Order and such order shall be final and non-appealable; and
- (d) if any filing under the HSR Act is required, the waiting period applicable to the transactions contemplated by this Agreement under the HSR Act shall have expired or early termination shall have been granted.

#### ARTICLE XI

#### SURVIVAL AND INDEMNIFICATION

11.1 Survival of Representations and Warranties. The parties hereto agree that (i) the Purchasers' representations and warranties contained in this Agreement shall survive the Closing and expire on the first anniversary of the Closing Date and (ii) the Sellers' representations and warranties contained in this Agreement shall survive the Closing, and expire on the first anniversary of the Closing Date. The parties hereto agree that the covenants contained in this Agreement to be performed at or after the Closing shall survive the Closing hereunder, and each party hereto shall be liable to the other after the Closing for any breach thereof.

## 11.2 Indemnification.

- (a) Sellers agree, jointly and severally, to indemnify and hold the Purchasers and their respective directors, officers, employees, Affiliates, agents, successors and permitted assigns (collectively, the "Purchaser Indemnified Parties") harmless from and against any and all losses, Liabilities, obligations, damages, costs and expenses (individually, a "Loss" and, collectively, "Losses") based upon, attributable to or resulting from:
  - (i) any and all Excluded Liabilities, whether or not such Liabilities reflect a breach of a representation or warranty of Sellers in this Agreement;

- (ii) any breach by Sellers of their representations, warranties or covenants contained herein which survive the Closing; and
- (iii) all actions, suits, proceedings, claims, demands, assessments, judgments, costs and expenses, including reasonable attorneys' fees, incident to the foregoing.
- (b) Purchasers agree, jointly and severally, to indemnify and hold Sellers and their directors, officers, employees, Affiliates, agents, successors and permitted assigns ("Seller Indemnified Parties") harmless from and against any Losses based upon, attributable to or resulting from:
  - (i) any and all Assumed Liabilities;
  - (ii) any breach by Purchasers of their representations, warranties or covenants contained herein which survive the Closing;
  - (iii) Liabilities with respect to the Business, the Purchased Assets or the Transferred Employees to the extent such Liabilities arise out of the operations of the Business after the Closing, in each case, other than the Excluded Liabilities; and
  - (iv) all actions, suits, proceedings, claims, demands, assessments, judgments, costs and expenses, including reasonable attorneys' fees, incident to the foregoing.
- (c) In the event that any Legal Proceedings shall be instituted or that any claim or demand shall be asserted by any Person in respect of which indemnification may be sought from Purchaser Indemnified Parties or Seller Indemnified Parties, as the case may be, pursuant to the provisions of this Section 11.2, the party seeking indemnification shall, to the extent of its knowledge thereof, cause prompt written notice of the commencement of such proceedings or the assertion of such claim or demand to be given to the indemnifying party, and shall afford to such indemnifying party the right, to the extent of its indemnification, at its option and at its own expense, to be represented by counsel of its choice and to defend against, negotiate, settle, or otherwise deal with any such legal proceeding, claim or demand; provided, however, that the failure by any party seeking indemnification hereunder to give prompt notice shall not release the indemnifying party of its indemnification obligations hereunder, except to the extent such failure actually prejudices such indemnifying party; provided further, however, that no indemnifying party shall agree to any settlement of any claim against any Person seeking indemnification without the consent of such Person (which shall not be unreasonably withheld) unless such settlement requires no payment on the part of such Person and contains a full release of such Person, with no admission of any wrongdoing on the part of such Person; and provided further, however, that if the liability or obligation which is the subject matter of such claim shall arise out of a transaction or cover any period or periods wherein Sellers and Purchasers shall be responsible for part of any such liability

or obligation, then Sellers and Purchasers jointly shall defend, contest, litigate, settle and otherwise deal with any such claims, each bearing its own expenses and each choosing its own counsel. After any final judgment or award shall have been rendered by a court, arbitration board, or administrative agency of competent jurisdiction, or a settlement shall have been consummated, or the parties shall have arrived at a mutually binding agreement, with respect to any matter which is the subject matter of an indemnity hereunder, the indemnified party shall forward to the indemnifying party notice of any sums due and owing by the indemnifying party with respect to such matter and the indemnifying party shall be required to pay all of the sums so owing to the indemnified party, by certified or bank cashier's check, within ten (10) Business Days after the date of such notice. The parties hereto agree to cooperate fully with each other in connection with the defense, negotiation or settlement of any such legal proceeding, claim or demand, and none of Purchasers or Sellers will compromise or settle any such legal proceeding, claim or demand without the prior written consent of the other, not to be unreasonably withheld or delayed.

- (d) Notwithstanding anything in this Agreement to the contrary, the indemnification provisions with respect to the representation on Current Assets set forth in Section 5.26 shall be subject to the additional provisions set forth on Schedule 11.2(d).
- Closing Date, Purchasers shall deposit in an account (the "Indemnity Account"), in immediately available funds, a portion of the Purchase Price in an amount equal to three million dollars (\$3,000,000) (the "Indemnity Escrow Amount"). Any payment that Sellers are obligated to make to any Purchaser Indemnified Parties pursuant to this Article XI, including under Section 11.2(d), shall be paid solely from the Indemnity Escrow Amount and following the Closing Sellers shall have no liability to Purchasers for breaches of, defaults under, or non-performance of this Agreement above amounts held in such Indemnity Escrow Account, except for fraud or willful breach of this Agreement. The Escrow Agreement shall provide that on the first anniversary of the Closing Date, so long as no claims for indemnification are outstanding, Sellers shall withdraw from the Indemnity Account the Indemnity Escrow Amount (to the extent not utilized to pay Purchasers for any indemnification claim).

## ARTICLE XII

#### TAX MATTERS

12.1 <u>Transfer Taxes</u>. Sellers and Purchaser shall share equally any sales, use, stamp, documentary stamp, filing, recording, transfer or similar fees or taxes or governmental charges (including any interest and penalty thereon) payable in connection with the transactions contemplated by this Agreement ("<u>Transfer Taxes</u>"). Seller shall seek to include in the Sales Order a provision that provides that the transfer of the Purchased Assets shall be free and clear of any stamp or similar taxes under Bankruptcy Code Section 1146(c). Sellers and Purchasers shall cooperate and consult with each other prior to filing any Tax Returns in respect of Transfer Taxes. Sellers and Purchasers shall

cooperate and otherwise take commercially reasonable efforts to obtain any available refunds for Transfer Taxes.

- 12.2 Cooperation. Purchasers and Sellers agree to furnish or cause to be furnished to each other, as promptly as practicable, such information and assistance relating to the Purchased Assets as is reasonably necessary for the preparation and filing of any Tax Return, for the preparation for and proof of facts during any tax audit, for the preparation for any tax protest, for the prosecution or defense of any suit or other proceeding relating to tax matters and for the answer of any governmental or regulatory inquiry relating to tax matters. From and after the Closing Date, Purchasers agree that they will provide reasonable access to Sellers and their attorneys, accountants and other representatives (after reasonable notice and during normal business hours and without charge) to such files, books and records as Sellers may reasonably deem necessary to prepare for, file, prove, answer, prosecute or defend any claim, suit, inquiry or other proceeding, related to Taxes in connection with the Purchased Assets or the Business.
- obligations levied with respect to the Purchased Assets for any taxable period that includes the Closing Date and ends after the Closing Date, whether imposed or assessed before or after the Closing Date, shall be prorated between Sellers and Purchasers as of 12:01 a.m. (Eastern time) on the Closing Date. If any Taxes subject to proration are paid by Purchasers, on the one hand, or Sellers, on the other hand, the proportionate amount of such Taxes paid (or in the event a refund of any portion of such Taxes previously paid is received, such refund) shall be paid promptly by (or to) the other after the payment of such Taxes (or promptly following the receipt of any such refund).
- Real Estate Tax Protests. Prior to the Closing, all real estate assessment protests and proceedings, if any, affecting the Real Property or any portion thereof for the tax year in which the Closing occurs (the "Closing Date Tax Year") and prior years will be prosecuted under Sellers' direction and control, provided that Sellers shall not settle, without Purchasers' prior written consent, not to be unreasonably withheld or delayed, any such protest or proceeding affecting the Closing Date Tax Year or any subsequent Tax Year. From and after the Closing, all real estate assessment protests and proceedings for the Closing Date Tax Year and all tax years thereafter occurring shall be prosecuted under the Purchasers' direction and control and Sellers shall do everything reasonably necessary to transfer over to the Purchasers the right to prosecute and settle such protests and proceedings for the Closing Date Tax Year. In the event of any reduction in the assessed valuation of the Real Property for the Closing Date Tax Year or any prior tax year, the amount of any tax savings shall (a) with respect to tax years ending prior to the Closing, be payable to Sellers; and (b) with respect to the Closing Date Tax Year, after deduction of customary expenses and attorneys fees incurred in connection with obtaining same, be adjusted between Sellers and Purchaser as of the Closing.

# ARTICLE XIII

#### **MISCELLANEOUS**

- 13.1 Expenses. Except as otherwise provided in this Agreement, Sellers and Purchasers shall bear their own expenses incurred in connection with the negotiation and execution of this Agreement and each other agreement, document and instrument contemplated by this Agreement and the consummation of the transactions contemplated hereby and thereby. Purchasers and Sellers shall bear their respective filing fees associated with HSR Act filings, if any.
- 13.2 Injunctive Relief. Damages at law may be an inadequate remedy for the breach of any of the covenants, promises and agreements contained in this Agreement, and, accordingly, any party hereto shall be entitled to injunctive relief with respect to any such breach, including specific performance of such covenants, promises or agreements or an order enjoining a party from any threatened, or from the continuation of any actual, breach of the covenants, promises or agreements contained in this Agreement. The rights set forth in this Section 13.2 shall be in addition to any other rights which a party may have at law or in equity pursuant to this Agreement.

# 13.3 Submission to Jurisdiction; Consent to Service of Process.

- (a) Without limiting any party's right to appeal any order of the Bankruptcy Court, (i) the Bankruptcy Court shall retain exclusive jurisdiction to enforce the terms of this Agreement and to decide any claims or disputes which may arise or result from, or be connected with, this Agreement, any breach or default hereunder, or the transactions contemplated hereby, and (ii) any and all proceedings related to the foregoing shall be filed and maintained only in the Bankruptcy Court, and the parties hereby consent to and submit to the jurisdiction and venue of the Bankruptcy Court and shall receive notices at such locations as indicated in Section 13.7 hereof; provided, however, that if the Bankruptcy Case has closed, the parties agree to unconditionally and irrevocably submit to the exclusive jurisdiction of the United States District Court for the Southern District of New York sitting in New York County or the Commercial Division, Civil Branch of the Supreme Court of the State of New York sitting in New York County and any appellate court from any thereof, for the resolution of any such claim or dispute. The parties hereby irrevocably waive, to the fullest extent permitted by applicable law, any objection which they may now or hereafter have to the laying of venue of any such dispute brought in such court or any defense of inconvenient forum for the maintenance of such dispute. Each of the parties hereto agrees that a judgment in any such dispute may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by Law.
- (b) Each of the parties hereto hereby consents to process being served by any party to this Agreement in any suit, action or proceeding by delivery of a copy thereof in accordance with the provisions of <u>Section 13.7</u>.

- 13.4 <u>Waiver of Right to Trial by Jury</u>. Each party to this Agreement waives any right to trial by jury in any action, matter or proceeding regarding this Agreement or any provision hereof.
- Entire Agreement; Amendments and Waivers. This Agreement (including the schedules and exhibits hereto), together with the Ancillary Documents, represents the entire understanding and agreement between the parties hereto with respect to the subject matter hereof. This Agreement can be amended, supplemented or changed, and any provision hereof can be waived, only by written instrument making specific reference to this Agreement signed by the party against whom enforcement of any such amendment, supplement, modification or waiver is sought. No action taken pursuant to this Agreement, including any investigation by or on behalf of any party, shall be deemed to constitute a waiver by the party taking such action of compliance with any representation, warranty, covenant or agreement contained herein. The waiver by any party hereto of a breach of any provision of this Agreement shall not operate or be construed as a further or continuing waiver of such breach or as a waiver of any other or subsequent breach. No failure on the part of any party to exercise, and no delay in exercising, any right, power or remedy hereunder shall operate as a waiver thereof, nor shall any single or partial exercise of such right, power or remedy by such party preclude any other or further exercise thereof or the exercise of any other right, power or remedy. All remedies hereunder are cumulative and are not exclusive of any other remedies provided by law.
- 13.6 Governing Law. This Agreement shall be governed by and construed in accordance with the laws of the State of New York applicable to contracts made and performed in such State.
- be in writing and shall be deemed given (i) when delivered personally by hand (with written confirmation of receipt), (ii) when sent by facsimile (with written confirmation of transmission) or (iii) one business day following the day sent by overnight courier (with written confirmation of receipt), in each case at the following addresses and facsimile numbers (or to such other address or facsimile number as a party may have specified by notice given to the other party pursuant to this provision):

If to Sellers, to:

Twin Laboratories Inc. 150 Motor Parkway Suite 210 Hauppauge, New York 11788 Facsimile: (631) 630-3474 Attention: Ross Blechman

With a copy (which shall not constitute notice) to:

Weil, Gotshal & Manges LLP 767 Fifth Avenue New York, NY 10153 Facsimile: (212) 310-8007

Attention: Simeon Gold and Michael Kessler

If to Purchasers, to:

IdeaSphere Inc. 3133 Orchard Vista Drive, SE Grand Rapids, MI 49546 Facsimile: (616) 575-0556 Attention: Mark Fox

With a copy to:

Kramer Levin Naftalis & Frankel LLP 919 Third Avenue New York, NY 10022 Facsimile: (212) 715-8000 Attention: Kenneth H. Eckstein

- 13.8 Severability. If any term or other provision of this Agreement is invalid, illegal, or incapable of being enforced by any law or public policy, all other terms or provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to any party. Upon such determination that any term or other provision is invalid, illegal, or incapable of being enforced, the parties hereto shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in an acceptable manner in order that the transactions contemplated hereby are consummated as originally contemplated to the greatest extent possible.
- 13.9 <u>Binding Effect; Assignment.</u> This Agreement shall be binding upon and inure to the benefit of the parties and their respective successors and permitted assigns. Nothing in this Agreement shall create or be deemed to create any third party beneficiary rights in any person or entity not a party to this Agreement except as provided below. No assignment of this Agreement or of any rights or obligations hereunder may be made by any Seller, on the one hand, or any Purchaser, on the other hand, (by operation of law or otherwise) without the prior written consent of the other parties hereto and any attempted assignment without the required consents shall be void except that any Purchaser may (i) assign any or all of its rights and interests hereunder and obligations hereunder to one or more Affiliates and (ii) designate one or more Affiliates to perform its obligations hereunder (in any or all of which cases, the applicable Purchaser nonetheless shall remain responsible for the performance of all its obligations hereunder). No assignment of any obligations hereunder shall relieve the parties hereto of any such obligations. Upon any

such permitted assignment, the references in this Agreement to Purchaser shall also apply to any such assignee unless the context otherwise requires.

- 13.10 Non-Recourse. No past, present or future director, officer, employee, incorporator, member, partner, stockholder, Affiliate, agent, attorney or representative, of any Seller shall have any liability for any obligations or liabilities of any Seller under this Agreement or the Ancillary Agreements of or for any claim based on, in respect of, or by reason of the transactions contemplated hereby and thereby. No past, present or future director, officer, employee, incorporator, member, partner, stockholder, Affiliate, agent, attorney or representative of Purchasers shall have any liability for any obligations or liabilities of Purchasers under this Agreement or the Ancillary Agreements or for any claim based on, in respect of, or by reason of, the transactions contemplated hereby and thereby.
- 13.11 Counterparts. This Agreement may be executed in one or more counterparts, each of which will be deemed to be an original copy of this Agreement and all of which, when taken together, will be deemed to constitute one and the same agreement.

[Signature Page Follows]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed by their respective officers thereunto duly authorized, as of the date first written above.

Sellers:
TWINLAB CORPORATION
By: Name: Title:
TWIN LABORATORIES INC.
By: Name: Title:
TWIN LABORATORIES (UK) LTD.
By:Name: Title:
Purchasers:
IDEASPHERE, INC.
Ву:
Name: Title:

By:	TL ACQUISITION CORP.
Name: Title:	Name:

## Schedule 11.2(d)

- (a) Not later than sixty (60) days after the Closing Date, Purchasers shall prepare and deliver to Sellers a statement (the "Final Closing Statement"), which shall include a worksheet setting forth a reasonably detailed calculation of (i) consolidated Current Assets of Sellers (the "Final Current Assets") as of the Closing Date, prepared in accordance with GAAP in a manner consistent with past practice and the books and records of each Seller as of the Closing Date.
- (b) (i) If the Final Current Assets, net of any Inventory and accounts receivable relating to Excluded Products, exceeds forty seven million dollars (\$47,000,000) by more than two hundred fifty thousand dollars (\$250,000) (the amount of such excess over two hundred fifty thousand dollars (\$250,000) being referred to as the "Excess"), Purchasers shall pay to Sellers the Excess.
- (ii) If the Final Current Assets, net of any Inventory and accounts receivable relating to Excluded Products, is less than forty seven million dollars (\$47,000,000) by more than two hundred fifty thousand dollars (\$250,000) (the amount of such difference over two hundred fifty thousand dollars (\$250,000) being referred to as the "Deficit"), Seller shall pay to Purchasers the Deficit provided that the Deficit shall be paid solely from the Indemnity Account.
- (iii) Any amount payable by Purchasers or Sellers pursuant to clause (i) or (ii) of this paragraph (b) is referred to as the "True-Up Amount."
- (iv) Any True-Up Amount shall be paid in cash within ten (10) days following Sellers' receipt of the Final Closing Statement by wire transfer of immediately available funds to an account designated by the receiving party, plus interest thereon from and including the Closing Date through and including the day before the date of such payment, at a per annum rate equal to the prime rate (the "Prime Rate") as reported in the Wall Street Journal (under the heading "Money Rates") on the Closing Date.
- thirty (30) days to review the Final Closing Statement. Upon the expiration of such 30-day period, Sellers shall be deemed to have accepted, and shall be bound by, the Final Closing Statement and the calculation therein of the Final Current Assets, unless Sellers have informed the other parties in writing of their disagreement with the Final Closing Statement prior to the expiration of such thirty (30) day period (the "Objection"), specifying each of the disputed items and setting forth in reasonable detail the basis for each such dispute (each, a "Disputed Item"). Purchasers shall have thirty (30) days from the date on which it receives the Objection (the "Objection Date") to review and respond to such Objection. If Sellers and Purchasers are able to negotiate a mutually agreeable resolution of each Disputed Item, and each signs a certificate to that effect, the Final Closing Statement and the calculation therein of the True-Up Amount, as adjusted to reflect such resolution, shall be deemed final, non-appealable and binding for purposes of this Agreement.

- (d) If within thirty (30) days of the Objection Date, Purchasers and Sellers are unable to resolve all of their disagreements with respect to the Disputed Items, Purchasers and Sellers shall refer the Disputed Items to a mutually acceptable "Big Four" accounting firm (the "Accounting Firm") to make a final, non-appealable and binding determination as to such Disputed Items. The Accounting Firm shall be directed to make a determination of the Disputed Items in accordance with GAAP promptly, but no later than thirty (30) days, after acceptance of its appointment. Purchasers and Sellers shall make readily available to the Accounting Firm all relevant books, records and employees of the Business reasonably requested by the Accounting Firm in connection with the Accounting Firm's review of any Disputed Item.
- Purchasers and Sellers in the same proportion that the dollar amount of Disputed Items which are not resolved in favor of Purchasers and Sellers (as applicable) bears to the total dollar amount of Disputed Items resolved by the Accounting Firm. For illustration purposes only, (A) if the total amount of Disputed Items by Sellers is \$1,000,000, and Sellers are awarded \$500,000 by the Accounting Firm, Purchasers and Sellers shall bear the Accounting Firm's fees and expenses equally; or (B) if the total amount of Disputed Items by Sellers is \$1,000,000, and Sellers are awarded \$250,000 by the Accounting Firm, Sellers shall bear 75 percent (75%) and Purchasers shall bear 25 percent (25%) of the Accounting Firm's fees and expenses. Each of Purchasers and Sellers shall bear the fees, costs and expenses of its own accountants and all of its other expenses incurred in connection with matters contemplated by this Schedule 11.2(d).
- (f) In the event an Objection is submitted, payment of any True-Up Amount, shall be made within ten (10) days following the final resolution of all Disputed Items by the parties or the Accounting Firm by wire transfer of immediately available funds to an account designated by Purchasers or Sellers, as the case may be, plus interest thereon through and including the day before the date of such payment, at a per annum rate equal to the Prime Rate as in effect on the Closing Date.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed by their respective officers thereunto duly authorized, as of the date first written above.

Sellers:	
TWINLAB CORPORATION	
By: Xoro Blechmen Name: Ross Burchmen	
Name:\ Ross Blechman\ Title: Res.	-
TWIN LABORATORIES INC.	ž.
By: Kon Bleckmen Namer Ross Blechmen	
Name Ross Oceann Title: Pees	
TWIN LABORATORIES (UK) LTD.	
By: Name: Ross Blechman	
Name: Ross Blechman Title: Pees.	•
Purchasers:	
IDEASPHERE, INC.	
By:	
Name: Title:	

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed by their respective officers thereunto duly authorized, as of the date first written above.

Sellers:	
TWINLAB CORPORATION	
By:	<u> </u>
Title:	
TWIN LABORATORIES INC.	•
Ву	. '
Name: Title:	•
TWIN LABORATORIES (UK) LTD	•
Ву:	<u> </u>
Name; Title:	•
Purchasers;	
IDEASPHERE, INC	4
By. Aluca Class	

Title:

By. Name:

Title:

# AMENDMENT NO. 1 TO ASSET PURCHASE AGREEMENT

TWINLAB LEGAL

This Amendment No. 1 (this "Amendment"), dated as of December 5, 2003, amends the Asset Purchase Agreement, dated as of September 4, 2003 (the "Purchase Agreement"), among Twinlab Corporation, a Delaware corporation, Twin Laboratories Inc., a Utah corporation, and Twin Laboratories (UK) Ltd., a company organized under the laws of the United Kingdom (each, a "Seller," and collectively, "Sellers"), and IdeaSphere, Inc., a Michigan corporation ("IdeaSphere"), and TL Acquisition Corp., a Delaware corporation ("Acquisition Sub," each of IdeaSphere and Acquisition Sub being individually referred to as a "Purchaser", and together as "Purchasers"), as previously modified by that certain letter agreement from the Purchasers to the Sellers, dated as of November 14, 2003.

WHEREAS, the parties hereto (each, a "Party" and collectively, the "Parties") are all of the existing parties to the Purchase Agreement; capitalized terms used herein and not otherwise defined shall have the meanings given to such terms in the Purchase Agreement;

WHEREAS, pursuant to Section 13.5 of the Purchase Agreement, the Parties may amend, supplement or change the Purchase Agreement at any time prior to the Closing Date in accordance with the provisions of such Section;

WHEREAS, the Sellers, in Section 5.26 of the Purchase Agreement, represented that as of the Closing, the Current Assets of the Sellers, net of any Inventory and accounts receivable relating to Excluded Products, will be no less than forty seven million dollars (\$47,000,000), calculated in accordance with GAAP; and

WHEREAS, the Current Assets of the Sellers, net of any Inventory and accounts receivable relating to Excluded Products, is expected to be significantly below the agreed upon aniount:

NOW, THEREFORE, in consideration of the foregoing and the mutual agreements set forth herein, the Parties hereto agree as follows:

The following reference shall be added before the reference for the definition of "Alternative Transaction" in Section 1.2 of the Purchase Agreement:

Term	Section
11.2(d) Amount	11.3

The following reference shall be added after the reference for the definition of "Financial Statements" in Section 1.2 of the Purchase Agreement:

<u>Tean</u>	<u>Section</u>
General Indemnity Claim Amount	11.3

Section 3.1 of the Purchase Agreement is hereby deleted in full and replaced with the following:

RELIGIOS NAMES A

- 3.1 Consideration. The aggregate cash consideration for the Purchased Assets shall be an amount in cash equal to fifty seven million five hundred thousand dollars (\$57,500,000) less the Zions Mortgage Balance if the Zions Mortgage is assumed by Purchasers (the "Cash Purchase Price"), plus the Employee and Related Liabilities in an amount up to three million seven hundred thousand dollars (\$3,700,000) (the "Purchase Price"). The aggregate consideration for the Purchased Assets shall also include the assumption of the other Assumed Liabilities.
- 4. The second sentence of <u>Section 5.26</u> of the Purchase Agreement is hereby deleted in full and replaced with the following:

As of the Closing, the Sellers' Current Assets plus pre-paid amounts for inventory not yet received and advertising, each as set forth on <u>Schedule 5.26</u>, net of any inventory and accounts receivable relating to Excluded Products, will be no less than thirty pine million five hundred thousand dollars (\$39,500,000), calculated in accordance with GAAP.

- 5. Clause (b)(i) of Schedule 11.2(d) of the Purchase Agreement is hereby deleted in full and replaced with the following:
  - (i) If the Final Current Assets plus pre-paid amounts for Inventory not yet received and advertising, each as set forth on Schedule 5.26, net of any Inventory and accounts receivable relating to Excluded Products, exceeds thirty nine million five hundred thousand dollars (\$39,500,000) (the amount of such excess being referred to as the "Excess"), Purchasers shall pay to Sellers the Excess, up to an aggregate limit of one million five hundred thousand dollars (\$1,500,000).
- 6. Clause (b)(ii) of Schedule 11.2(d) of the Purchase Agreement is hereby deleted in full and replaced with the following:
  - (ii) If the Final Current Assets plus pre-paid amounts for Inventory not yet received and advertising, each as set forth on Schedule 5.26, net of any Inventory and accounts receivable relating to Excluded Products, is less than thirty nine million five hundred thousand dollars (\$39,500,000) (the amount of such difference being referred to as the "Deficit"), Seller shall pay to Purchasers the Deficit, provided that the Deficit shall be paid solely from the 11.2(d) Amount.
- 7. Section 11.3 of the Purchase Agreement is hereby deleted in full and replaced with the following:

KL) - \$19802 6

- Indemnity Escrow. Pursuant to the terms of Section 3.3 11.3 hereof, on the Closing Date, Purchasers shall deposit in an account (the "Indemnity Account"), in immediately available funds, a portion of the Purchase Price in an amount equal to one million five hundred thousand dollars (\$1,500,000) (the "Indemnity Escrow Amount"), of which five hundred thousand dollars (\$500,000) shall constitute the "11.2(d) Amount" and the remaining one million dollars (\$1,000,000) shall constitute the "General Indemnity Claim Amount"). Any payment that Sellers are obligated to make to any Purchaser Indemnified Parties following the procedures in Schedule 11.2(d) hereto shall be paid solely from the 11.2(d) Amount. Any payment that Sellers are obligated to make to any Purchaser Indemnified Parties pursuant to this Article XI, other than payments which are to be paid solely from the 11.2(d) Amount pursuant to the preceding sentence, shall be paid solely from the General Indomnity Claim Amount. Following the Closing, Sellers shall have no liability to Purchasers for breaches of, defaults under, or non-performance of this Agreement above the amounts held in the Indemnity Escrow Account, except for fraud or willful breach of this Agreement. The Escrow Agreement shall provide that on the first Business Day following the final determination and payment of the True-Up Amount pursuant to Schedule 11.2(d), Setlers shall withdraw from the Indemnity Account the 11.2(d) Amount (to the extent not utilized to pay Purchasers for any applicable True Up Amount). The Escrow Agreement shall further provide that on the first auniversary of the Closing Date, so long as no claims for indemnification, other than claims for indemnification for a breach of any representation on Current Assets set forth in Section 5.26 hereof, are outstanding, Sellers shall withdraw from the Indemnity Account the General Indemnity Claim Amount (to the extent not utilized to pay Purchasers for any such indemnification claim).
- 8. Except as expressly set forth herein, this Amendment shall not, by implication or otherwise, limit, impair, constitute a waiver of, or otherwise affect the rights and remedies of any Party under the Asset Purchase Agreement, and shall not alter, modify, amend or in any way affect any of the terms, conditions, obligations, covenants or agreements contained in the Asset Purchase Agreement, all of which are ratified and affirmed in all respects and shall continue in full force and effect.
- This Amendment shall be governed by and construed in accordance with the laws of the State of New York applicable to contracts made and performed in such State.
- 10. This Amendment may be executed in one or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.

KI 2 (2598)Ž I

10/21/04 12:49 FAX 5186303485

→ VARNUM

To the extent that any provision of this Amendment conflicts or is inconsistent 11. with the terms of the Purchase Agreement, this Amendment shall govern, it being agreed that in such event any provisions of the Purchase Agreement required to be amended to give effect to this Amendment shall be deemed amended.

[REMAINDER OF THIS PAGE INTENTIONALLY LEFT BLANK]

KI2 149400.4

IN WITNESS WHEREOF, the Sellers have caused this Bill of Sale to be duly executed and delivered as of the 5 day of December, 2003.

TWINLAB LEGAL

TWINLAB CORPORATION

By: \_

Name: Title:

TWIN LABORATORIES INC.

By: \_

Name: Title:

TWIN LABORATORIES (UK) LTD.

Ву:

Name

(1,3:2230201.4

**PURCHASERS** 

IDEASPHERE, INC.

Title: CF01CL0

TL ACQUISITION CORP.

Title: cfo/cho

AL 2-4024

# AMENDMENT NO. 2 TO ASSET PURCHASE AGREEMENT

This Amendment No. 2 (this "Amendment"), dated as of December 19, 2003, amends the Asset Purchase Agreement, dated as of September 4, 2003 (the "Purchase Agreement"), among Twinlab Corporation, a Delaware corporation, Twin Laboratories Inc., a Utah corporation, and Twin Laboratories (UK) Ltd., a company organized under the laws of the United Kingdom (each, a "Seller," and collectively, "Sellers"), and Idea Sphere, Inc., a Michigan corporation ("IdeaSphere"), and TL Acquisition Corp., a Delaware corporation ("Acquisition Sub," each of ("IdeaSphere and Acquisition Sub being individually referred to as a "Purchaser", and together as "Purchasers"), as previously modified by (i) that certain letter agreement from the Purchasers to "Purchasers"), as of November 14, 2003 and (ii) that certain Amendment No. 1 to the Purchase Agreement, dated as of December 5, 2003.

WHEREAS, the parties hereto (each, a "Party" and collectively, the "Parties") are all of the existing parties to the Purchase Agreement; capitalized terms used herein and not otherwise defined shall have the meanings given to such terms in the Purchase Agreement;

WHEREAS, pursuant to <u>Section 13.5</u> of the Purchase Agreement, the Parties may amend, supplement or change the Purchase Agreement at any time prior to the Closing Date in accordance with the provisions of such Section;

WHEREAS, the Parties agreed to use their best efforts to close on December 9, 2003 or as possible thereafter, but in no event later than December 19, 2003 and to amend the Purchase Agreement to provide for a purchase price adjustment as described herein;

WHEREAS, CIT has agreed to extend the terms of the DTP Financing until December 19, 2003, in exchange for receiving the Extension Payment (as defined herein);

WHEREAS, the Parties have agreed to such other amendments to the Purchase Agreement as are set forth herein;

NOW, THEREFORE, in consideration of the foregoing and the mutual agreements set forth herein, the Farties hereto agree as follows:

1. Section 1.1 of the Purchase Agreement is hereby amended to add the following new definitions:

"Accounts Payable Arising in the Ordinary Course of Business" means all accounts payable arising in the ordinary course of business of the Sellers with respect to the Purchased Assets and Assumed Liabilities, it being understood that none of (i) restructuring costs; (ii) legal and professional consultant fees; (iii) any liabilities falling within the categories set forth on Schedule 2.3 (a) of the Purchase Agreement; or (iv) liabilities related to Excluded Assets, Excluded Products or Excluded Liabilities shall be included in the term "Accounts Payable Arising in the Ordinary Course of Business."

Ka.: 3341914 9

"Administrative Claim" means a claim asserted or arising under Section 503 (a) (1) of the Bankruptcy Code that is entitled to priority under Section 507 (a) (1) of the Bankruptcy Code.

"CIT" means The CIT Group/Business Credit, Inc.

"Closing Date Current Liabilities" means, with respect to the Sellers, the sum of (i) the principal outstanding under the DIP Financing as of the Closing Date, (1) without giving effect to (A) the repayment of the indebtedness under the DIP Financing out of the proceeds of the Purchase Price, (B) any payment during the period beginning on December 10, 2003 and ending on the Closing Date (inclusive) of accounts payable that are not Accounts Payable Arising in the Ordinary Course of Business and (C) the amount by which the indebtedness under the DIP Financing increases due to the Extension Payment but (2) assuming that all accounts receivable relating to the Purchased Assets collected during the period beginning on December 10, 2003 and ending on the Closing Date that were not applied to reduce the indebtedness under the DIF Financing were so applied, plus (ii) any Administrative Claims relating to Accounts Payable Arising in the Ordinary Course of Business outstanding as of the Closing Date (as determined in accordance with GAAP), calculated without giving effect to the three million dollars (\$3,000,000) reduction in accounts payable owed by Sellers to Purchasers in accordance with Section 3.1(y).

"December 9 Current Liabilities" means, with respect to the Sellers the sum of (i) the principal outstanding under the DIP Financing as of December 9, 2003 and (ii) any Administrative Claims relating to Accounts Payable Arising in the Ordinary Course of Business outstanding as of December 9, 2003 (as determined in accordance with GAAP).

"DIP Financing" means that certain Amended and Restated Financing Agreement, dated as of September 4, 2003, among CIT, as the agent and lender, the other financial institutions party thereto, as lenders, and Twin Laboratories Inc., as the Borrower.

"Extension Payment" means that certain cash payment of one hundred thousand dollars (\$100,000) to be made to CIT upon agreement by CIT to extend the maturity date of the DIP Financing to December 19, 2003.

"Inventory Financing" means the financing provided by the Purchasers and their Affiliates to Sellers for the purchase of

Inventory, pre-paid Inventory and pre-paid advertising, in each case which is included in Final Current Assets.

2. (a) The definition of "Excluded Products" in Section 1.1 of the Purchase Agreement is deleted in full and replaced with the following:

"Excluded Products" means any and all products that include (i) alkaloids from the herb known as "Ma Huang," also known as ephedra and (ii) Kava Kava, including in each case the products set forth on Schedule 1.1(b) and (iii) androstenediol, any derivative or modified version thereof, including but not limited to 4-androstenediol diethylcarbonate ester and 1-androsetediol ethylcarbonate, including the products set forth on Schedule 1.1(b).

- (b) Schedule 1.1(b) to the Purchase Agreement shall be amended by adding to the Exhibit thereto the Addendum to Exhibit to Schedule 1.1(b) attached hereto.
- (c) In connection with the amendment of the definition of "Excluded Products," Sellers shall be entitled to add two hundred ten thousand dollars (\$210,000) to the calculation of Inventory for purposes of the calculation of "Final Current Assets," as defined in Schedule 11.2 (d) to the Purchase Agreement (the "Inventory Adjustment").
- (d) Sellers hereby agree to destroy all products in their Inventory that are Excluded Products within sixty (60) days of the Closing Date.
- 3. Section 3.1 of the Purchase Agreement is hereby deleted in full and replaced with the following:
  - Onsideration. The aggregate cash consideration for the Purchased Assets shall be an amount in cash equal to fifty seven million five hundred thousand dollars (\$57,500,000) less (x) the Zions Mortgage Balance if the Zions Mortgage is assumed by Purchasers and (y) three million dollars (\$3,000,000) representing an estimate of the Inventory Financing (it being understood that the accounts payable of the Sellers owing to the Purchasers as of the Closing Date shall be reduced by such three million dollars (\$3,000,000)) (the "Cash Purchase Price"), plus the Employee and Related Liabilities in an amount up to three million seven hundred thousand dollars (\$3,700,000) (the "Purchase Price"). The aggregate consideration for the Purchased Assets shall also include the assumption of the other Assumed Liabilities. The Purchase Price is subject to adjustment pursuant to Section 3.3(b) hereof.
- 4. Section 3.3 of the Purchase Agreement shall be deleted in full and replaced with the following:

3

#### 3.3 Payment of Purchase Price.

- (a) On the Closing Date, (i) Purchasers shall pay the Cash Purchase Price (less (x) the Deposit Amount, together with all accrued interest and investment income thereon, and (y) the Indemnity Escrow Amount) to Sellers, which shall be paid by wire transfer of immediately available funds into an account designated by Sellers, (ii) Purchasers shall deposit the Indemnity Escrow Amount with the Escrow Agent pursuant to the terms of the Escrow Agreement, and (iii) the Deposit Amount, together with all accrued interest and investment income thereon, shall be released in accordance with Section 3.2 and the terms of Escrow Agreement. The Indemnity Escrow Amount shall be released in accordance with Section 11.3 below and the terms of the Escrow Agreement.
- (b) (i) Not later than sixty (60) days after the Closing Date, Sellers shall prepare and deliver to Purchasers a statement (the "Adjustment Statement"), which shall include a worksheet setting forth a reasonably detailed calculation of (x) the Closing Date Current Liabilities and (y) the December 9 Current Liabilities, each prepared in accordance with GAAP in a manner consistent with past practice and the books and records of each Seller.
- (ii) (A) If the December 9 Current Liabilities are less than the Closing Date Current Liabilities, then Purchasers shall pay to Sellers an amount equal to such difference. If the December 9 Current Liabilities are more than the Closing Date Current Liabilities, then Sellers shall pay to Purchasers an amount equal to such difference.
- (B) Any amount due under clause (ii)(A) above shall be paid in cash within ten (10) days following Purchasers' receipt of the Adjustment Statement by wire transfer of immediately available funds to an account designated by the receiving party, plus interest thereon from and including the Closing Date through and including the day before the date of such payment, at a per annum rate equal to the prime rate (the "Prime Rate") as reported in the Wall Street Journal (under the heading "Money Rates") on the Closing Date. Any amount due to Purchasers under clause (iii)(A) above (following, if applicable, the procedures described in clauses (iii) through (vi) below) shall be an Administrative Claim, and Sellers hereby agree not to object to any such Administrative Claim tendered by Purchasers. Notwithstanding anything in Section 11.3 to the contrary, payments due under this Section 3.3 (b) shall not be paid out of, or limited to, the Indemnity Escrow Amount.

4

- Upon receipt of the Adjustment Statement, Purchasers shall have thirty (30) days to review the Adjustment Statement. Upon the expiration of such 30-day period, Purchasers shall be deemed to have accepted, and shall be bound by, the Adjustment Statement and the calculation therein of the December 9 Current Liabilities and the Closing Date Current Liabilities, unless Purchasers have informed the other parties in writing of their disagreement with the Adjustment Statement prior to the expiration of such thirty (30) day period (the "Adjustment Objection"), specifying each of the disputed items and setting forth in reasonable detail the basis for each such dispute (each, a "Adjustment Disputed Item"). Sellers shall have thirty (30) days from the date on which they receive the Adjustment Objection (the "Adjustment Objection Date") to review and respond to such Adjustment Objection. If Sellers and Purchasers are able to negotiate a mutually agreeable resolution of each Adjustment Disputed Item, and each signs a certificate to that effect, the Adjustment Statement and the calculation therein of the December 9 Current Liabilities and the Closing Date Current Liabilities, as adjusted to reflect such resolution, shall be deemed final, non-appealable and binding for purposes of this Agreement.
- (iv) If within thirty (30) days of the Adjustment Objection Date, Purchasers and Sellers are unable to resolve all of their disagreements with respect to the Adjustment Disputed Items, Purchasers and Sellers shall refer the Adjustment Disputed Items to a mutually acceptable "Big Four" accounting firm (the "Adjustment Accounting Firm") to make a final, non-appealable and binding determination as to such Adjustment Disputed Items. The Adjustment Accounting Firm shall be directed to make a determination of the Adjustment Disputed Items in accordance with GAAP promptly, but no later than thirty (30) days, after acceptance of its appointment. Purchasers and Sellers shall make readily available to the Adjustment Accounting Firm all relevant books, records and employees of the Business reasonably requested by the Adjustment Accounting Firm in connection with the Adjustment Accounting Firm's review of any Adjustment Disputed Item.
- (v) The Ices and expenses of the Adjustment Accounting Firm shall be borne by Purchasers and Sellers in the same proportion that the dollar amount of Adjustment Disputed Items which are not resolved in favor of Purchasers and Sellers (as applicable) bears to the total dollar amount of Adjustment Disputed Items resolved by the Adjustment Accounting Firm. For illustration purposes only, (A) if the total amount of Adjustment Disputed Items by Purchasers is \$1,000,000, and Purchasers are awarded \$500,000 by the Adjustment Accounting Firm, Sellers and Purchasers shall bear

the Accounting Firm's fees and expenses equally; or (B) if the total amount of Adjustment Disputed Items by Purchasers is \$1,000,000, and Purchasers are awarded \$250,000 by the Adjustment Accounting Firm, Purchasers shall bear 75 percent (75%) and Sellers shall bear 25 percent (25%) of the Adjustment Accounting Firm's fees and expenses. Each of Purchasers and Sellers shall bear the fees, costs and expenses of its own accountants and all of its other expenses incurred in connection with matters contemplated by this Section 3.3(b).

- (vi) In the event an Adjustment Objection is submitted, payment of any amounts owing as an adjustment to Purchase Price under this Section 3.3(b), shall be made within ten (10) days following the final resolution of all Adjustment Disputed Items by the parties or the Adjustment Accounting Firm by wire transfer of immediately available funds to an account designated by Purchasers or Sellers, as the case may be, plus interest thereon through and including the day before the date of such payment, at a per annum rate equal to the Prime Rate as in effect on the Closing Date. Purchasers shall have an Administrative Claim for any amounts due under this Section 3.3(b), and Sellers hereby agree not to object to any such Administrative Claims tendered by Purchasers. Notwithstanding anything in Section 11.3 to the contrary, payments due under this Section 3.3(b) shall not be paid out of, or limited to, the Indemnity Escrow Amount.
- 5. Section 4.1 of the Purchase Agreement is hereby deleted in full and replaced with the following:
  - Closing Date. Subject to the satisfaction of the conditions set forth in Sections 10.1, 10.2 and 10.3 hereof (or the waiver thereof by the party entitled to waive such conditions), the closing of the purchase and sale of the Purchased Assets and the assumption of the Assumed Liabilities provided for in Article II hereof (the "Closing") shall take place at the offices of Weil, Gotshal & Manges LLP located at 767 Fifth Avenue, New York, New York (or at such other place as the parties may designate in writing) on December 19, 2003, unless another time or date, or both, are agreed to in writing by the parties hereto. The Sellers and the Purchasers shall use their reasonable best efforts to hold the Closing as soon as possible. The date on which the Closing shall be held is referred to in this Agreement as the "Closing Date."
- 6. Section 4.4(a) of the Purchase Agreement is hereby deleted in full and replaced with the following:

6

KL2 32489149

- (a) by Purchasers or Sellers, if the Closing shall not have occurred by the close of business on December 19, 2003 (the "Termination Date"); provided, that if the Closing shall not have occurred on or before the Termination Date due to a material breach of any representations, warranties, covenants or agreements contained in this Agreement by Purchasers or Sellers, then the breaching parties may not terminate this Agreement pursuant to this Section 4.4(a);
- 7. The second sentence of <u>Section 5.26</u> of the Purchase Agreement is hereby deleted in (all) and replaced with the following:

As of December 9, 2003, the Sellers' Current Assets plus pre-paid amounts for Inventory not yet received and pre-paid advertising, net of any Inventory and accounts receivable relating to Excluded Products, will be no less than thirty nine million five hundred thousand dollars (\$39,500,000), calculated in accordance with GAAP.

8. Clause (a) of Schedule 11.2(d) of the Purchase Agreement is hereby deleted in full and replaced with the following:

Not later than sixty (60) days after the Closing Date, Furchasers shall prepare and deliver to Sellers a statement (the "Final Closing Statement"), which shall include a worksheet setting forth a reasonably detailed calculation of (i) consolidated Current Assets of Sellers plus pre-paid amounts for Inventory not yet received and pre-paid advertising, net of any Inventory and accounts receivable relating to Excluded Products as of December 9, 2003, prepared in accordance with GAAP in a manner consistent with past practice and the books and records of each Seller as of such date (the "Final Current Assets").

- 9. Clause (b)(i) of Schedule 11.2(d) of the Purchase Agreement is hereby deleted in full and replaced with the following:
  - (i) If the Final Current Assets plus the Inventory Adjustment exceeds thirty nine million five hundred thousand dollars (\$39,500,000) (the amount of such excess being referred to as the "Excess"), Purchasers shall pay to Sellers the Excess, up to an aggregate limit of one million five hundred thousand dollars (\$1,500,000).
- 10. Clause (b)(ii) of Schedule 11.2(d) of the Purchase Agreement is hereby deleted in full and replaced with the following:
  - (ii) If the Final Current Assets plus the Inventory Adjustment is less than thirty nine million five hundred thousand dollars (\$39,500,000) (the amount of such difference being referred to as

7

-L2 2341414 9

the "Deficit"), Sellers shall pay to Purchasers the Deficit, provided that the Deficit shall be paid solely from the 11.2(d) Amount, which has an aggregate limit of five hundred thousand dollars (\$500,000).

- 11. Schedules 1.1(b), 1.1(d), 1.1(e), 1.1(f), 5.7, 5.8, 5.9(e), 5.10(a), 5.13 and 5.14(b) are hereby deleted in their entirety and replaced with Schedules 1.1(b), 1.1(d), 1.1(e), 1.1(f), 5.7, 5.8, 5.9(e), 5.10(a), 5.13 and 5.14(b) attached hereto.
  - A new Section 8.15 shall be added to the Purchase Agreement to read as follows:
    - 8.15 Administrative Claims. (a) If the Closing shall occur and (i) the Inventory Financing exceeded three million dollars (\$3,000,000), then the Purchasers shall have an Administrative Claim for the amount of such excess and Sellers hereby agree that they will not object to an such Administrative Claim tendered by the Purchasers pursuant to this Section 8.15(a) or (ii) the Inventory Financing was less than three million dollars (\$3,000,000), then the Purchasers shall reimburse to the Sellers an amount equal to such difference.
    - (b) If the Closing shall not occur, (x) the Purchasers shall have an Administrative Claim for the amount of any and all financing provided by the Purchasers to the Sellers either on, before or after December 9, 2003 for the purchase of Inventory, pre-paid Inventory and pre-paid advertising and (y) Sellers do not waive their right to object to any such Administrative Claim tendered by the Purchasers pursuant to this Section 8.15(b).
- 13. The Purchasers shall reimburse the Sellers an amount equal to one-half of the Extension Payment, by wire transfer of immediately available funds, no later than 15 days after receipt of an invoice therefor from the Sellers.
- Except as expressly set forth herein, this Amendment shall not, by implication or otherwise, limit, impair, constitute a waiver of, or otherwise affect the rights and remedies of any Party under the Asset Purchase Agreement, and shall not alter, modify, amend or in any way affect any of the terms, conditions, obligations, covenants or agreements contained in the Asset Purchase Agreement, all of which are ratified and affirmed in all respects and shall continue in full force and effect. Without limiting the foregoing, nothing in this Amendment shall be construed to cause the Purchasers to incur or assume any Liabilities that are Excluded Liabilities.
- 15. This Amendment shall be governed by and construed in accordance with the laws of the State of New York applicable to contracts made and performed in such State.
- 16. This Amendment may be executed in one or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.

8

KIA MAIDIA 9

→ VARNUM

10/21/04 12:52 FAX 5186303485

17. To the extent that any provision of this Amendment conflicts or is inconsistent with the terms of the Purchase Agreement, this Amendment shall govern, it being agreed that in such event any provisions of the Purchase Agreement required to be amended to give effect to this Amendment shall be deemed amended.

[REMAINDER OF THIS PAGE INTENTIONALLY LEFT BLANK]

9

XL2:2241914.3

IN WITNESS WHEREOF, the Parties have executed this Amendment as of the date first written above.

# **SELLERS**

TWINLAB CORPORATION

By: \_\_\_\_\_\_\_\_ Name: Title:

TWIN LABORATORIES INC.

By: 70° Name: Title:

TWIN LABORATORIES (UK) LTD.

Name: Title:

→ VARNUM

**PURCHASERS** 

IDEA SPHERE, INC.

Title: POS. L-COO

TL ACQUISITION CORP.

Title: LED

11

44.241544.5

→ VARNUM

## AMENDMENT NO. 3 TO ASSET PURCHASE AGREEMENT

This Amendment No. 3 (this "Amendment"), effective as of December 19, 2003, amends the Asset Purchase Agreement, dated as of September 4, 2003 (the "Purchase Agreement"), among TL Administration Corporation (formerly known as Twinlab Corporation), a Delaware corporation, TL Administration Inc. (formerly known as Twin Laboratories Inc.), a Utah corporation, and TL Administration (UK) Ltd. (formerly known as Twin Laboratories (UK) Ltd.), a company organized under the laws of the United Kingdom (each, a "Seller," and collectively, "Sellers"), and IdeaSphere, Inc., a Michigan corporation ("IdeaSphere"), and Twinlab Corporation (formerly known as TL Acquisition Corp.), a Delaware corporation ("Acquisition Sub," each of IdeaSphere and Acquisition Sub being individually referred to as a "Purchaser", and together as "Purchasers"), as previously modified by (i) that certain letter agreement from the Purchasers to the Sellers, dated as of November 14, 2003, (ii) that certain Amendment No. 1 to the Purchase Agreement, dated as of December 5, 2003 and (iii) that certain Amendment No. 2 to the Purchase Agreement dated as of December 19, 2003.

WHEREAS, the parties hereto (each, a "<u>Party</u>" and collectively, the "<u>Parties</u>") are all of the existing parties to the Purchase Agreement; capitalized terms used herein and not otherwise defined shall have the meanings given to such terms in the Purchase Agreement;

WHEREAS, pursuant to <u>Section 13.5</u> of the Purchase Agreement, the Parties may amend, supplement or change the Purchase Agreement (including the schedules and exhibits thereto) in accordance with the provisions of such Section;

NOW, THEREFORE, in consideration of the foregoing and the mutual agreements set forth herein, the Parties hereto agree as follows:

1. Clause (b)(iv) of Schedule 11.2(d) to the Purchase Agreement is hereby deleted in its entirety and replaced with the following:

If Sellers do not timely deliver an Objection (as hereinafter defined) to Purchasers pursuant to clause (c) of this Schedule 11.2(d), then any True-Up Amount shall be paid in cash, within ten (10) days following the expiration of Sellers' 30-day review period of the Final Closing Statement, by wire transfer of immediately available funds to an account designated by the receiving party, plus interest thereon from and including the Closing Date through and including the day before the date of such payment, at a per annum rate equal to the prime rate (the "Prime Rate") as reported in the Wall Street Journal (under the heading "Money Rates") on the Closing Date.

2. The first sentence of <u>Section 3.3(b)(ii)(B)</u> to the Purchase Agreement is hereby deleted in its entirety and replaced with the following:

If Purchasers do not timely deliver an Adjustment Objection (as hereinafter defined) to Sellers pursuant to Section 3.3(b)(iii), then any amount due under Section 3.3(b)(iii)(A) shall be paid in cash,

NY2N37280097XTF9C07LDQCY78300,0005

3. Except as expressly set forth herein, this Amendment shall not, by implication or otherwise, limit, impair, constitute a waiver of, or otherwise affect the rights and remedies of any Party under the Purchase Agreement, and shall not alter, modify, amend or in any way affect any of the terms, conditions, obligations, covenants or agreements contained in the Asset Purchase Agreement, all of which are ratified and affirmed in all respects and shall continue in full force and effect.

Rate") as reported in the Wall Street Journal (under the heading

'Money Rates") on the Closing Date.

- 4. This Amendment shall be governed by and construed in accordance with the laws of the State of New York applicable to contracts made and performed in such State.
- 5. This Amendment may be executed in one or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- 6. To the extent that any provision of this Amendment conflicts or is inconsistent with the terms of the Purchase Agreement, this Amendment shall govern, it being agreed that in such event any provisions of the Purchase Agreement required to be amended to give effect to this Amendment shall be deemed amended.

[REMAINDER OF THIS PAGE INTENTIONALLY LEFT BLANK]

2

MY2-\1372800\07\TP9C071.DOC\78300.0003

!

→ VARNUM

IN WITNESS WHEREOF April, 2004.	, the Parties have executed this Amendment this	day of
	SELLERS	
	TL ADMINISTRATION CORPORATION	
	Name: Denis O'Connor Title: Vice President	
	TL ADMINISTRATION INC.	
	By: Denis O'Connor Title: Vice President	<del></del>
	TL ADMINISTRATION (UK) LTD.	
	By: Denis O'Conner Title: Vice President	<del>-</del>
	<u>FURCHASERS</u>	
	IDEASPHERE, INC.	
, pro-	By: Name: Title:	
	TWINLAB CORPORATION	
	By: Name: Title:	<del></del>
	! 	
NY21/19/2800/07/TF9/C07LDOC\7/8300.0003	3	

IN WITNESS WHEREOF, the Parties have executed this Amendment this day o April, 2004.	f
SELLERS	
TL ADMINISTRATION CORPORATION	
By:	
Title:	
TL ADMINISTRATION INC.	
Ву:	
Name: Title:	
TL ADMINISTRATION (UK) LTD.	
By:	
Name:	
Title:	
PURCHASERS	
IDEASPHERE, INC.	
By: Name: Title:	
TWINLAB CORPORATION	
By: And A Name: Title:	

NY2AJ372900407AYF9C07LDOC578300.0803

**RECORDED: 03/20/2015** 

3